ADDITIONAL EXHIBITS TO THE DECLARATION OF STEPHEN M. JURIS FILED ON BEHALF OF MULTIPLE OBJECTING SECTION 524(G) ASBESTOS TRUSTS ON AUGUST 2, 2010.

EXHIBITS H-N

EXHIBIT H

Exhibit D-1 Owens Corning/Fibreboard Form of Asbestos Personal Injury Distribution Procedures

EXHIBIT D-1

OWENS CORNING/FIBREBOARD

FORM OF ASBESTOS PERSONAL INJURY TRUST DISTRIBUTION PROCEDURES

EXHIBIT D-1

OWENS CORNING/FIBREBOARD

ASBESTOS PERSONAL INJURY TRUST DISTRIBUTION PROCEDURES

TABLE OF CONTENTS

			Page
SECT	TION I	— Introduction	
		The state of the s	1
	1.1	Purpose	
	1.2	Purpose	1
		Interpretation	1
SECT	TON H	- Overview	
		- Overview	2
	2.1	PI Trust Goals	
	2,2	Claims Liquidation Procedures	2
	2.3	Claims Liquidation Procedures	3
	2.4	Application of the Payment Percentage Determination of the Maximum Annual Payment and	6
	,	Maximum Available Deservation Annual Payment and	
	2.5	Maximum Available Payment	8
	2.6	Crusting I divinically Desired	9
	4.0	Indemnity and Contribution Claims	12
SECT	ION III		
	.021 221	— TDP Administration	12
	3.1	PI Trust Advisors Communication	
	4-7 1 E	PI Trust Advisory Committee and Future Claimants'	
	3.2	Representative	12
	J.2	Consent and Consultation Procedures	13
SECTI	ON IV	— Payment Percentage; Periodic Estimates	
		t wy most r dicontage, reflouic issumates	13
	4.1	Uncertainty of OC's and Fibreboard's Total Personal Injury	
		Asbestos Liabilítics Asbestos Liabilítics	
	4.2	Asbestos Liabilities Computation of Payment Parameters	13
	4.3	Computation of Payment Percentage Applicability of the Payment Percentage	14
		remaining of the rayment rementage	15

ECTION	V — Re	solution	of PI Tr	rust Claims	18			
5.1	Orde	ring, Pr	rocessing	and Payment of Claims	10			
	(a)	Örde	ring of C	Claims	18 18			
	,	(1)	Establ	lishment of the FIFO Processing Queues				
		(2)	Effect	of Statutes of Limitations and Repose	18			
	(b)		essing of	Claims	19			
	(c)	Payn	nent of C	laims	20			
5.2		lution o	f Pre_Per	tition Liquidated DI Tarret Claims	21			
	(a)	Proce	ario-ro escina an	tition Liquidated PI Trust Claims	22			
	(b)	Mare	-some an	d Payment	22			
5.3		o acitul	f Inlian	of Security	24 24			
0,5	(a)	Resolution of Unliquidated Pl Trust Claims						
	(4)	(1)	anca Ve	view Process	26			
				neral	26			
		(2)	Ciaima	s Processing under Expedited Review	27			
		(3)	Diseas	se Levels, Scheduled Values				
	a.s	v	and M	edical/Exposure Criteria	27			
	(p)	Indiv	idual Re	view Process	32			
		(1)	in Gen	ieral	32			
			(A)	Review of Medical/Exposure Criteria	33			
			(B)	Review of Liquidated Value	34			
		(2)	Valuat	ion Factors to be Considered in				
			Individ	lual Review	35			
		(3)	Proces.	sing and Payment Limitations for Claims				
			Involvi	ing Disease Levels III and II	36			
			(A)	Disease Level III Claims	36			
			(B)	Disease Level II Claims	37			
		(4)	Schedu	iled, Average and Maximum Values	37			
5.4	Categ	orizing	Claims a	as Extraordinary	J)			
	and/o	or Exigent Hardship						
	(a)	Extra	ordinary	Claims	39			
	(b)	Exige	nt Hards!	hip Claims	40			
5.5	Secon	dary Ex	cposure (Claims	40			
5.6	Indire	ct PI Ti	ust Clair	ns	41			
5.7	Evide	ntiary R	tequirem	ents				
	(a)	Medic	al Evide	nce	43			
		(1)	In Gene	eral	43			
		, ,	(A)	Disease Levels I – IV	43			
				Disease Levels V – VIII	44			
			(C)	Treatment of Certain Pre-Petition Claims	45			
		(2)	Credibi	lity of Medical Evidence	45			
	(b)		ure Evid	ence	46			
	. ,	(1)	in Gene	2 ra/	47			
		(2)	Signific	ant Occupational Exposure	47 48			
		(3)	OC or r	ribrenogra Exposure	48 48			
5.8	Claim	s Audit	rrogram	***************************************	49			
5.9	Second	d Diseas	se (Malis	mancy) Claims	7/			

	5.10		5
		Contraction of Amount of A	51
			5
	5.11		5
	2	Litigation	5
SEC	TION V	I — Claims Materials	
		I — Claims Materials	52
	6.I	Claims Materials	ـر سو
	6.2	Content of Claims Materials	52
	6.3	Windrawar of Deletral of Claims	53
	6.4	Filing Requirements and Fees	53
	6.5	Confidentiality of Claimants' Submissions.	54
	54	outside outsid	
SEC	TION V	II — General Guidelines for Liquidating and Paying Claims	<i>5.5</i>
			55
	7.1	Showing Required	55
	7.2	Costs Considered	55 55
	7.3	Discretion to vary Order and Amounts of Payments in	دد
		Event of Limited Liquidity	56
	7.4	runnive Damages	
	7.5	interest	56
		(a) In General	57
		(b) Omigardated P1 Trust Claims	57
		(c) Interest on Liquidated Pre-Petition Claims.	58
	7.6	Suits in the Toll System	58 50
	7.7	Payment of Judgments for Money Damages	59
	7.8	Releases	60
	7.9	rinid-raity 3crylces	60
	7.10	PI Trust Disclosure of Information	60
			61
SECT	ION VI	II Miscellaneous	61
	8.1	Amendments	
	8.2	Severability	61
	8.3	Governing Law	61
			62

OWENS CORNING/FIBREBOARD

ASBESTOS PERSONAL INJURY TRUST DISTRIBUTION PROCEDURES

The Asbestos Personal Injury Trust Distribution Procedures ("TDP") contained herein provide for resolving all Asbestos Personal Injury Claims for which Owens Corning ("OC") and/or its wholly owned subsidiary, Fibreboard Corporation ("Fibreboard"), and their predecessors, successors, and assigns have legal responsibility (respectively, OC Asbestos Personal Injury Claims ("OC Claims") and FB Asbestos Personal Injury Claims ("Fibreboard Claims"), which terms are defined in the Sixth Amended Joint Plan of Reorganization for Owens Corning and its Affiliated Debtors and Debtors-in-Possession (As Modified) ("Plan") (hereinafter collectively referred to in this TDP as "PI Trust Claims")). The Plan and the Asbestos Personal Injury Trust Agreement ("PI Trust Agreement") establish the Owens Coming/Fibreboard Asbestos Personal Injury Trust (the "PI Trust"). The Trustees of the PI Trust ("Trustees") shall implement and administer this TDP in accordance with the PI Trust Agreement. Capitalized terms used herein and not otherwise defined shall have the meanings assigned to them in the Plan and the PI Trust Agreement.

SECTION 1

Introduction

- 1.1 Purpose. This TDP has been adopted pursuant to the PI Trust Agreement. It is designed to provide fair, equitable, and substantially similar treatment for all PI Trust Claims that may presently exist or may arise in the future.
- 1.2 Interpretation. Except as may otherwise be provided below, nothing in this TDP shall be deemed to create a substantive right for any claimant. The rights and benefits, if

any, provided herein to holders of Pl Trust Claims shall vest in such holders as of the Effective Date.

SECTION II

Overview

2.1 PI Trust Goals. The goal of the PI Trust is to treat all holders of PI Trust Claims equitably and in accordance with the requirements of Section 524(g) of the Bankruptey Code. To achieve that goal, the PI Trust consists of two separate Sub-Accounts, an OC Sub-Account for payment of OC Claims and a Fibreboard Sub-Account for payment of Fibreboard Claims (together the "PI Trust Sub-Accounts").

A claimant may assert separate claims against the OC Sub-Account and the Fibreboard Sub-Account based on separate exposures to asbestos or asbestos-containing products manufactured or distributed by OC and Fibreboard, respectively ("Multiple Exposure Claims"); however, all such Multiple Exposure Claims must be filed by the claimant at the same time. To the extent that the OC Sub-Account and the Fibreboard Sub-Account each has separate liability to a claimant based on Multiple Exposure Claims, each Sub-Account shall pay the claimant the liquidated value of the separate claim for which it is liable, subject to applicable Payment Percentage, Maximum Annual Payment, Maximum Available Payment and Claims Payment Ratio limitations set forth below.

This TDP sets forth procedures for processing and paying all PI Trust Claims from the two Sub-Accounts generally on an impartial, first-in-first-out ("FIPO") basis, with the intention of paying all claimants over time as equivalent a share as possible of the value of their claims

based on historical values for substantially similar claims in the tort system. To this end, this TDP establishes a single schedule of eight asbestos-related diseases ("Disease Levels"), seven of which have presumptive medical and exposure requirements ("Medical/Exposure Criteria") that are applicable to both OC and Fibreboard Claims, as well as two separate schedules with liquidated values ("Scheduled Values"), anticipated average values ("Average Values"), and caps on liquidated values ("Maximum Values") that are applicable to OC Claims and Fibreboard Claims, respectively.

These Disease Levels, Medical/Exposure Criteria, Scheduled Values, Average Values and Maximum Values, which are set forth in Sections 5.3 and 5.4 below, have all been selected and derived with the intention of achieving a fair allocation of the assets held by the separate OC and Fibreboard Sub-Accounts as among their respective claimants suffering from different disease processes in light of the best available information considering the settlement histories of OC and Fibreboard, and the rights that OC and Fibreboard claimants would have in the tort system absent the bankruptcy.

2.2 Claims Liquidation Procedures. PI Trust Claims shall be processed based on their place in separate FIFO Processing Queues to be established for each of the two PI Trust Sub-Accounts pursuant to Section 5.1(a) below. The PI Trust shall take all reasonable steps to resolve OC and Fibreboard Claims as efficiently and expeditiously as possible at each stage of claims processing and arbitration. To this end, the PI Trust, in its sole discretion, may conduct settlement discussions with claimants' representatives with respect to more than one claim at a time, provided that the claimants' respective positions in the FIFO Processing Queues are

As used in this TDP, the phrase "in the tort system" shall include only claims asserted by way of litigation and not claims asserted against a trust established pursuant to section 524(g) and/or section 105 of the Bankruptcy Code or any other applicable law.

maintained, and each claim is individually evaluated pursuant to the valuation factors set forth in Section 5.3(b)(2) below. The PI Trust shall also make every effort to resolve each year at least that number of PI Trust Claims required to exhaust the Maximum Annual Payment and the Maximum Available Payment for Category A and Category B claims, as those terms are defined below.

The PI Trust shall liquidate all OC and Fibreboard Claims except Foreign Claims (as defined in Section 5.3(b)(1) below) that meet the presumptive Medical/Exposure Criteria of Disease Levels I – V, VII and VIII under the Expedited Review Process described in Section 5.3(a) below. PI Trust Claims involving Disease Levels I – V, VII and VIII that do not meet the presumptive Medical/Exposure Criteria for the relevant Disease Level may undergo the PI Trust's Individual Review Process described in Section 5.3(b) below. In such a case, notwithstanding that the claim does not meet the presumptive Medical/Exposure Criteria for the relevant Disease Level, the PI Trust can offer the claimant an amount up to the Scheduled Value of that Disease Level if the PI Trust is satisfied that the claimant has presented a claim that would be cognizable and valid in the tort system.

In lieu of liquidating such claimant's claim under the Expedited Review Process, OC and Fibreboard claimants holding PI Trust Claims involving Disease Levels II - VIII may alternatively seek to establish liquidated values for their claims that are greater than their Scheduled Values by electing the PI Trust's Individual Review Process. However, the liquidated values of PI Trust Claims that undergo the Individual Review Process for valuation purposes may be determined to be less than the Scheduled Values, and in any event shall not exceed the respective Maximum Values for the Disease Levels set forth for OC and Fibreboard Claims in Section 5.3(b)(4) below, unless the claims qualify as Extraordinary Claims as defined

in Section 5.4(a) below, in which case their liquidated value cannot exceed the Maximum Values specified in that provision for such claims. OC and Fibreboard Level VI (Lung Cancer 2) Claims and all Foreign Claims may be liquidated only pursuant to the PI Trust's Individual Review Process.

Based upon OC 's and Fibreboard's claims settlement history in light of applicable tort law, and current projections of present and future unliquidated claims, the Scheduled Values and Maximum Values set forth in Section 5.3(b)(4) for OC and Fibreboard Claims, respectively, have been established for each of the Disease Levels that are eligible for Individual Review of their liquidated values, with the expectation that the combination of settlements at the Scheduled Values and those resulting from the Individual Review Process should result in the Average Values also set forth in that provision.

All unresolved disputes over a claimant's medical condition, exposure history and/or the liquidated value of the claim shall be subject to mandatory pro bono evaluation and mediation and then to binding or non-binding arbitration pursuant to Section 5.10 below, at the election of the claimant, under the ADR Procedures that are provided in Attachment A hereto. PI Trust Claims that are the subject of a dispute with the PI Trust that cannot be resolved by non-binding arbitration may enter the tort system as provided in Sections 5.11 and 7.6 below. However, if and when an OC or Fibreboard claimant obtains a judgment in the tort system, the judgment will be payable (subject to the Payment Percentage, Maximum Available Payment, and Claims Payment Ratio provisions set forth below) as provided in Section 7.7 below.

2.3 Application of the Payment Percentage. After the liquidated value of an OC or Fibreboard Claim other than a claim involving Other Asbestos Disease (Disease Level I – Casb Discount Payment), as defined in Section 5.3(a)(3) below, is determined pursuant to the

procedures set forth herein for Expedited Review, Individual Review, arbitration, or litigation in the tort system, the claimant will ultimately receive a pro-rata share of that value based on the Payment Percentages separately set for OC and Fibreboard Claims pursuant to Section 4.2 below. These Payment Percentages shall also apply to all Pre-Petition Liquidated Claims as provided in Section 5.2 below.

The Initial Payment Percentage for the OC Sub-Account has been set at forty percent (40%), and the Initial Payment Percentage for the Fibreboard Sub-Account has been set at twenty-five percent (25%). These Initial Payment Percentages shall apply to all OC and Fibreboard PI Trust Voting Claims accepted as valid by the PI Trust, unless adjusted by the PI Trust with the consent of the PI Trust Advisory Committee ("TAC") and the Legal Representative for Future Asbestos Claimants ("Future Claimants' Representative") (who are described in Section 3.1 below) pursuant to Section 4.2 below, and except as provided in Section 4.3 below with respect to supplemental payments in the event an Initial Payment Percentage for a Sub-Account is changed.

The term "PI Trust Voting Claims" includes (i) Pre-Petition Liquidated Claims as provided in Section 5.2 below; (ii) OC and Fibreboard Claims filed against OC and/or Fibreboard in the tort system or actually submitted to OC and/or Fibreboard pursuant to an administrative settlement agreement prior to the Petition Date of October 5, 2000; and (iii) all claims filed against another asbestos defendant in the tort system prior to the date the Plan was first filed with the Bankruptcy Court (January 17, 2003 (the "Plan Filing Date")), provided, however, that (1) the holder of a claim described in subsection (i), (ii) or (iii) above, or his or her authorized agent, actually voted to accept or reject the Plan pursuant to the voting procedures established by the Bankruptcy Court unless such holder certifies to the satisfaction of the

Trustees that he or she was prevented from voting in this proceeding as a result of circumstances resulting in a state of emergency affecting, as the case may be, the holder's residence, principal place of business or legal representative's principal place of business at which the holder or his or her legal representative receives notice and/or maintains material records relating to his or her PI Trust Voting Claim, and (2) the claim was subsequently filed with the PI Trust pursuant to Section 6.1 below by the Initial Claims Filing Date as defined in Section 5.1(a) below.

The Initial Payment Percentages for the OC and Fibreboard Sub-Accounts set forth above have been calculated on the assumption that the Average Values set forth in Section 5.3(b)(4) below will be achieved with respect to existing present claims and projected future claims involving Disease Levels II — VIII. However, either or both of these Payment Percentages may be adjusted upwards or downwards from time to time pursuant to Section 4.2 below by the PI Trust with the consent of the TAC and the Future Claimants' Representative to reflect then-current estimates of the assets and liabilities allocable to OC and Fibreboard Claims, respectively, as well as the then-estimated value of pending and future OC and Fibreboard Claims. However, any adjustment to the Initial Payment Percentages shall be made only pursuant to Section 4.2 below. If the Payment Percentage for either the OC or Fibreboard Sub-Account is increased over time, claimants whose OC or Fibreboard Claims were liquidated and paid in prior periods under the TDP will receive additional payments only as provided in Section 4.3 below. Because there is uncertainty in the prediction of both the number and severity of future claims, and the amount of the PI Trust's assets, no guarantee can be made of any Payment Percentage for either OC or Fibreboard Claims.

2.4 Determination of the Maximum Annual Payment and Maximum AvailablePayment. For each of the OC and the Fibreboard Sub-Accounts, the PI Trust shall estimate or

model the amount of eash flow anticipated to be necessary over the entire life of the Sub-Account to ensure that amounts will be available to treat all holders of OC and/or Fibreboard Claims as similarly as possible, given the assets and liabilities allocable to each of the two Sub-Accounts. In each year, for each Sub-Account, the PI Trust will be empowered to pay out all of the income earned during the year by the Sub-Account (net of taxes payable with respect thereto), together with a portion of the Sub-Account's principal, calculated so that the application of the Sub-Account's assets over its life shall correspond with the needs created by the anticipated flow of claims to the Sub-Account (the "Maximum Annual Payment"), taking into account the Payment Percentage provisions set forth in Section 2.3 above and Sections 4.2 and 4.3 below. The PI Trust's distributions from each Sub-Account to all holders of claims against the Sub-Account for that year shall not exceed the Maximum Annual Payment determined for that year.

In distributing the Maximum Annual Payment from each Sub-Account, the PI Trust shall first allocate the amount in question to outstanding Pre-Petition Liquidated Claims (as defined in Section 5.2(a) below) against the Sub-Account, and to liquidated claims against the Sub-Account involving Disease Level I (Cash Discount Payment), in proportion to the aggregate value of each group of claims. The remaining portion of the Maximum Annual Payment (the "Maximum Available Payment"), if any, shall then be allocated and used to satisfy all other previously liquidated PI Trust Claims against the Sub-Account, subject to the Claims Payment Ratio for the Sub-Account set forth in Section 2.5 below.

In the event there are insufficient amounts in any year to pay the total number of outstanding Pre-Petition Liquidated Claims and/or previously liquidated Disease Level I Claims against the Sub-Account, the available amounts allocated to that group of claims shall be paid to

the maximum extent to claimants in the particular group based on their place in their Sub-Account's FIFO Payment Queue. Claims in either group for which there are insufficient amounts in the Sub-Account shall be carried over to the next year and placed at the head of the FIFO Payment Queue for that Sub-Account.

2.5 Claims Payment Ratio. Based upon OC 's and Fibreboard's claims settlement history and analysis of present and future claims, a single Claims Payment Ratio has been determined for both Sub-Accounts, which, as of the Effective Date, has been set at 65% for Category A claims, which consist of PI Trust Claims against OC and/or Fibreboard involving severe asbestosis and malignancies (Disease Levels IV – VIII) that were unliquidated as of the Petition Date, and at 35% for Category B claims, which are PI Trust Claims against OC and/or Fibreboard involving non-malignant Asbestosis or Pleural Disease (Disease Levels II and III) that were similarly unliquidated as of the Petition Date. However, the Claims Payment Ratio shall not apply to any Pre-Petition Liquidated Claims or to any claims for Other Ashestos Disease (Disease Level I - Cash Discount Payment) payable from either OC or Fibreboard Sub-Accounts.

In each year, after the determination of the Maximum Available Payment described in Section 2.4 above, 65% of that amount will be available to pay Category A claims and 35% will be available to pay Category B claims that have been liquidated since the Petition Date. In the event there are insufficient amounts in either the OC or Fibreboard Sub-Accounts in any year to pay the liquidated claims within either or both of the Categories, the available amounts allocated to the particular Category within the Sub-Account shall be paid to the maximum extent to claimants in that Category based on their place in the Sub-Account's FIFO Payment Queue described in Section 5.1(c) below, which will be based upon the date of claim liquidation.

Claims for which there are insufficient amounts allocated to the relevant Category within a Sub-Account shall be carried over to the next year where they will be placed at the head of the Sub-Account's FIFO Payment Queue. If there are excess amounts in either or both Categories within a Sub-Account, because there is an insufficient amount of liquidated claims to exhaust the respective Sub-Account's Maximum Available Payment amount for that Category, then the excess amounts for either or both Categories will be rolled over and remain dedicated to the respective Category to which they were originally allocated.

The 65%/35% Claims Payment Ratio and its rollover provision shall apply to all OC and Fibreboard PI Trust Voting Claims as defined in Section 2.3 above (except Pre-Petition Liquidated Claims and Other Asbestos Claims (Disease Level I – Cash Discount Payment)) and shall not be amended until the third anniversary of the date the PI Trust first accepts for processing proof of claim forms and other materials required to file a claim with the PI Trust. Thereafter, the Sub-Account's Claims Payment Ratio and its rollover provision shall be continued absent circumstances, such as a significant change in law or medicine, necessitating amendment to avoid a manifest injustice. However, the accumulation, rollover and subsequent delay of claims against one or both Sub-Accounts resulting from the application of the Claims Payment Ratio, shall not, in and of itself, constitute such circumstances. Nor may an increase in the numbers of Category B claims against a Sub-Account beyond those predicted or expected be considered as a factor in deciding whether to reduce the percentage allocated to Category A claims.

In considering whether to make any amendments to the Claims Payment Ratio and/or its rollover provisions for either Sub-Account, the Trustees should also consider the reasons for which the Claims Payment Ratio and its rollover provisions were adopted, the settlement

histories of OC and Fibreboard that gave rise to its calculation, and the foreseeability or lack of the foreseeability of the reasons why there would be any need to make an amendment. In that regard, the Trustees should keep in mind the interplay between the Payment Percentage and the Claims Payment Ratio as it affects the net cash actually paid to claimants from either Sub-Account.

In any event, no amendment to the Claims Payment Ratio to reduce the percentage allocated to Category A claims may be made without the unanimous consent of the TAC members and the consent of the Future Claimants' Representative, and the percentage allocated to Category A claims may not be increased without the consent of the TAC and the Future Claimants' Representative. In case of any amendments to the Claims Payment Ratio, consents shall be governed by the consent process set forth in Sections 5.7(b) and 6.6(b) of the PI Trust Agreement. The Trustees, with the consent of the TAC and the Future Claimants' Representative, may offer the option of a reduced Payment Percentage to holders of claims in either Category A or Category B against either Sub-Account in return for prompter payment by the Sub-Account (the "Reduced Payment Option").

2.6 Indemnity and Contribution Claims. As set forth in Section 5.6 below, PI Trust Claims for indemnity and contribution (defined in the Plan as OC Indirect Asbestos Personal Injury Claims and Fibreboard Indirect Asbestos Personal Injury Claims, and hereinafter referred to as "Indirect PI Trust Claims") against either the OC or the Fibreboard Sub-Accounts, if any, will be subject to the same categorization, evaluation, and payment provisions of this TDP as all other OC and Fibreboard Claims.

SECTION III

TDP Administration

- Pursuant to the Plan and the PI Trust Agreement, the PI Trust and this TDP shall be administered by the Trustees in consultation with the TAC, which represents the interests of holders of present PI Trust Claims against OC and Fibreboard, and the Future Claimants' Representative, who represents the interests of holders of PI Trust Claims against OC and/or Fibreboard that will be asserted in the future. The Trustees shall obtain the consent of the TAC and the Future Claimants' Representative on any amendments to these Procedures pursuant to Section 8.1 below, and on such other matters as are otherwise required below and in Section 2.2(f) of the PI Trust Agreement. The Trustees shall also consult with the TAC and the Future Claimants' Representative on such matters as are provided below and in Section 2.2(e) of the PI Trust Agreement. The initial members of the TAC and the initial Future Claimants' Representative are identified in the PI Trust Agreement.
- 3.2 Consent and Consultation Procedures. In those circumstances in which consultation or consent is required, the Trustees will provide written notice to the TAC and the Puture Claimants' Representative of the specific amendment or other action that is proposed. The Trustees will not implement such amendment nor take such action unless and until the parties have engaged in the Consultation Process described in Sections 5.7(a) and 6.6(a), or the Consent Process described in Sections 5.7(b) and 6.6(b) of the PI Trust Agreement, respectively.

SECTION IV

Payment Percentage; Periodic Estimates

4.1 Uncertainty of OC's and Fibreboard's Total Personal Injury Asbestos

Liabilities. As discussed above, there is inherent uncertainty regarding OC's and Fibreboard's total asbestos-related tort liabilities, as well as the total value of the assets available to the OC

and Fibreboard Sub-Accounts to pay PI Trust Claims asserted against each Sub-Account.

Consequently, there is inherent uncertainty regarding the amounts that holders of PI Trust Claims will receive. To seek to ensure substantially similar treatment of all present and future PI Trust Claims against either the OC or the Fibreboard Sub-Accounts, the Trustees must determine from time to time the percentage of full liquidated value that holders of PI Trust Claims against the Sub-Account will be likely to receive, i.e., the "Payment Percentage" described in Section 2.3 above and Section 4.2 below.

Computation of Payment Percentage. As provided in Section 2.3 above, the 4.2 Initial Payment Percentage for claims against the OC Sub-Account shall be forty percent (40%), and for claims against the Fibreboard Sub-Account twenty-five percent (25%). These percentages shall apply to all OC and Fibreboard PI Trust Voting Claims as defined in Section 2.3 above, unless the Trustees, with the consent of the TAC and the Future Claimants' Representative, determine that the Initial Payment Percentage for one or both Sub-Accounts should be changed to assure that the PI Trust will be in a financial position to pay holders of unliquidated and/or unpaid PI Trust Voting Claims and present and future PI Trust Claims against the OC and Fibreboard Sub-Accounts, respectively, in substantially the same manner. In making any such adjustment, the Trustees, the TAC and the Future Claimants' Representative shall take into account the fact that the holders of PI Trust Voting Claims voted on the Plan relying on the findings of experts that the Initial Payment Percentage for each Sub-Account represented a reasonably reliable estimate of the PI Trust's total assets and liabilities over its life based on the best information available at the time, and shall thus give due consideration to the expectations of PI Trust Voting Claimants that the Initial Payment Percentage would be applied to their PI Trust Claims.

Except with respect to PI Trust Voting Claims to which the Initial Payment Percentage applies, the Payment Percentage for either the OC or the Fibreboard Sub-Accounts shall be subject to change pursuant to the terms of this TDP and the PI Trust Agreement if the Trustees determine that an adjustment is required. No less frequently than once every three years, commencing with the first day of January occurring after the Plan is consummated, the Trustees shall reconsider the then applicable Payment Percentage for each of the OC and Fibreboard Sub-Accounts to assure that the respective percentage is based on accurate, current information and may, after such reconsideration, change the Payment Percentage for either Sub-Account if necessary with the consent of the TAC and the Future Claimants' Representative.

The Trustees shall also reconsider the then applicable Payment Percentages for either or both Sub-Accounts at shorter intervals if they deem such reconsideration to be appropriate or if requested to do so by the TAC or the Puture Claimants' Representative. The Trustees must base their determination of the Payment Percentage on current estimates of the number, types, and values of present and future PI Trust Claims against the respective Sub-Accounts, the value of the assets then available to the respective Sub-Accounts for their payment, all anticipated administrative and legal expenses of the respective Sub-Accounts, and any other material matters that are reasonably likely to affect the sufficiency of the respective Sub-Accounts' assets to pay a comparable percentage of full value to all holders of claims against the Sub-Accounts. When making these determinations, the Trustees shall exercise common sense and flexibly evaluate all relevant factors. The Payment Percentage applicable to Category A or Category B claims against the respective Sub-Accounts may not be reduced to alleviate delays in payments of claims in the other Category; both Categories will receive the same Payment Percentage, but the payment from either or both Sub-Accounts may be deferred as needed pursuant to Section 7.3 below, and

a Reduced Payment Option may be instituted for either Sub-Account as described in Section 2.5 above.

4.3 Applicability of the Payment Percentage. Except as set forth below in this Section 4.3 with respect to supplemental payments, no holder of a PI Trust Voting Claim other than a PI Trust Voting Claim for Other Asbestos Disease (Disease Level I - Cash Discount Payment) as defined in Section 5.3(a)(3) below shall receive a payment that exceeds the PI Trust's determination of the Initial Payment Percentage for the relevant Sub-Account of the liquidated value of the claim. Except as otherwise provided in Section 5.1(c) below for PI Trust Claims involving deceased or incompetent claimants for which court or probate approval of the PI Trust's offer is required, no holder of any other PI Trust Claim shall receive a payment that exceeds the Payment Percentage for the respective Sub-Account in effect at the time of payment. PI Trust Claims involving Other Asbestos Disease (Disease Level I - Cash Discount Payment) shall not be subject to such Sub-Account's Payment Percentage, but shall instead be paid the full amount of their Scheduled Value as set forth in Section 5.3(a)(3) below.

If a redetermination of the respective Sub-Account's Payment Percentage has been proposed in writing by the Trustees to the TAC and the Future Claimants' Representative but has not yet been adopted, the claimant shall receive the lower of such Sub-Account's current Payment Percentage or the proposed Payment Percentage. However, if the proposed Payment Percentage for such Sub-Account was the lower amount but was not subsequently adopted, the claimant shall thereafter receive the difference between the lower proposed amount and the higher current amount. Conversely, if the proposed Payment Percentage for such Sub-Account was the higher amount and was subsequently adopted, the claimant shall thereafter receive the difference between the lower current amount and the higher adopted amount.

There is uncertainty surrounding the amount of the PI Trust's future assets. There is also uncertainty surrounding the totality of the PI Trust Claims to be paid over time as well as the extent to which changes in existing federal and state law could affect the PI Trust's liabilities under this TDP. If the value of the PI Trust's future assets increases significantly and/or if the value or volume of PI Trust Claims actually filed with the PI Trust is significantly lower than originally estimated, the PI Trust shall use those proceeds and/or claims savings, as the case may be, first to maintain the Payment Percentage then in effect.

If the Trustees, with the consent of the TAC and the Future Claimants' Representative, make a determination to increase the Payment Percentage due to a material change in the estimates of the PI Trust's future assets and/or liabilities, the Trustees shall also make supplemental payments to all claimants who previously liquidated their claims against the PI Trust and received payments based on a lower Payment Percentage. The amount of any such supplemental payment shall be the liquidated value of the claim in question times the newly adjusted Payment Percentage, less all amounts previously paid to the claimant with respect to the claim (excluding the portion of such previously paid amounts that was attributable to interest paid pursuant to Section 7.5 below).

The Trustees' obligation to make a supplemental payment to a claimant shall be suspended in the event the payment in question would be less than \$100.00, and the amount of the suspended payment shall be added to the amount of any prior supplemental payment/payments that was/were also suspended because it/they would have been less than \$100.00. However, the Trustees' obligation shall resume and the Trustees shall pay any such aggregate supplemental payments due the claimant at such time that the total exceeds \$100.00.

SECTION V

Resolution of PI Trust Claims.

5.1 Ordering, Processing and Payment of Claims.

5.1(a) Ordering of Claims.

5.1(a)(1) Establishment of FIFO Processing Queues. The PI Trust will order separately all OC and Fibreboard Claims that are sufficiently complete to be reviewed for processing purposes on a FIFO basis except as otherwise provided herein (the "FIFO Processing Queues"). For all claims filed on or before the date six months after the date that the PI Trust first makes available proof of claim forms and other claims materials required to file a claim with the PI Trust (the "Initial Claims Filing Date"), a claimant's position in either FIFO Processing Queue shall be determined as of the earliest of (i) the date prior to the Petition Date (if any) that the specific claim was either filed against OC or Fibreboard in the tort system or was actually submitted to OC or Fibreboard pursuant to an administrative settlement agreement; (ii) the date before the Petition Date that the claim was filed against another asbestos defendant in the tort system if at the time the claim was subject to a tolling agreement with OC or Fibreboard; (iii) the date after the Petition Date but before the Initial Claims Filing Date that the claim was filed against another asbestos defendant in the tort system; (iv) the date after the Petition Date but before the Effective Date the claimant filed a proof of claim form in OC's and/or Fibreboard's Chapter 11 proceeding; or (v) the date after the Petition Date the claimant submitted a ballot in OC's Chapter 11 proceeding for purposes of voting on the Plan pursuant to the voting procedures approved by the Bankruptcy Court.

Following the Initial Claims Filing Date, the claimant's position in one of the two FIFO Processing Queues shall be determined by the date the claim was filed with the PI Trust. If any

claims are filed on the same date, the claimant's position in the FIFO Processing Queue shall be determined by date of the claimant's diagnosis of asbestos-related disease. If any claims are filed and diagnosed on the same date, the claimant's position in the FIFO Processing Queue shall be determined by the date of the claimant's birth, with older claimants given priority over younger claimants.

5.1(a)(2) Effect of Statutes of Limitations and Repose. All

unliquidated PI Trust Claims must meet either, (i) for claims first filed in the tort system against OC or Fibreboard, respectively prior to the Petition Date, the applicable federal, state and foreign statute of limitation and repose that was in effect at the time of the filing of the claim in the tort system, or (ii) for claims that were not filed against either OC or Fibreboard in the tort system prior to the Petition Date, the applicable federal, state or foreign statute of limitation that was in effect at the time of the filing with the PI Trust.

However, the running of the relevant statute of limitation shall be tolled as of the earliest of (A) the actual filing of the claim against OC or Fibreboard prior to the Petition Date, whether in the tort system or by submission of the claim to OC or Fibreboard pursuant to an administrative settlement agreement; (B) the filing of the claim against another defendant in the tort system prior to the Petition Date if the claim was tolled against OC or Fibreboard at the time by an agreement or otherwise; (C) the filing of a claim after the Petition Date but prior to the Initial Claims Filing Date against another defendant in the tort system; (D) the date after the Petition Date but before the Effective Date that a proof of claim was filed against OC or Fibreboard in OC's and or Fibreboard's Chapter 11 proceeding; (E) the date a ballot was submitted by the claimant in OC's and or Fibreboard's Chapter 11 proceeding for purposes of voting on the Plan pursuant to the voting procedures approved by the Bankruptcy Court; or (F)

the filing of a proof of claim with the requisite supporting documentation with the PI Trust after the Initial Claims Filing Date.

If a PI Trust Claim meets any of the tolling provisions described in the preceding sentence and the claim was not barred by the applicable federal, state or foreign statute of limitation at the time of the tolling event, it will be treated as timely filed if it is actually filed with the PI Trust within three (3) years after the Initial Claims Filing Date. In addition, any claims that were first diagnosed after the Petition Date, irrespective of the application of any relevant federal, state or foreign statute of limitation or repose, may be filed with the PI Trust within three (3) years after the date of diagnosis or within three (3) years after the Initial Claims Filing Date, whichever occurs later. However, the processing of any PI Trust Claim by the PI Trust may be deferred at the election of the claimant pursuant to Section 6.3 below.

- 5.1(b) Processing of Claims. As a general practice, the PI Trust will review its claims files on a regular basis and notify all claimants whose claims are likely to come up in either the OC or Fibreboard FIFO Processing Queue in the near future. However, claims that were not filed (i) against OC or Fibreboard in the tort system or actually submitted to OC or Fibreboard pursuant to an administrative settlement agreement prior to the Petition Date, or (ii) against another asbestos defendant in the tort system prior to the Plan Filing Date, shall not be processed until after the Initial Claims Filing Date.
- 5.1(c) Payment of Claims. PI Trust Claims against the OC and/or Fibreboard Sub-Accounts that have been liquidated under the provisions of this TDP by the Expedited Review Process as provided in Section 5.3(a) below, by the Individual Review Process as provided in Section 5.3(b) below, by arbitration as provided in Section 5.10 below, or by litigation in the tort system provided in Section 5.11 below, shall be paid in FIFO order from the

relevant Sub-Account based on the date their liquidation became final (the "FIFO Payment Queue"), all such payments being subject to the applicable Payment Percentage, the Maximum Annual Payment, the Maximum Available Payment, and the Claims Payment Ratio, except as otherwise provided herein. Pre-Petition Liquidated Claims, as defined in Section 5.2 below, shall be subject to the Maximum Annual Payment and Payment Percentage limitations, but not to the Maximum Available Payment and Claims Payment Ratio provisions set forth above.

Where the claimant is deceased or incompetent, and the settlement and payment of his or her claim must be approved by a court of competent jurisdiction or through a probate process prior to acceptance of the claim by the claimant's representative, an offer made by the PI Trust on the claim shall remain open so long as proceedings before that court or in that probate process remain pending, provided that the PI Trust has been furnished with evidence that the settlement offer has been submitted to such court or probate process for approval. If the offer is ultimately approved by the court or through the probate process and accepted by the claimant's representative, the PI Trust shall pay the claim from the relevant Sub-Account in the amount so offered, multiplied by the Payment Percentage in effect for such Sub-Account at the time the offer was first made.

If any claims are liquidated on the same date, the claimant's position in a Sub-Account's FIFO Payment Queue shall be determined by the date of the diagnosis of the claimant's asbestos-related disease. If any claims are liquidated on the same date and the respective holders' asbestos-related diseases were diagnosed on the same date, those claimants' positions in the Sub-Account's FIFO Payment Queue shall be determined by the PI Trust based on the dates of the claimants' birth, with older claimants given priority over younger claimants.

5.2 Resolution of Pre-Pctition Liquidated Pl Trust Claims.

5.2(a) Processing and Payment. As soon as practicable after the Effective Date, the PI Trust shall pay, upon submission by the claimant of the appropriate documentation, all PI Trust Claims that were liquidated (i) by a binding settlement agreement for the particular claim entered into prior to the Pctition Date that is judicially enforceable by the claimant, (ii) after the Petition Date according to the terms of a binding settlement agreement entered into prior to the Petition Date (a "Pre-Petition Agreement"), (iii) by a jury verdict or non-final judgment in the tort system obtained prior to the Petition Date, or (iv) by a judgment that became final and non-appealable prior to the Petition Date (collectively "Pre-Petition Liquidated Claims"). In order to receive payment from the PI Trust, the holder of a Pre-Petition Liquidated Claim must submit all documentation necessary to demonstrate to the PI Trust that the claim was liquidated in the manner described in the preceding sentence, which documentation shall include (A) a court authenticated copy of the jury verdict, non-final judgment or final judgment, if applicable, and (B) the name, social security number and date of birth of the claimant and the name and address of the claimant's lawyer; provided, however, that such documentation shall not be required with respect to any Pre-Petition Liquidated Claim that OC or Fibreboard has identified to the PI Trust as a Pre-Petition Liquidated Claim as to which all conditions to payment under the applicable agreement, jury verdict or judgment have been satisfied. OC and Fibreboard shall deliver to the PI Trust a list of the Pre-Petition Liquidated Claims that OC and Fibreboard have approved for payment (the "Approved Pre-Petition Liquidated Claims"), which claims shall be entitled to rely upon the exception set forth in the preceding sentence.

The liquidated value of a Pre-Petition Liquidated Claim shall be the unpaid portion of the amount agreed to in the binding settlement agreement or Pre-Petition Agreement, the unpaid portion of the amount awarded by the jury verdict or non-final judgment, or the unpaid portion of

the amount of the final judgment, as the case may be, plus interest, if any, that has accrued on that amount in accordance with the terms of a binding settlement agreement or Pre-Petition Agreement, if any, or under applicable state law for settlements or judgments as of the Petition Date; however, except as otherwise provided in Section 7.4 below, the liquidated value of a Pre-Petition Liquidated Claim shall not include any punitive or exemplary damages. In addition, the amounts payable with respect to such claims shall not be subject to or taken into account in consideration of the Claims Payment Ratio and the Maximum Available Payment limitations, but shall be subject to the Maximum Annual Payment and Payment Percentage provisions. In the absence of a Final Order of the Bankruptcy Court determining whether a settlement agreement is binding and judicially enforceable, a dispute between a claimant and the PI Trust over this issue shall be resolved pursuant to the same procedures in this TDP that are provided for resolving the validity and/or liquidated value of a PI Trust Claim (i.e., arbitration and litigation in the tort system as set forth in Sections 5.10 and 5.11 below).

The PI Trust shall pay the Approved Pre-Petition Liquidated Claims as expeditiously as possible. The other Pre-Petition Liquidated Claims shall be processed and paid from the OC and/or Fibreboard Sub-Accounts in accordance with their order in separate FIFO queues to be established for each Sub-Account by the PI Trust based on the date the PI Trust received all required documentation for the particular claim. If any Pre-Petition Liquidated Claims are filed with the PI Trust on the same date, the claimant's position in the Sub-Account's FIFO queue for such claims shall be determined by the date on which the claim was liquidated. If any Pre-Petition Liquidated Claims are both filed with the PI Trust and liquidated by a Sub-Account on the same dates, those claimants' positions in the FIFO queue shall be determined by the dates of the claimants' birth, with older claimants given priority over younger claimants.

- 5.2(b) Marshalling of Security. Holders of Pre-Petition Liquidated Claims that are secured by letters of credit, appeal bonds, or other security or sureties shall first exhaust their rights against any applicable security or surety before making a claim against the PI Trust. Only in the event that such security or surety is insufficient to pay the Pre-Petition Liquidated Claim in full shall the deficiency be processed and paid as a Pre-Petition Liquidated Claim.
- establishment of the PI Trust, the Trustees with the consent of the TAC and the Future
 Claimants' Representative shall adopt procedures for reviewing and liquidating all unfiquidated
 PI Trust Claims, which shall include deadlines for processing such claims. Such procedures shall
 also require claimants seeking resolution of unliquidated PI Trust claims to first file a proof of
 claim form, together with the required supporting documentation, in accordance with the
 provisions of Sections 6.1 and 6.2 below. It is anticipated that the PI Trust shall provide an
 initial response to the claimant within six months of receiving the proof of claim form.

The proof of claim form shall require the claimant to assert his or her OC and/or Fibreboard Claim for the highest Disease Level for which the claim qualifies at the time of filing. Irrespective of the Disease Level alleged on the proof of claim form, each OC and/or Fibreboard Claims shall be deemed to be a claim for the highest Disease Level for which the claim qualifies at the time of filing, and all lower Disease Levels for which the claim may also qualify at the time of filing or in the future shall be treated as subsumed into the higher Disease Level for both processing and payment purposes.

Upon filing of a valid proof of claim form with the required supporting documentation, the claim shall be placed in the relevant OC and/or Fibreboard FIFO Processing Queue in accordance with the ordering criteria described in Section 5.1(a) above. The PI Trust shall

provide the claimant with six-months notice of the date by which it expects to reach the claim in the FIFO Processing Queue, following which the claimant shall promptly (i) advise the PI Trust whether the claim should be liquidated under the PI Trust's Expedited Review Process described in Section 5.3(a) below or, in certain circumstances, under the PI Trust's Individual Review Process described in Section 5.3(b) below; (ii) provide the PI Trust with any additional medical and/or exposure evidence that was not provided with the original claim submission; and (iii) advise the PI Trust of any change in the claimant's Disease Level. If a claimant fails to respond to the PI Trust's notice prior to the reaching of the claim in the FIFO Processing Queue, the PI Trust will process and liquidate the claim under the Expedited Review Process based upon the medical/exposure evidence previously submitted by the claimant, although the claimant shall retain the right to request Individual Review as described in Section 5.3(b) below.

5.3(a) Expedited Review Process.

5.3(a)(1) In General. The PI Trust's Expedited Review Process is designed primarily to provide an expeditious, efficient and inexpensive method for liquidating all OC and Fibreboard Claims (except those involving Lung Cancer 2 - Disease Level VI and all Foreign Claims, which must be liquidated pursuant to the PI Trust's Individual Review process) where the claim can easily be verified by the PI Trust as meeting the presumptive Medical/Exposure Criteria for the relevant Disease Level. Expedited Review thus provides claimants with a substantially less burdensome process for pursuing PI Trust Claims than does the Individual Review Process described in Section 5.3(b) below. Expedited Review is also intended to provide qualifying claimants a fixed and certain claims payment.

Thus, claims that undergo Expedited Review and meet the presumptive

Medical/Exposure Criteria for the relevant Disease Level shall be paid the Scheduled Value (or

Values in the case of Multiple Exposure Claims) for such Disease Level set forth in Section 5.3(a)(3) below. However, except for claims involving Other Asbestos Disease (Disease Level I), all claims liquidated by Expedited Review shall be subject to the applicable Payment Percentage, the Maximum Annual Payment, the Maximum Available Payment, and the Claims Payment Ratio limitations set forth herein. Claimants holding OC and/or Fibreboard Claims that cannot be liquidated by Expedited Review because they do not meet the presumptive Medical/Exposure Criteria for the relevant Disease Level may elect the PI Trust's Individual Review Process set forth in Section 5.3(b) below.

Subject to the provisions of Section 5.8, the claimant's eligibility to receive the Scheduled Value for his or her PI Trust Claim pursuant to the Expedited Review Process shall be determined solely by reference to the Medical/Exposure Criteria set forth below for each of the Disease Levels eligible for Expedited Review.

5.3(a)(2) Claims Processing under Expedited Review. All claimants seeking liquidation of an OC and/or Fibreboard Claim pursuant to Expedited Review shall file the PI Trust's proof of claim forms provided in Attachment B hereto. As a proof of claim form is reached in the OC or Fibreboard FIFO Processing Queue, the PI Trust shall determine whether the claim described therein meets the Medical/Exposure Criteria for one of the seven Disease Levels eligible for Expedited Review, and shall advise the claimant of its determination. If a Disease Level is determined, the PI Trust shall tender to the claimant an offer of payment from the relevant OC or Fibreboard Sub-Account of the Scheduled Value for the relevant Disease Level multiplied by the applicable Payment Percentage, together with a form of release approved by the PI Trust. If the claimant accepts the Scheduled Value and returns the release properly executed, the claim shall be placed in the Sub-Account's FIFO Payment Queue, following which

the PI Trust shall disburse payment subject to the limitations of the Maximum Available Payment and Claims Payment Ratio, if any.

Criteria. The eight Disease Levels covered by this TDP, together with the Medical/Exposure Criteria for each, and the separate OC and Fibreboard Scheduled Values for the seven Disease Levels eligible for Expedited Review, are set forth below. These Disease Levels, Scheduled Values, and Medical/Exposure Criteria shall apply to all PI Trust Voting Claims (other than Pre-Petition Liquidated Claims) filed with the PI Trust on or before the Initial Claims Filing Date provided in Section 5.1 above for which the claimant elects the Expedited Review Process.

Thereafter, for purposes of administering the Expedited Review Process and with the consent of the TAC and the Future Claimants' Representative, the Trustees may add to, change or eliminate Disease Levels, Scheduled Values, or Medical/Exposure Criteria; develop subcategories of Disease Levels, Scheduled Values or Medical/Exposure Criteria; or determine that a novel or exceptional asbestos personal injury claim is compensable even though it does not meet the Medical/Exposure Criteria for any of the then current Disease Levels.

Disease Level	OC/F	ibreboard Scheduled Values	Medical/Exposure Criteria
Mesothelioma (Level	VIII)	\$215,000/\$135,000	(1) Diagnosis ² of mesothelioma; and (2) credible evidence of OC or Fibreboard Exposure (as defined in Section 5.7(b)(3) below)
Lung Cancer 1 (Level	VII)	\$ 40,000/\$27,000	(1) Diagnosis of a primary lung cancer plus evidence of an underlying Bilateral Asbestos-

The requirements for a diagnosis of an asbestos-related disease that may be compensated under the provisions of this TDP are set forth in Section 5.7 below.

Related Nonmalignant Disease³, (2) six months OC or Fibreboard Exposure prior to December 31, 1982, (3) Significant Occupational Exposure to asbestos, 4 and (4) supporting medical documentation establishing asbestos exposure as a contributing factor in causing the lung cancer in question.

Lung Cancer 2 (Level VI)

None

(1) Diagnosis of a primary lung cancer; (2) OC or Fibreboard Exposure prior to December 31, 1982, and (3) supporting medical documentation establishing asbestos exposure as a contributing factor in causing the lung cancer in question.

Lung Cancer 2 (Level VI) claims are claims that do not meet the more

Evidence of "Bilateral Asbestos-Related Nonmalignant Disease" for purposes of meeting the criteria for establishing Disease Levels I, II, III, V, and VII, means either (i) a chest X-ray read by a qualified B reader of 1/0 or higher on the ILO scale or (ii)(x) a chest x-ray read by a qualified B reader or other Qualified Physician, (y) a CT scan read by a Qualified Physician, or (z) pathology, in each case showing either bilateral interstitial fibrosis, bilateral pleural plaques. bilateral pleural thickening, or bilateral pleural calcification. Solely for claims filed against OC or Fibreboard or another asbestos defendant in the tort system prior to the Petition Date, if an ILO reading is not available, either (i) a chest X-ray or a CT scan read by a Qualified Physician, or (ii) pathology, in each case showing either bilateral interstitial fibrosis, bilateral pleural plaques, bilateral pleural thickening, or bilateral pleural calcification, consistent with or compatible with a diagnosis of asbestos-related disease, shall be evidence of a "Bilateral Asbestos-Related Nonmalignant Disease" for purposes of meeting the presumptive medical requirements of Disease Levels I, II, III, V and VII. Pathological evidence of asbestosis may be based on the pathological grading system for asbestosis described in the Special Issue of the Archives of Pathology and Laboratory Medicine, "Asbestos-associated Diseases," Vol. 106, No. 11, App. 3 (October 8, 1982). For all purposes of this TDP, a "Qualified Physician" is a physician who is board certified (or in the case of Canadian claims or Foreign Claims, a physician who is certified or qualified under comparable medical standards or criteria of the jurisdiction in question) in one or more relevant specialized fields of medicine such as pulmonology, radiology, internal medicine or occupational medicine; provided, however, subject to the provisions of Section 5.8, that the requirement for board certification in this provision shall not apply to otherwise qualified physicians whose x-rays and/or CT scan readings are submitted for deceased holders of PI Trust Claims.

[&]quot;Significant Occupational Exposure" is defined in Section 5.7 below.

stringent medical and/or exposure requirements of Lung Cancer (Level VII) claims. All claims in this Disease Level will be individually evaluated. The estimated likely Average Value of the individual evaluation awards for this category for OC Claims is \$20,000 and for Fibreboard Claims is \$12,000, with such awards capped at a Maximum Value of \$50,000 for OC Claims and \$30,000 for Fibreboard Claims, unless the claim qualifies for Extraordinary Claim treatment (discussed in Section 5.4 below).

Level VI claims that show no evidence of either an underlying Bilateral Asbestos-Related Nonmalignant Disease or Significant Occupational Exposure may be individually evaluated, although it is not expected that such claims will be treated as having any significant value, especially if the claimant is also a Smoker. In any event, no presumption of validity will be available for any claims in this category.

Other Cancer (Level V)

\$ 22,000/\$12,000

(1) Diagnosis of a primary colorectal, laryngeal, esophageal, pharyngeal, or stomach cancer, plus evidence of an underlying Bilateral Asbestos-Related Nonmalignant

There is no distinction between Non-Smokers and Smokers for either Lung Cancer (Level VII) or Lung Cancer (Level VI), although a claimant who meets the more stringent requirements of Lung Cancer (Level VII) (evidence of an underlying Bilateral Asbestos-Related Nonmalignant Disease plus Significant Occupational Exposure), and who is also a Non-Smoker, may wish to have his or her claim individually evaluated by the PI Trust. In such a case, absent circumstances that would otherwise reduce the value of the claim, it is anticipated that the liquidated value of the claim might well exceed the Scheduled Values for Lung Cancer (Level VII) claims against OC and Fibreboard, respectively, shown above. "Non-Smoker" means a claimant who either (a) never smoked or (b) has not smoked during any portion of the twelve (12) years immediately prior to the diagnosis of the lung cancer.

Disease, (2) six months OC or Fibreboard Exposure prior to December 31, 1982, (3) Significant Occupational Exposure to asbestos, and (4) supporting medical documentation establishing asbestos exposure as a contributing factor in causing the other cancer in question.

Severe Asbestosis (Level IV) \$42,000/\$29,000

(1) Diagnosis of asbestosis with ILO of 2/1 or greater, or asbestosis determined by pathological evidence of asbestos, plus (a)TLC less than 65%, or (b) FVC less than 65% and FEV1/FVC ratio greater than 65%, (2) six months OC or Fibreboard Exposure prior to December 31, 1982, (3) Significant Occupational Exposure to asbestos, and (4) supporting medical documentation establishing asbestos exposure as a contributing factor in causing the pulmonary disease in question.

Asbestosis/
Pleural Disease (Level III) \$ 19,000/\$11,500

(1) Diagnosis of Bilateral Asbestos-Related Nonmalignant Disease plus (a) TLC less than 80%, or (b) FVC less than 80% and FEV1/FVC ratio greater than or equal to 65%, and (2) six months OC or Fibreboard Exposure prior to December 31, 1982, (3) Significant Occupational Exposure to asbestos, and (4) supporting medical documentation establishing asbestos exposure as a contributing factor in causing the pulmonary disease in question.

Asbestosis/
Pleural Disease (Level II) \$ 8,000/\$4,500

(1) Diagnosis of a Bilateral Asbestos-Related Nonmalignant Disease, and (2) six months OC or Fibreboard Exposure prior to December 31, 1982, and (3) five

years cumulative occupational exposure to asbestos.

Other Asbestos Disease (Level 1 - Cash Discount Payment) \$ 400/\$240

(1) Diagnosis of a Bilateral Asbestos-Related Nonmalignant Disease or an asbestos-related malignancy other than mesothelioma, and (2) OC or Fibreboard Exposure prior to December 31, 1982.

5.3(b) Individual Review Process

5.3(b)(1) In General. Subject to the provisions set forth below, an OC or Fibreboard Claimant may elect to have his or her PI Trust Claim reviewed for purposes of determining whether the claim would be compensable in the tort system even though it does not meet the presumptive Medical/Exposure Criteria for any of the Disease Levels set forth in Section 5.3(a)(3) above. In addition or alternatively, an OC or Fibreboard claimant may elect to have a claim undergo the Individual Review Process for purposes of determining whether the liquidated value of the claim exceeds the Scheduled Value for the relevant Disease Level also set forth in said provision. However, until such time as the PI Trust has made an offer on a claim pursuant to Individual Review, the claimant may change his or her Individual Review election and have the claim liquidated pursuant to the PI Trust's Expedited Review Process. In the event of such a change in the processing election, the claimant shall nevertheless retain his or her place in the FIFO Processing Queue.

The liquidated value of all Foreign Claims shall be established pursuant to the PI Trust's Individual Review Process. PI Trust Claims of individuals exposed in Canada who were resident in Canada when such claims were filed ("Canadian Claims") shall not be considered Foreign Claims hereunder and shall be eligible for liquidation under the Expedited Review Process.

Accordingly, a "Foreign Claim" is a PI Trust Claim with respect to which the claimant's exposure to an asbestos-containing product for which OC and or Fibreboard has legal responsibility occurred outside of the United States and its Territories and Possessions and outside of the Provinces and Territories of Canada.

In reviewing Foreign Claims, the PI Trust shall take into account all relevant procedural and substantive legal rules to which the claims would be subject in the Claimant's Jurisdiction as defined in Section 5.3(b)(2) below. The PI Trust shall determine the liquidated value of Foreign Claims based on historical settlements and verdicts in the Claimant's Jurisdiction as well as the other valuation factors set forth in Section 5.3(b)(2) below.

For purposes of the Individual Review process, the Trustees, with the consent of the TAC and the Future Claimants' Representative, may develop separate Medical/Exposure Criteria and standards, as well as separate requirements for physician and other professional qualifications, which shall be applicable to Foreign Claims; provided, however, that such criteria, standards or requirements shall not effectuate substantive changes to the claims eligibility requirements under this TDP, but rather shall be made only for the purpose of adapting those requirements to the particular licensing provisions and/or medical customs or practices of the foreign country in question.

At such time as the PI Trust has sufficient historical settlement, verdict and other valuation data for claims from a particular foreign jurisdiction, the Trustees, with the consent of the TAC and the Future Claimants' Representative, may also establish a separate valuation matrix for such Foreign Claims based on that data.

5.3(b)(1)(A) Review of Medical/Exposure Criteria. The

PI Trust's Individual Review Process provides an OC or Fibreboard claimant with an

opportunity for individual consideration and evaluation of a PI Trust Claim that fails to meet the presumptive Medical/Exposure Criteria for Disease Levels I – V, VII or VIII. In such a case, the PI Trust shall either deny the claim, or, if the PI Trust is satisfied that the claimant has presented a claim that would be cognizable and valid in the tort system, the PI Trust can offer the claimant a liquidated value amount up to the Scheduled Value for that Disease Level, unless the claim qualifies as an Extraordinary Claim as defined in Section 5.4(a) below, in which case its liquidated value cannot exceed the Maximum Value for such a claim.

5.3(b)(1)(B) Review of Liquidated Value. Claimants

holding claims involving Disease Levels II – VIII shall also be eligible to seek Individual Review of the liquidated value of their OC and Fibreboard Claims, as well as of their medical/exposure evidence. The Individual Review Process is intended to result in payments from the OC and/or Fibreboard Sub-Accounts equal to the full liquidated value for each claim multiplied by the Payment Percentage; however, the liquidated value of any OC or Fibreboard Claim that undergoes Individual Review may be determined to be less than the Scheduled Value the claimant would have received under Expedited Review. Moreover, the liquidated value for a claim involving Disease Levels II – VIII shall not exceed the Maximum Value for the relevant Disease Level set forth in Section 5.3(b)(4) below, unless the claim meets the requirements of an Extraordinary Claim described in Section 5.4(a) below, in which case its liquidated value cannot exceed the Maximum Value set forth in that provision for such claims. Because the detailed examination and valuation process pursuant to Individual Review requires substantial time and effort, claimants electing to undergo the Individual Review Process may be paid the liquidated value of their PI Trust Claims later than would have been the case had the claimant elected the Expedited Review Process. Subject to the provisions of Section 5.8, the PI Trust shall devote

reasonable resources to the review of all claims to ensure that there is a reasonable balance maintained in reviewing all classes of claims.

5.3(b)(2) Valuation Factors to be Considered in Individual Review.

The PI Trust shall liquidate the value of each OC and Fibreboard Claim that undergoes Individual Review based on the historic liquidated values of other similarly situated claims in the tort system for the same Disease Level. The PI Trust will thus take into consideration all of the factors that affect the severity of damages and values within the tort system including, but not limited to credible evidence of (i) the degree to which the characteristics of a claim differ from the presumptive Medical/Exposure Criteria for the Disease Level in question; (ii) factors such as the claimant's age, disability, employment status, disruption of household, family or recreational activities, dependencies, special damages, and pain and suffering; (iii) whether the claimant's damages were (or were not) caused by asbestos exposure to an asbestos-containing product prior to December 31, 1982 for which OC or Fibreboard has legal responsibility (for example, alternative causes, and the strength of documentation of injuries); (iv) the industry of exposure; (v) settlements and verdict histories in the Claimant's Jurisdiction for similarly situated claims; and (vi) settlement and verdict histories for the claimant's law firm for similarly situated claims.

For these purposes, the "Claimant's Jurisdiction" is the jurisdiction in which the claim was filed (if at all) against OC or Fibreboard in the tort system prior to the Petition Date. If the claim was not filed against OC or Fibreboard in the tort system prior to the Petition Date, the claimant may elect as the Claimant's Jurisdiction either (i) the jurisdiction in which the claimant resides at the time of diagnosis or when the claim is filed with the PI Trust; or (ii) a jurisdiction in which the claimant experienced exposure to an asbestos-containing product for which OC or Fibreboard has legal responsibility.

With respect to the "Claimant's Jurisdiction" in the event a personal representative or authorized agent makes a claim under this TDP for wrongful death with respect to which the governing law of the Claimant's Jurisdiction could only be the Alabama Wrongful Death Statute, the Claimant's Jurisdiction for such claim shall be the Commonwealth of Pennsylvania, and such claimant's damages shall be determined pursuant to the statutory and common laws of the Commonwealth of Pennsylvania without regard to its choice of law principles. The choice of law provision in Section 7.4 below applicable to any claim with respect to which, but for this choice of law provision, the applicable law of the Claimant's Jurisdiction pursuant to Section 5.3(b)(2) is determined to be the Alabama Wrongful Death Statute, shall only govern the rights between the PI Trust and the claimant, and, to the extent the PI Trust seeks recovery from any entity that provided insurance coverage to OC and or Fibreboard, the Alabama Wrongful Death Statute shall govern.

5.3(b)(3) Processing and Payment Limitations for Claims Involving

Disease Levels III and II. The PI Trust shall administer Individual Review for Disease Levels

III and II so that Individual Review does not reduce payments to claimants electing the

Scheduled Value for such PI Trust Claims under Expedited Review. As one means of

implementing this requirement, the following shall apply for Disease Levels III and II claims:

5.3(b)(3)(A) Disease Level III Claims. No more than 13% or 9% of Disease Level III claims paid in any year from either the OC or the Fibreboard Sub-Account, respectively, shall be PI Trust Claims allowed under Individual Review, and the total payments to such Disease Level III claims allowed under Individual Review shall be no more than 17% or 13% of payments to all Disease Level III claimants from either the OC or Fibreboard Sub-Account, respectively, during any year.

5.3(b)(3)(B) Disease Level II Claims. No more than 15% or 20% of Disease Level II claims paid in any year from either the OC or the Fibreboard Sub-Account, respectively, shall be PI Trust Claims allowed under Individual Review, and the total payments to such Disease Level II claims allowed under Individual Review shall be no more than 24% or 33% of payments to all Disease Level II claimants from either the OC or Fibreboard Sub-Account, respectively, during any year.

5.3(b)(4) Scheduled, Average and Maximum Values. The Scheduled, Average and Maximum Values for the Disease Levels compensable under this TDP from the OC and Fibreboard Sub-Accounts are the following:

OC SUB-ACCOUNT

Scheduled Disease	Scheduled Value	Average Value	Maximum Value
Mesothelioma (Level VIII)	\$215,000	\$270,000	\$650,000
Lung Cancer 1 (Level VII)	\$ 40,000	\$ 50,000	\$150,000
Lung Cancer 2 (Level VI)	None	\$ 20,000	\$ 50,000
Other Cancer (Level V)	\$ 22,000	\$ 25,000	\$ 60,000
Severe Asbestosis (Level IV)	\$ 42,000	\$ 50,000	\$150,000
Asbestosis/Pleural Disease (Level III)	\$ 19,000	\$ 20,000	\$ 35,000

Asbestosis/Pleural Disease (Level II)	\$ 8,000	\$ 9,000	\$ 20,000
Other Asbestos Disease Cash Discount Payment (Level I)	\$ 400	None	None

FIBREBOARD SUB-ACCOUNT

Scheduled Disease	Scheduled Value	Average Value	Maximum Value
Mesothelioma (Level VIII)	\$135,000	\$180,000	\$450,000
Lung Cancerl (Level VII)	\$ 27,000	\$ 35,000	\$ 90,000
Lung Cancer 2 (Level VI)	None	\$ 12,000	\$ 30,000
Other Cancer (Level V)	\$ 12,000	\$ 15,000	\$ 36,000
Severe Asbestosis (Level IV)	\$ 29,000	\$ 30,000	\$ 90,000
Asbestosis/Pleural Disease (Level III)	\$ 11,500	\$ 12,000	\$ 21,000
Asbestosis/Pleural Disease (Level II)	\$ 4,500	\$ 5,400	\$ 12,000
Other Asbestos Disease Cash Discount Payment (Level I)	\$ 240	None	None

These OC and Fibreboard Scheduled Values, Average Values and Maximum Values shall apply to all PI Trust Voting Claims other than Pre-Petition Liquidated Claims filed with the PI Trust on or before the Initial Claims Filing Date as provided in Section 5.1 above. Thereafter, the PI Trust, with the consent of the TAC and the Future Claimants' Representative pursuant to Sections 5.7(b) and 6.6(b) of the PI Trust Agreement, may change these valuation amounts for good cause and consistent with other restrictions on the amendment power.

5.4 Categorizing Claims as Extraordinary and/or Exigent Hardship

that otherwise satisfies the Medical Criteria for Disease Levels II - VIII, and that is held by a claimant whose exposure to asbestos (i) occurred predominately as the result of working in a manufacturing facility of OC or Fibreboard during a period in which OC or Fibreboard was manufacturing asbestos-containing products at that facility, or (ii) was at least 75% the result of exposure to an asbestos-containing product for which OC or Fibreboard has legal responsibility, and in either case there is little likelihood of a substantial recovery elsewhere. All such Extraordinary Claims shall be presented for Individual Review and, if valid, shall be entitled to an award of up to a Maximum Value of five (5) times the Scheduled Value for claims qualifying for Disease Levels II – V, VII and VIII, and five (5) times the Average Value for claims in Disease Level VI, multiplied by the applicable Payment Percentage.

Any dispute as to Extraordinary Claim status shall be submitted to a special Extraordinary Claims Panel to be established by the PI Trust with the consent of the TAC and the Future Claimants' Representative. All decisions of the Extraordinary Claims Panel shall be final and not subject to any further administrative or judicial review. An Extraordinary Claim, following its liquidation, shall be placed in the Trust's FIFO Queue ahead of all other PI Trust Claims except Pre-Petition Liquidated Claims, Disease Level I (Other Asbestos Disease) Claims and Exigent Hardship Claims, which shall be paid first in that order in said Queue, based on its date of liquidation and shall be subject to the Maximum Available Payment and Claims Payment Ratio described above.

5.4(b) Exigent Hardship Claims. At any time the PI Trust may liquidate and pay PI Trust Claims that qualify as Exigent Hardship Claims as defined below. Such claims may

be considered separately no matter what the order of processing otherwise would have been under this TDP. An Exigent Hardship Claim, following its liquidation, shall be placed first in the relevant Sub-Account's FIFO Payment Queue ahead of all other liquidated claims except Pre-Petition Liquidated Claims and Disease Level I (Other Asbestos Disease) Claims, and shall be subject to the Maximum Available Payment and Claims Payment Ratio described above. A PI Trust Claim qualifies for payment as an Exigent Hardship Claim if the claim meets the Medical/Exposure Criteria for Severe Asbestosis (Disease Level IV) or an asbestos-related malignancy (Disease Levels V-VIII), and the PI Trust, in its sole discretion, determines (i) that the claimant needs financial assistance on an immediate basis based on the claimant's expenses and all sources of available income, and (ii) that there is a causal connection between the claimant's dire financial condition and the claimant's asbestos-related disease.

resulting solely from exposure to an occupationally exposed person, such as a family member, the claimant is entitled to seek Individual Review of his or her OC and/or Fibreboard Claim pursuant to Section 5.3(b) above. In such a case, the claimant must establish that the occupationally exposed person would have met the exposure requirements under this TDP that would have been applicable had that person filed a direct claim against the PI Trust. In addition, the claimant with secondary exposure must establish that he or she is suffering from one of the eight Disease Levels described in Section 5.3(a)(3) above or an asbestos-related disease otherwise compensable under this TDP, that his or her own exposure to the occupationally exposed person occurred within the same time frame as the occupationally exposed person was exposed to asbestos products produced by OC or Fibreboard, and that such secondary exposure to OC or Fibreboard products was a cause of the claimed disease. The proof of claim form

included in Attachment B hereto contains an additional section for Secondary Exposure Claims.

All other liquidation and payment rights and limitations under this TDP shall be applicable to such claims.

Indirect PI Trust Claims, Indirect PI Trust Claims asserted against either the OC 5.6 or Fibreboard Sub-Accounts based upon theories of contribution or indemnification under applicable law, shall be treated as presumptively valid and paid by the PI Trust subject to the applicable Payment Percentage if (a) such claim satisfied the requirements of the Bar Date for such claims established by the Bankruptcy Court, if applicable, and is not otherwise disallowed by Section 502(e) of the Code or subordinated under Section 509(c) of the Code, and (b) the holder of such claim (the "Indirect Claimant") establishes to the satisfaction of the Trustees that (i) the Indirect Claimant has paid in full the liability and obligation of the Trust to the individual claimant to whom the PI Trust would otherwise have had a liability or obligation under these Procedures (the "Direct Claimant"), (ii) the Direct Claimant and the Indirect Claimant have forever and fully released the Trust from all liability to the Direct Claimant, and (iii) the claim is not otherwise barred by a statute of limitation or repose or by other applicable law. In no event shall any Indirect Claimant have any rights against the PI Trust superior to the rights of the related Direct Claimant against the PI Trust, including any rights with respect to the timing, amount or manner of payment. In addition, no Indirect Claim may be liquidated and paid in an amount that exceeds what the Indirect Claimant has actually paid the related Direct Claimant.

To establish a presumptively valid Indirect PI Trust Claim, the Indirect Claimant's aggregate liability for the Direct Claimant's claim must also have been fixed, liquidated and paid fully by the Indirect Claimant by settlement (with an appropriate full release in favor of the PI Trust) or a Final Order (as defined in the Plan) provided that such claim is valid under the

applicable state law. In any case where the Indirect Claimant has paid the claim of a Direct Claimant against the PI Trust under applicable law by way of a settlement, the Indirect Claimant shall obtain for the benefit of the PI Trust a release in form and substance satisfactory to the Trustees.

If an Indirect Claimant cannot meet the presumptive requirements set forth above, including the requirement that the Indirect Claimant provide the PI Trust with a full release of the Direct Claimant's claim, the Indirect Claimant may request that the PI Trust review the Indirect PI Trust Claim individually to determine whether the Indirect Claimant can establish under applicable state law that the Indirect Claimant has paid all or a portion of a liability or obligation that the PI Trust had to the Direct Claimant as of the effective date of this TDP. If the Indirect Claimant can show that it has paid all or a portion of such a liability or obligation, the PI Trust shall reimburse the Indirect Claimant the amount of the liability or obligation so paid, times the then applicable Payment Percentage. However, in no event shall such reimbursement to the Indirect Claimant be greater than the amount to which the Direct Claimant would have otherwise been entitled. Further, the liquidated value of any Indirect PI Trust Claim paid by the PI Trust to an Indirect Claimant shall be treated as an offset to or reduction of the full liquidated value of any Pl Trust Claim that might be subsequently asserted by the Direct Claimant against the PI Trust.

Any dispute between the PI Trust and an Indirect Claimant over whether the Indirect Claimant has a right to reimbursement for any amount paid to a Direct Claimant shall be subject to the ADR procedures provided in Section 5.10 below and set forth in Attachment A hereto. If such dispute is not resolved by said ADR procedures, the Indirect Claimant may litigate the dispute in the tort system pursuant to Sections 5.11 above and 7.6 below.

The Trustees may develop and approve a separate proof of claim form for Indirect PI
Trust Claims. Indirect PI Trust Claims that have not been disallowed, discharged, or otherwise
resolved by prior order of the Bankruptcy Court shall be processed in accordance with
procedures to be developed and implemented by the Trustees consistent with the provisions of
this Section 5.6, which procedures (a) shall determine the validity, allowability and
enforceability of such claims; and (b) shall otherwise provide the same liquidation and payment
procedures and rights to the holders of such claims as the PI Trust would have afforded the
holders of the underlying valid PI Trust Claims. Nothing in this TDP is intended to preclude a
trust to which asbestos-related liabilities are channeled from asserting an Indirect PI Trust Claim
against the PI Trust subject to the requirements set forth herein.

5.7 Evidentiary Requirements

5.7(a) Medical Evidence.

5.7(a)(1) In General. All diagnoses of a Disease Level shall be accompanied by either (i) a statement by the physician providing the diagnosis that at least 10 years have elapsed between the date of first exposure to asbestos or asbestos-containing products and the diagnosis, or (ii) a history of the claimant's exposure sufficient to establish a 10-year latency period. A finding by a physician after the Petition Date that a claimant's disease is "consistent with" or "compatible with" asbestosis will not alone be treated by the PI Trust as a diagnosis. 6

All diagnoses of Asbestosis/Pleural Disease (Disease Levels II and III) not based on pathology shall be presumed to be based on findings of bilateral asbestosis or pleural disease, and all diagnoses of Mesothelioma (Disease Level VIII) shall be presumed to be based on findings that the disease involves a malignancy. However, the PI Trust may refute such presumptions.

5.7(a)(1)(A). Disease Levels I-IV. Except for claims filed

against OC, Fibreboard or another asbestos defendant in the tort system prior to the Petition

Date, all diagnoses of a non-malignant asbestos-related disease (Disease Levels I-IV) shall be
based in the case of a claimant who was living at the time the claim was filed, upon a physical
examination of the claimant by the physician providing the diagnosis of the asbestos-related
disease. In addition, all living claimants must provide (i) for Disease Levels I-III, evidence of
Bilateral Asbestos-Related Nonmalignant Disease (as defined in Footnote 3 above); (ii)for
Disease Level IV, an ILO reading of 2/1 or greater or pathological evidence of asbestosis, and
(iii) for Disease Levels III and IV, pulmonary function testing.⁷

In the case of a claimant who was deceased at the time the claim was filed, all diagnoses of a non-malignant asbestos-related disease (Disease Levels I-IV) shall be based upon either (i) a physical examination of the claimant by the physician providing the diagnosis of the asbestos-related disease; or (ii) pathological evidence of the non-malignant asbestos-related disease; or (iii) in the case of Disease Levels I-III, evidence of Bilateral Asbestos-Related Nonmalignant Disease (as defined in Footnote 3 above), and for Disease Level IV, either an ILO

[&]quot;Pulmonary function testing" or "PFT" shall mean testing that is in material compliance with the quality criteria established by the American Thoracic Society ("ATS") and is performed on equipment which is in material compliance with ATS standards for technical quality and calibration. PFT performed in a hospital accredited by the JCAHO, or performed, reviewed or supervised by a board certified pulmonologist or other Qualified Physician shall be presumed to comply with ATS standards, and the claimant may submit a summary report of the testing. If the PFT was not performed in a JCAHO accredited hospital, or performed, reviewed or supervised by a board certified pulmonologist or other Qualified Physician, the claimant must submit the full report of the testing (as opposed to a summary report); provided, however, that if the PFT was conducted prior to the Effective Date of the Plan and the full PFT report is not available, the claimant must submit a declaration signed by a Qualified Physician or other qualified party in the form provided by the PI Trust certifying that the PFT was conducted in material compliance with ATS standards.

reading of 2/I or greater or pathological evidence of asbestosis; or (iv) for either Disease Level III or IV, pulmonary function testing.

5.7(a)(1)(B). Disease Levels V - VIII. All diagnoses of an asbestos-related malignancy (Disease Levels V - VIII) shall be based upon either (i) a physical examination of the claimant by the physician providing the diagnosis of the asbestos-related disease, or (ii) on a diagnosis of such a malignant Disease Level by a board-certified pathologist.

5.7(a)(1)(C). Exception to the Exception for Certain

Pre-Petition Claims. If the holder of a PI Trust Claim that was filed against OC, Fibreboard or another defendant in the tort system prior to the Petition Date has available a report of a diagnosing physician engaged by the holder or his or her law firm who conducted a physical examination of the holder as described in Section 5.7(a)(1)(A), or if the holder has filed such medical evidence and/or a diagnosis of the asbestos-related disease by a physician not engaged by the holder or his or her law firm who conducted a physical examination of the claimant with another asbestos-related personal injury settlement trust that requires such evidence, without regard to whether the diagnosing physician was engaged by the holder or his or her law firm, the holder shall provide such medical evidence to the PI Trust notwithstanding the exception in Sections 5.7(a)(1)(A).

5.7(a)(2) Credibility of Medical Evidence. Before making any payment to a claimant, the PI Trust must have reasonable confidence that the medical evidence provided in support of the claim is credible and consistent with recognized medical standards. The PI Trust may require the submission of X-rays, CT scans, detailed results of pulmonary function tests, laboratory tests, tissue samples, results of medical examination or

reviews of other medical evidence, and may require that medical evidence submitted comply with recognized medical standards regarding equipment, testing methods and procedure to assure that such evidence is reliable. Medical evidence (i) that is of a kind shown to have been received in evidence by a state or federal judge at trial, (ii) that is consistent with evidence submitted to OC to settle for payment similar disease cases prior to OC 's bankruptcy, or (iii) that is a diagnosis by a physician shown to have previously qualified as a medical expert with respect to the asbestos-related disease in question before a state or federal judge, is presumptively reliable, although the PI Trust may seek to rebut the presumption.

In addition, claimants who otherwise meet the requirements of this TDP for payment of a PI Trust Claim shall be paid irrespective of the results in any litigation at any time between the claimant and any other defendant in the tort system. However, any relevant evidence submitted in a proceeding in the tort system involving another defendant, other than any findings of fact, a verdict, or a judgment, may be introduced by either the claimant or the PI Trust in any Individual Review proceeding conducted pursuant to 5.3(b) or any Extraordinary Claim proceeding conducted pursuant to 5.4(a).

5.7(b) Exposure Evidence

5.7(b)(1) In General. As set forth in Section 5.3(a)(3) above, to qualify for any Disease Level, the claimant must demonstrate a minimum exposure to an asbestos-containing product manufactured or distributed by OC or Pibreboard. Claims based on conspiracy theories that involve no exposure to an asbestos-containing product produced by OC or Fibreboard are not compensable under this TDP. To meet the presumptive exposure requirements of Expedited Review set forth in Section 5.3(a)(3) above, the claimant must show (i) for all Disease Levels, OC or Fibreboard Exposure as defined in Section 5.7(b)(3) below prior

to December 31, 1982; (ii) for Asbestos/Pleural Disease Level II, six months OC or Fibreboard Exposure prior to December 31, 1982, plus five years cumulative occupational asbestos exposure; and (iii) for Asbestosis/Pleural Disease (Disease Level III), Severe Asbestosis (Disease Level IV), Other Cancer (Disease Level V) or Lung Cancer 1 (Disease Level VII), the claimant must show six months OC or Fibreboard Exposure prior to December 31, 1982, plus Significant Occupational Exposure to asbestos as defined below. If the claimant cannot meet the relevant presumptive exposure requirements for a Disease Level eligible for Expedited Review, the claimant may seek Individual Review pursuant to Section 5.3(b) above of his or her exposure to an asbestos-containing product for which by OC or Fibreboard has legal responsibility.

5.7(b)(2) Significant Occupational Exposure. "Significant Occupational Exposure." Significant Occupational Exposure" means employment for a cumulative period of at least five years, with a minimum of two years prior to December 31, 1982 in an industry and an occupation in which the claimant (a) handled raw asbestos fibers on a regular basis; (b) fabricated asbestos-containing products so that the claimant in the fabrication process was exposed on a regular basis to raw asbestos fibers; (c) altered, repaired or otherwise worked with an asbestos-containing product such that the claimant was exposed on a regular basis to asbestos fibers; or (d) was employed in an industry and occupation such that the claimant worked on a regular basis in close proximity to workers engaged in the activities described in (a), (b) and/or (c).

5.7(b)(3) OC or Fibreboard Exposure. All PI Trust claimants must demonstrate meaningful and credible exposure, which occurred prior to December 31, 1982, to asbestos or asbestos-containing products supplied, specified, manufactured, installed, maintained, or repaired by either OC or Fibreboard, and/or any entity, including an OC or Fibreboard contracting unit, for which OC or Fibreboard has legal liability ("OC/Fibreboard Exposure").

That meaningful and credible exposure evidence may be established by an affidavit or sworn statement of the claimant, by an affidavit or sworn statement of a co-worker or the affidavit or sworn statement of a family member in the case of a deceased claimant (providing the PI Trust finds such evidence reasonably reliable), by invoices, employment, construction or similar records, or by other credible evidence. The specific exposure information required by the PI Trust to process a claim under either Expedited or Individual Review shall be set forth on the proof of claim form to be used by the PI Trust. The PI Trust can also require submission of other or additional evidence of exposure when it deems such to be necessary.

Future Claimants' Representative may develop methods for auditing the reliability of medical evidence, including additional reading of X-rays, CT scans and verification of pulmonary function tests, as well as the reliability of evidence of exposure to asbestos, including exposure to asbestos-containing products manufactured or distributed by OC or Fibreboard prior to December 31, 1982. In the event that the PI Trust reasonably determines that any individual or entity has engaged in a pattern or practice of providing unreliable medical evidence to the Trust, it may decline to accept additional evidence from such provider in the future.

Further, in the event that an audit reveals that fraudulent information has been provided to the PI Trust, the PI Trust may penalize any claimant or claimant's attorney by disallowing the PI Trust Claim and/or by other means including, but not limited to, requiring the source of the fraudulent information to pay the costs associated with the audit and any future audit or audits, reordering the priority of payment of all affected claimants' PI Trust Claims, raising the level of scrutiny of additional information submitted from the same source or sources, refusing to accept additional evidence from the same source or sources, seeking the prosecution of the claimant or

claimant's attorney for presenting a fraudulent claim in violation of 18 U.S.C. §152, and seeking sanctions from the Bankruptcy Court.

Section 2.1 that provides that a claimant may not assert more than one PI Trust Claim hereunder, the holder of a PI Trust Claim involving a non-malignant asbestos-related disease (Disease Levels I through IV) may file a new PI Trust Claim against the PI Trust for a malignant disease (Disease Levels V – VIII) that is subsequently diagnosed. Any additional payments to which such claimant may be entitled with respect to such malignant asbestos-related disease shall not be reduced by the amount paid for the non-malignant asbestos-related disease, provided that the malignant disease had not been diagnosed at the time the claimant was paid with respect to his or her original claim involving the non-malignant disease.

5.10 Arbitration.

the TAC and the Future Claimants' Representative, shall institute binding and non-binding arbitration procedures in accordance with the ADR Procedures included in Attachment A hereto for resolving disputes concerning whether a Pre-Petition settlement agreement with OC or Fibreboard is binding and judicially enforceable in the absence of a Final Order of the Bankruptcy Court determining the issue, whether the PI Trust's outright rejection or denial of a claim was proper, or whether the claimant's medical condition or exposure history meets the requirements of this TDP for purposes of categorizing a claim involving Disease Levels I – VIII. Binding and non-binding arbitration shall also be available for resolving disputes over the liquidated value of a claim involving Disease Levels II – VIII as well as disputes over OC's or Fibreboard's share of

the unpaid portion of a Pre-Petition Liquidated Claim described in Section 5.2 above and disputes over the validity of an Indirect PI Trust Claim.

In all arbitrations, the arbitrator shall consider the same medical and exposure evidentiary requirements that are set forth in Section 5.7 above. In the case of an arbitration involving the liquidated value of a claim involving Disease Levels II – VIII, the arbitrator shall consider the same valuation factors that are set forth in Section 5.3(b)(2) above. With respect to all claims eligible for arbitration, the claimant, but not the PI Trust, may elect either non-binding or binding arbitration. The ADR Procedures set forth in Attachment A hereto may be modified by the PI Trust with the consent of the TAC and the Future Claimants' Representative. Such amendments may include adoption of mediation procedures as well as establishment of an Extraordinary Claims Panel to review such claims pursuant to Section 5.4(a) above.

5.10(b) Claims Eligible for Arbitration. In order to be eligible for arbitration, the claimant must first complete the Individual Review Process set forth in Section 5.3(b) above, as well as either the Pro-Bono Evaluation or the Mediation processes set forth in the ADR Procedures included in Attachment A, with respect to the disputed issue. Individual Review will be treated as completed for these purposes when the claim has been individually reviewed by the PI Trust, the PI Trust has made an offer on the claim, the claimant has rejected the liquidated value resulting from the Individual Review, and the claimant has notified the PI Trust of the rejection in writing. Individual Review will also be treated as completed if the PI Trust has rejected the claim.

5.10(c) Limitations on and Payment of Arbitration Awards. In the case of a non-Extraordinary Claim involving Disease Levels II – VIII, the arbitrator shall not return an award in excess of the Maximum Value for the appropriate Disease Level as set forth in Section 5.3(b)(4) above, and for an Extraordinary Claim involving one of those Disease Levels, the

arbitrator shall not return an award greater than the maximum extraordinary value for such a claim as set forth in Section 5.4(a) above. A claimant who submits to arbitration and who accepts the arbitral award will receive payments in the same manner as one who accepts the PI Trust's original valuation of the claim.

5.11 Litigation. Claimants who elect non-binding arbitration and then reject their arbitral awards retain the right to institute a lawsuit in the tort system against the PI Trust pursuant to Section 7.6 below. However, a claimant shall be eligible for payment of a judgment for monetary damages obtained in the tort system from the PI Trust's available cash only as provided in Section 7.7 below.

SECTION VI

Claims Materials

Claims Materials. The PI Trust shall prepare suitable and efficient claims materials ("Claims Materials") for all PI Trust Claims, and shall provide such Claims Materials upon a written request for such materials to the PI Trust. The proof of claim form to be submitted to the PI Trust shall require the claimant to assert the highest Disease Level for which the claim qualifies at the time of filing, and shall include a certification by the claimant or his or her attorney sufficient to meet the requirements of Rule 11(b) of the Federal Rules of Civil Procedure. In developing its claim filing procedures, the PI Trust shall make every reasonable effort to provide claimants with the opportunity to utilize currently available technology at their discretion, including filing claims and supporting documentation over the internet and electronically by disk or CD-rom. The proof of claim forms to be used by the PI Trust shall be developed by the TAC and submitted to the PI Trust and the Future Claimants' Representative

for approval. The proof of claim forms may be changed by the PI Trust with the consent of the TAC and the Future Claimants' Representative.

- Content of Claims Materials. The Claims Materials shall include a copy of this TDP, such instructions as the Trustees shall approve, and a detailed proof of claim form. If feasible, the forms used by the PI Trust to obtain claims information shall be the same or substantially similar to those used by other asbestos claims resolution organizations. Instead of collecting some or all of the claims information from a claimant or the claimant's attorney, the PI Trust may also obtain such information from electronic data bases maintained by any other asbestos claims resolution organization. However, the PI Trust shall inform the claimant that it plans to obtain information as available from such other organizations and may do so unless the claimant objects in writing or provides such information directly to the PI Trust. If requested by the claimant, the PI Trust shall accept information provided electronically. The claimant may, but will not be required to, provide the PI Trust with evidence of recovery from other asbestos defendants and claims resolution organizations.
- at any time upon written notice to the PI Trust and file another claim subsequently without affecting the status of the claim for statute of limitations purposes, but any such claim filed after withdrawal shall be given a place in the FIFO Processing Queue based the date of such subsequent filing. A claimant can also request that the processing of his or her PI Trust Claim by the PI Trust be deferred for a period not to exceed three (3) years without affecting the status of the claim for statute of limitation purposes, in which case the claimant shall also retain his or her original place in the FIFO Processing Queue. During the period of such deferral, interest on such claimant's PI Trust Claim as provided in Section 7.5 hereunder shall not accrue and payment

thereof shall be deemed waived by the claimant. Except for PI Trust Claims held by representatives of deceased or incompetent claimants for which court or probate approval of the PI Trust's offer is required, or a PI Trust Claim for which deferral status has been granted, a claim will be deemed to have been withdrawn if the claimant neither accepts, rejects, nor initiates arbitration within six months of the PI Trust's written offer of payment or rejection of the claim. Upon written request and good cause, the PI Trust may extend either the deferral or withdrawal period for an additional six months.

- 6.4 Filing Requirements and Fees. The Trustees shall have the discretion to determine, with the consent of the TAC and the Futures Representative, (a) whether a claimant must have previously filed an asbestos-related personal injury claim in the tort system to be eligible to file the claim with the PI Trust and (b) whether a filing fee should be required for any PI Trust Claims.
- a holder of a PI Trust Claim or a proof of claim form and materials related thereto shall be treated as made in the course of settlement discussions between the holder and the PI Trust and intended by the parties to be confidential and to be protected by all applicable state and federal privileges, including, but not limited to, those directly applicable to settlement discussions. The PI Trust will preserve the confidentiality of such claimant submissions, and shall disclose the contents thereof only, with the permission of the holder, to another trust established for the benefit of asbestos personal injury claimants pursuant to section 524(g) and/or section 105 of the Bankruptcy Code or other applicable law, to such other persons as authorized by the holder, or in response to a valid subpoena of such materials issued by the Bankruptcy Court. Furthermore, the PI Trust shall provide counsel for the holder a copy of any such subpoena immediately upon

being served. The PI Trust shall on its own initiative or upon request of the claimant in question take all necessary and appropriate steps to preserve said privilege before the Bankruptcy Court and before those courts having appellate jurisdiction related thereto.

SECTION VII

General Guidelines for Liquidating and Paying Claims

- 7.1 Showing Required. To establish a valid PI Trust Claim, a claimant must meet the requirements set forth in this TDP. The PI Trust may require the submission of X-rays, CT scans, laboratory tests, medical examinations or reviews, other medical evidence, or any other evidence to support or verify the PI Trust Claim, and may further require that medical evidence submitted comply with recognized medical standards regarding equipment, testing methods, and procedures to assure that such evidence is reliable.
- Tosts Considered. Notwithstanding any provisions of this TDP to the contrary, the Trustees shall always give appropriate consideration to the cost of investigating and uncovering invalid PI Trust Claims so that the payment of valid PI Trust Claims is not further impaired by such processes with respect to issues related to the validity of the medical evidence supporting a PI Trust Claim. The Trustees shall also have the latitude to make judgments regarding the amount of transaction costs to be expended by the PI Trust so that valid PI Trust Claims are not unduly further impaired by the costs of additional investigation. Nothing herein shall prevent the Trustees, in appropriate circumstances, from contesting the validity of any claim against the PI Trust whatever the costs, or declining to accept medical evidence from sources that the Trustees have determined to be unreliable pursuant to the Claims Audit Program described in Section 5.8 above.

7.3 Discretion to Vary the Order and Amounts of Payments in Event of Limited Liquidity. Consistent with the provisions hereof and subject to the FIFO Processing and Liquidation Queues, the Maximum Annual Payment, the Maximum Available Payment and the Claims Payment Ratio requirements set forth above, the Trustees shall proceed as quickly as possible to liquidate valid PI Trust Claims, and shall make payments to holders of such claims in accordance with this TDP from the OC and/or Fibreboard Sub-Accounts promptly as monies become available and as claims are liquidated, while maintaining sufficient assets within each Sub-Account to pay future valid claims in substantially the same manner.

Because the PI Trust's income over time remains uncertain, and decisions about payments must be based on estimates that cannot be done precisely, they may have to be revised in light of experiences over time, and there can be no guarantee of any specific level of payment for claims against either Sub-Account. However, the Trustees shall use their best efforts to treat similar claims in substantially the same manner, consistent with their duties as Trustees, the purposes of the PI Trust, the established allocation of monies to claims in Categories A and B, and the practical limitations imposed by the inability to predict the future with precision. In the event that either or both of the OC or the Fibreboard Sub-Accounts face temporary periods of limited liquidity, the Trustees may, with the consent of the TAC and the Future Claimants' Representative, suspend the normal order of payment from such Sub-Account, may temporarily limit or suspend payments from such Sub-Account altogether, and may offer a Reduced Payment Option for the Sub-Account as described in Section 2.5 above.

7.4 Punitive Damages. Except as provided below for claims asserted under the Alabama Wrongful Death Statute, in determining the value of any liquidated or unliquidated PI Trust Claim, punitive or exemplary damages, i.e., damages other than compensatory damages,

shall not be considered or allowed, notwithstanding their availability in the tort system.

Similarly, no punitive or exemplary damages shall be payable with respect to any claim litigated against the PI Trust in the tort system pursuant to Sections 5.11 above and 7.6 below. The only damages that may be awarded pursuant to this TDP to Alabama Claimants who are deceased and whose personal representatives pursue their claims only under the Alabama Wrongful Death Statute shall be compensatory damages determined pursuant to the statutory and common law of the Commonwealth of Pennsylvania, without regard to its choice of law principles. The choice of law provision in Section 7.4 herein applicable to any claim with respect to which, but for this choice of law provision, the applicable law of the Claimant's Jurisdiction pursuant to Section 5.3(b)(2) is determined to be the Alabama Wrongful Death Statute, shall only govern the rights between the PI Trust and the claimant including, but not limited to, suits in the tort system pursuant to Section 7.6, and to the extent the PI Trust seeks recovery from any entity that provided insurance to OC or Fibreboard, the Alabama Wrongful Death Statute shall govern.

7.5 Interest.

7.5(a) In General. Except for PI Trust Claims involving Other Asbestos

Disease (Disease Level I – Cash Discount Payment) and subject to the limitations set forth

below, interest shall be paid on all PI Trust Claims with respect to which the claimant has had to

wait a year or more for payment, provided, however, that no claimant shall receive interest

for a period in excess of seven (7) years. The interest rate for each year shall be the coupon

issue yield equivalent (as determined by the Secretary of the Treasury) of the average accepted auction price for the first auction of 5-year Treasury Notes occurring in such year.

- Scheduled Value of any unliquidated PI Trust Claims. Interest shall be payable on the Scheduled Value of any unliquidated PI Trust Claim that meets the requirements of Disease Levels II V, VII and VIII, whether the claim is liquidated under Expedited Review, Individual Review, or by arbitration. No interest shall be paid on any claim involving Disease Level I, or on any claim liquidated in the tort system pursuant to Section 5.11 above and Section 7.6 below. Interest on an unliquidated PI Trust Claim that meets the requirements of Disease Level VI shall be based on the Average Value of such a claim. Interest on all such unliquidated claims shall be measured from the date of payment back to the earliest of the date that is one year after the date on which (a) the claim was filed against OC or Fibreboard prior to the Petition Date; (b) the claim was filed against another defendant in the tort system on or after the Petition Date but before the Effective Date; or (c) the claim was filed with the PI Trust after the Effective Date.
- 7.5(c) Interest on Liquidated Pre-Petition Claims. Interest shall also be payable on the liquidated value of all Pre-Petition Liquidated Claims described in Section 5.2(a) above. In the case of Pre-Petition Liquidated Claims liquidated by verdict or judgment, interest shall be measured from the date of payment back to the date that is one year after the date that the verdict or judgment was entered. In the case of Pre-Petition Liquidated Claims liquidated by a binding, judicially enforceable settlement, interest shall be measured from the date of payment back to the date that is one year after the Petition Date.
- 7.6 Suits in the Tort System. If the holder of a disputed claim disagrees with the PI

 Trust's determination regarding the Discase Level of the claim, the claimant's exposure history

 or the liquidated value of the claim, and if the holder has first submitted the claim to non-binding

arbitration as provided in Section 5.10 above, the holder may file a lawsuit in the Claimant's Jurisdiction as defined in Section 5.3(b)(2) above. Any such lawsuit must be filed by the claimant in her or her own right and name and not as a member or representative of a class, and no such lawsuit may be consolidated with any other lawsuit. All defenses (including, with respect to the PI Trust, all defenses which could have been asserted by OC or Fibreboard) shall be available to both sides at trial; however, the PI Trust may waive any defense and/or concede any issue of fact or law. If the claimant was alive at the time the initial pre-petition complaint was filed or on the date the proof of claim was filed with the PI Trust, the case will be treated as a personal injury case with all personal injury damages to be considered even if the claimant has died during the pendency of the claim.

7.7 Payment of Judgments for Money Damages. If and when an OC or Fibreboard claimant obtains a judgment in the tort system, the claim shall be placed in the relevant FIFO Payment Queue based on the date on which the judgment became final. Thereafter, the claimant shall receive from the OC or Fibreboard Sub-Account an initial payment (subject to the applicable Payment Percentage, the Maximum Available Payment, and the Claims Payment Ratio provisions set forth above) of an amount equal to one-hundred percent (100%) of the greater of (i) the PI Trust's last offer to the claimant or (ii) the award that the claimant declined in non-binding arbitration. The claimant shall receive the balance of the judgment, if any, in five equal installments in years six (6) through ten (10) following the year of the initial payment (also subject to the applicable Payment Percentage, the Maximum Available Payment and the Claims Payment Ratio provisions set forth above in effect on the date of the payment of the subject installment).

In the case of non-Extraordinary claims involving Disease Levels II - VIII, the total amounts paid with respect to such claims shall not exceed the Maximum Values for such Disease

Levels set forth in Section 5.3(b)(4). In the case of Extraordinary Claims, the total amounts paid with respect to such claims shall not exceed the Maximum Value for such claims set forth in Section 5.4(a) above. In the case of claims involving Disease Level I, the total amounts paid shall not exceed the Scheduled Value of such claims. Under no circumstances shall interest be paid pursuant to Section 7.5 or under any statute on any judgments obtained in the tort system pursuant to Sections 5.11 and 7.6 above.

- 7.8 Releases. The Trustees shall have the discretion to determine the form and substance of the releases to be provided to the PI Trust in order to maximize recovery for claimants against other tortfeasors without increasing the risk or amount of claims for indemnification or contribution from the PI Trust. As a condition to making any payment to a claimant, the PI Trust shall obtain a general, partial, or limited release as appropriate in accordance with the applicable state or other law. If allowed by state law, the endorsing of a check or draft for payment by or on behalf of a claimant may, in the discretion of the PI Trust, shall constitute such a release.
- 7.9 Third-Party Services. Nothing in this TDP shall preclude the PI Trust from contracting with another asbestos claims resolution organization to provide services to the PI Trust so long as decisions about the categorization and liquidated value of PI Trust Claims are based on the relevant provisions of this TDP, including the Disease Levels, Scheduled Values, Average Values, Maximum Values, and Medical/Exposure Criteria set forth above.
- 7.10 PI Trust Disclosure of Information. Periodically, but not less often than once a year, the PI Trust shall make available to claimants and other interested parties, the number of claims by Disease Levels that have been resolved both by the Individual Review Process and by

arbitration as well as by litigation in the tort system, indicating the amounts of the awards and the averages of the awards by jurisdiction.

SECTION VIII

Miscellaneous

- 8.1 Amendments. Except as otherwise provided herein, the Trustees may amend, modify, delete, or add to any provisions of this TDP (including, without limitation, amendments to conform this TDP to advances in scientific or medical knowledge or other changes in circumstances), provided they first obtain the consent of the TAC and the Future Claimants' Representative pursuant to the Consent Process set forth in Sections 5.7(b) and 6.6(b) of the PI Trust Agreement, except that the right to amend the Claims Payment Ratio is governed by the restrictions in Section 2.5 above, and the right to adjust the Payment Percentage is governed by Section 4.2 above. Nothing herein is intended to preclude the TAC or the Future Claimants' Representative from proposing to the Trustees, in writing, amendments to this TDP. Any amendment proposed by the TAC or the Future Claimants' Representative shall remain subject to Section 7.3 of the PI Trust Agreement.
- 8.2 Severability. Should any provision contained in this TDP be determined to be unenforceable, such determination shall in no way limit or affect the enforceability and operative effect of any and all other provisions of this TDP. Should any provision contained in this TDP be determined to be inconsistent with or contrary to OC's or Fibreboard's obligations to any insurance company providing insurance coverage to OC and/or Fibreboard in respect of claims for personal injury based on exposure to asbestos-containing products manufactured or produced by OC or Fibreboard, the PI Trust with the consent of the TAC and the Future Claimants' Representative may amend this TDP and/or the PI Trust Agreement to make the provision of

either or both documents consistent with the duties and obligations of OC or Fibreboard to said insurance company.

8.3 Governing Law. Except for purposes of determining the liquidated value of any PI Trust Claim, administration of this TDP shall be governed by, and construed in accordance with, the laws of the State of Delaware. The law governing the liquidation of PI Trust Claims in the case of Individual Review, arbitration or litigation in the tort system shall be the law of the Claimant's Jurisdiction as described in Section 5.3(b)(2) above.

EXHIBIT I

ATTACHMENT A

OWENS CORNING/FIBREBOARD ASBESTOS PERSONAL INJURY TRUST

ALTERNATIVE DISPUTE RESOLUTION PROCEDURES

OWENS CORNING/FIBREBOARD ASBESTOS PERSONAL INJURY TRUST

ALTERNATIVE DISPUTE RESOLUTION PROCEDURES

Pursuant to Section 5.10 of the Owens Corning/Fibreboard Asbestos Personal Injury Trust Distribution Procedures (the "TDP"), the Owens Corning/Fibreboard Asbestos Personal Injury Trust (the "Pl Trust") hereby establishes the following alternative dispute resolution ("ADR") procedures. All capitalized terms herein shall be as defined and/or referenced within the TDP.

I. OVERVIEW

The PI Trust shall appoint a Private Adjudication Center, at the cost of the PI Trust, to administer the ADR proceedings. To initiate these procedures, the claimant must make a written request to the PI Trust. Within twenty (20) days of a claimant's request for ADR, the PI Trust will send the claimant an ADR packet containing the documents necessary to pursue the ADR process. The ADR procedures shall not be construed as imparting to any claimant any substantive or procedural rights beyond those conferred by the TDP.

The ADR process available to the claimant includes both non-binding and binding elements. In addition, there are mandatory as well as voluntary options that can/will be utilized by the claimant and the PI Trust in proceeding toward settlement. As a general matter, the ADR procedures must be pursued by claimants on an individual basis. As a general matter, claims of different claimants cannot be grouped together even if the claimants are represented by the same counsel, unless the PI Trust, in its sole discretion, decides it would be expeditious to conduct ADR proceedings with respect to more than one claim involving differently exposed claimants with those claimants' representative. In such a case, however, the arbitrator, mediator or other neutral party must individually value each such claim using the valuation factors set forth in Section 5.3(b)(2) of the TDP, and the claimants' positions in the PI Trust's FIFO Processing and Payment Queues must be separately maintained. The requisite steps in the process are as follows, in order:

Mandatory ADR Proceedings (Two Stages)

Stage One (Claimant Must Select One):

- I. Pro Bono Evaluation
- Mediation

Stage Two: Arbitration (Binding or Non-Binding)

Initiation of ADR

Within twenty (20) days of a claimant's request for ADR, the PI Trust will send the claimant an ADR packet containing a copy of these procedures and the following:

- A Summary Outline of the ADR procedures with the time limits identified;
- 2. Form Affidavit of Completeness:
- 3. Election Form for Pro Bono Evaluation:
- Request for Mediation Form;
- Election Form and Agreement to submit to Binding Arbitration; and
- Election Form and Agreement to submit to Non-binding Arbitration.

A claimant who wishes to proceed through the ADR process must engage in one of the two ADR options (pro bono evaluation or mediation) before any form of arbitration. Only after either party rejects a non-binding arbitration award, may a claimant proceed to then commence a lawsuit in the tort system. It is the claimant's responsibility to comply with the ADR time deadlines. Although the deadlines may be extended by agreement or for cause shown, failure to comply with a deadline without obtaining an extension may result in withdrawal of the claim. Promptly after a claimant fails to comply with a specified deadline without obtaining an extension, the PI Trust shall send the claimant written notice of the failure to comply. If the claimant does not take any action on the claim, then thirty (30) days thereafter the claim will be deemed withdrawn under Section 6.3 of the TDP.

If the claimant requests arbitration, either binding or non-binding, then the PI Trust shall execute the appropriate election form and agreement. If the claimant requests binding arbitration, then the claimant and the PI Trust waive their respective rights to seek a jury trial as set forth in the TDP upon execution of the Agreement for Binding Arbitration.

If either party rejects a non-binding arbitration award, and the claimant has otherwise complied with the requirements of these ADR/Arbitration procedures and the Plan, then the claimant may commence a lawsuit against the Pl Trust in the Claimant's Jurisdiction as that term is defined in Section 5.3(b)(2) of the TDP.

II. ADR PROCEEDINGS SUMMARY

A. Showing Required

As set forth in the TDP, in order to establish a valid PI Trust Claim, a claimant must among other things make a demonstration of exposure to OC and/or Fibreboard asbestos-containing products.

B. Pro Bono Evaluation

This ADR alternative consists of an evaluation of the claim by an evaluator selected from a pro bono panel. The panel shall be comprised of asbestos litigation attorneys as mutually agreed upon between the PI Trust Advisory Committee (the "TAC") and the PI Trustees. The TAC will be provided, on a quarterly basis, with a list of the pro bono panelists. Each evaluation will have a pro bono evaluator randomly selected by the Private Adjudication Center from the

list of pro bono panelists. Within fifteen (15) days of the claimant's request for the pro bono evaluation, the individual pro bono evaluator shall be randomly chosen from the approved panel.

A pro bono evaluation will be done by document submission. The identity of the pro bono evaluator will not be disclosed to the claimant and the claimant's attorney. The PI Trust encourages identification of and not anonymity as to the alleged injured party so that medical records can be transmitted in their original form. The Private Adjudication Center will communicate to the parties the pro bono evaluator's written evaluation. The parties will communicate their respective rejection or acceptance of settlement upon the terms of the written evaluation. If either or both parties reject settlement upon those terms, then the claimant may submit an Election Form and Agreement for Binding or Non-binding Arbitration.

C. Mediation

The claimant may request telephone mediation as an ADR alternative. This process will require detailed written submissions to familiarize the mediator with the respective positions. The PI Trust shall establish and maintain a list of qualified regional mediators compensated by the PI Trust. The Private Adjudication Center shall select a qualified mediator from the list based upon location of claimant within fifteen (15) days after receipt of the Request for Mediation Form signed by the claimant and the PI Trust.

Claims shall be handled by each mediator in the order received by him or her, to the extent practicable. Any party may be represented by legal counsel. The mediator shall review the claim and the positions of the parties, such information as the parties may wish to submit as to a fair and equitable settlement, and all documents and medical reports relevant to the claim as submitted by the parties. At least five (5) business days prior to the mediation conference, claimant and the PI Trust shall each submit to the mediator a detailed written submission consisting of a confidential statement outlining the claimant's medical condition, exposure to OC and/or Fibreboard products and each party's detailed position on overall claim value.

The mediator shall confer with the parties and/or their legal representatives, individually and jointly. Such conference shall be conducted by telephone unless both parties agree otherwise. A representative of the PI Trust with settlement authority must participate in the conference. The mediator may request, but not require, that the claimant personally participate in the conference. Such conference shall be in the nature of a settlement conference. The mediator shall work with both sides toward reaching an acceptable, reasonable settlement. The mediator does not have the authority to impose a settlement on the parties. Ten (10) days after the conclusion of the mediation, if the parties have not settled the

matter, the claimant may submit to the PI Trust an Election Form and Agreement for Binding or Non-binding Arbitration.

D. Binding and Non-binding Arbitration Procedures

Upon completion of either pro bono evaluation or mediation, the claimant may request non-binding and/or binding arbitration. Binding arbitration will be conducted in the "final offer" format also known as "baseball style" arbitration. If the claim is arbitrated in either the binding or non-binding format, then the arbitrator shall return an award no greater than the Maximum Value for the category in which the claim properly falls in the Tables set forth in the TDP, unless the Extraordinary Claims Panel has previously determined that the claim should receive extraordinary claim treatment. In that case, the arbitrator shall return an award no greater than the Maximum Value for such claim as set forth in Section 5.4(a) of the TDP.

If the claimant requests arbitration, either binding or non-binding, then the PI Trust shall execute the appropriate Election Form and Agreement. The PI Trust may not decline the claimant's election of either binding or non-binding arbitration, but reserves all rights to reject any award in a non-binding arbitration proceeding. If the parties agree to engage in binding arbitration, then the claimant and the PI Trust waive their respective rights to seek a jury trial as set forth in the TDP.

III. RULES GOVERNING PRO BONO EVALUATION AND MEDIATION

Within ninety (90) days of a claimant's receipt of the ADR packet from the Pl Trust, the claimant must elect one of the two ADR procedures and return the appropriate form to the Pl Trust along with an executed Affidavit of Completeness.

A. Rules Governing Pro Bono Evaluation

1. Election and Time Limits

- a. If the claimant chooses pro bone evaluation, then within ninety (90) days of claimant's receipt of the ADR packet, the claimant must send the PI Trust the Election Form for Pro Bone Evaluation. (See Attachment A). The claimant or his/her attorney shall personally sign the Election Form for Pro Bone Evaluation.
- b. The claimant must also sign an Affidavit of Completeness (See Attachment B) and return it to the PI Trust with a copy to the Private Adjudication Center within ninety (90) days of receipt of the ADR packet. The claim will not proceed until the PI Trust has received a completed Election Form and Affidavit of Completeness from the claimant. The Affidavit of Completeness shall verify that

all information to be considered in the ADR process has been provided to the Pl Trust while the claim was under review by the Pl Trust.

- c. After receiving the signed Election Form and Affidavit of Completeness, the PI Trust shall review and sign the Election Form within five (5) business days of receipt.
- d. Within fifteen (15) days from the date the Pl Trust notifies the claimant's counsel of the Pl Trust's consent to the Election Form, the Pl Trust shall send a copy of the signed Election Form, the Affidavit of Completeness together with complete copies of all materials submitted to the Pl Trust by the claimant and factual information in the Pl Trust file, if any, gathered by the Pl Trust from other sources, and a completed Affidavit of Accuracy to the claimant's counsel and the Private Adjudication Center who will forward the materials to the selected pro bono evaluator at the time the evaluator is selected. The Pl Trust may not send the Private Adjudication Center any materials in the Pl Trust file that have not previously been provided to the claimant.

2. Selection of the Pro Bono Evaluator

Within fifteen (15) days of the date the Private Adjudication Center receives the claimant's election agreement, the Private Adjudication Center shall randomly select the pro bono evaluator from the list of pro bono panelists and notify the parties that the evaluator has been designated without disclosing the identity of the evaluator. The pro bono evaluator shall be selected from a panel of asbestos litigation plaintiff attorneys who have volunteered to serve the PI Trust at the request of the TAC. Pro bono assignments will be made on a rotating basis.

The identity of the pro bono evaluator shall not be disclosed to the claimant and the claimant's attorney. The injured party should not be anonymous so that medical records can be transmitted in their original form.

3. Submission of Written Arguments

Fifteen (15) days after the Pl Trust sends the complete file materials to the Private Adjudication Center, the claimant and the Pl Trust shall simultaneously exchange and submit written arguments to the Private Adjudication Center. The Private Adjudication Center will immediately forward the written arguments to the pro bono evaluator. The written arguments shall comply with the following rules:

- a. The argument shall not exceed ten (10) double spaced typewritten pages. In order to preserve anonymity in a pro bono evaluation, the name of counsel should not be mentioned. The argument may not introduce factual matter not contained in the documents in the PI Trust's file. The evaluator shall disregard any argument that does not comply with this rule.
- b. When a party fails to submit the written argument within the fifteen (15) days, the party waives written argument and the pro bono evaluator shall disregard any argument received after that time.

4. Evaluation of Documents

The pro bono evaluation is only a document review with complete anonymity preserved between claimant's counsel and the pro bono evaluator. The documents that the pro bono evaluator may consider shall be limited to the following:

- The documents in PI Trust's file forwarded to the pro bono evaluator.
- The claimant's Affidavit of Completeness and the PI Trust's Affidavit of Accuracy.
- c. The written arguments of the claimant and the PI Trust that comply with the rules for written arguments set forth above.
- Before the Private Adjudication Center forwards any documents to the pro bono evaluator it will redact all references to claimant's counsel.

5. Written Evaluation and Procedure for Acceptance/Rejection

Within fifteen (15) days after the submission of written arguments, the probono evaluator shall submit a written evaluation of the claim to the Private Adjudication Center who will promptly mail it to the parties.

Within fifteen (15) days after receipt of the pro bono evaluator's written evaluation, the claimant and the PI Trust will each communicate in writing to the Private Adjudication Center whether they will accept the amount of the pro bono evaluator's written evaluation to settle the claim. If both parties accept, then the Private Adjudication Center will immediately inform both parties that they have achieved a settlement and the PI Trust shall pay the claim pursuant to the TDP. If either or both parties reject the pro bono evaluator's written evaluation, then within five (5) business days

of receipt of both parties' written communication, the Private Adjudication Center shall send each party a notice of rejection of pro bono evaluator's written evaluation that will not indicate whether the opposing party has accepted or rejected the pro bono evaluator's written evaluation amount.

Arbitration May Proceed After Rejection of Pro Bono Evaluator's Written Evaluation

Within sixty (60) days after receipt of the notice of rejection of pro bono evaluator's written evaluation, the claimant may request arbitration by returning to the PI Trust a signed Election Form and Agreement for either Binding or Non-binding Arbitration.

B. Rules Governing Mediation

1. Election

If the claimant chooses mediation, then the claimant shall submit to the PI Trust a signed Request for Mediation Form (Attachment C) along with an executed Affidavit of Completeness within ninety (90) days of claimant's receipt of the ADR packet. Within five (5) business days of the PI Trust's receipt of the signed Request for Mediation Form, the PI Trust shall review and sign the form and forward a signed copy along with an executed Affidavit of Accuracy to the claimant and the Private Adjudication Center.

2. Selection of Mediator

Within fifteen (15) days of the signed Request for Mediation Form, the Private Adjudication Center shall retain a mediator from the approved list of mediators. The Private Adjudication Center shall select the mediator based upon the region in which the claimant is located. The mediator shall be compensated by the Pl Trust. The Private Adjudication Center shall schedule a mediation conference within sixty (60) days after receipt of the signed Request for Mediation Form. The mediation will be conducted by telephone conference unless the parties agree otherwise. Scheduling of the conference shall be coordinated with the mediator and the conferences shall take place in the order received by the mediator, to the extent practicable.

3. Submission of Materials to Mediator

At least five (5) business days prior to the mediation conference, the claimant and the PI Trust shall each submit to the mediator a detailed written submission consisting of a confidential statement outlining the claimant's medical condition, exposure to OC and/or Fibreboard products,

and each party's position on overall claim value. The parties may also submit to the mediator documents and medical reports that they believe are relevant to the claim. The mediator shall review the claim and the positions of the parties and the other information that the parties submit prior to the mediation conference. The mediation briefs shall comply with the following rules:

- a. The confidential statement should not exceed ten (10) double spaced typewritten pages exclusive of attachments.
- b. The submission may not introduce factual matter not contained in the documents in the PI Trust's file as certified by the Affidavit of Completeness.

4. Mediation Conference

Any party may be represented by legal counsel at the mediation conference. The mediator shall confer with the parties' legal representatives and, if the claimant is present and consents, with the claimant. A representative of the PI Trust with settlement authority must participate in the conference. The mediator may request, but not require, that the claimant personally participate in the conference.

5. Negotiations at the Mediation Conference

The mediator may facilitate settlement in any manner the mediator believes is appropriate. The mediator will help the parties focus on their underlying interests, explore resolution alternatives and develop settlement options. The mediator will decide when to hold joint conferences, and when to confer separately with each party.

The parties are expected to initiate and convey to the mediator proposals for settlement. Each party shall provide a rationale for any settlement terms proposed. Finally, if the parties fail to develop mutually acceptable settlement terms, before terminating the procedure, and only with the consent of the parties, (a) the mediator may submit to the parties a final settlement proposal; and (b) if the mediator believes he/she is qualified to do so, the mediator may give the parties an evaluation (which if all parties choose, and the mediator agrees, may be in writing) of the likely outcome of the case if it were tried to final judgment, subject to any limitations under the Plan, the TDP and ethical codes.

6. Confidentiality of Mediation

The entire mediation process is confidential. Unless agreed among all the parties or required to do so by law, the parties and the mediator shall not

disclose to any person who is not associated with participants in the process, including any judicial officer, any information regarding the process (including pre-process exchanges and agreements), contents (including written and oral information), settlement terms or outcome of the proceeding.

Under this procedure, the entire process is a compromise negotiation subject to Federal Rule of Evidence 408 and all state counterparts, together with any applicable statute protecting the confidentiality of mediation. All offers, promises, conduct and statements, whether oral or written, made in the course of the proceeding by any of the parties, their agents, employees, experts and attorneys, and by the mediator are confidential.

Such offers, promises, conduct and statements are privileged under any applicable mediation privilege and are inadmissible and not discoverable for any purpose, including impeachment, in litigation between the parties. However, any written or oral information or other materials submitted to the mediator by either the PI Trust or the claimant may be submitted by either party to the arbitrator in an arbitration that takes place under these ADR procedures.

In addition, evidence that is otherwise admissible or discoverable shall not be rendered inadmissible or non-discoverable solely as a result of its presentation or use during the mediation. The exchange of any tangible material shall be without prejudice to any claim that such material is privileged or protected as work-product within the meaning of Federal Rule of Civil Procedure 26 and all state and local counterparts.

The mediator and any documents and information in the mediator's possession will not be subpoensed in any such investigation, action or proceeding, and all parties will oppose any effort to have the mediator or documents subpoensed. The mediator will promptly advise the parties of any attempt to compel him/her to divulge information received in mediation.

7. Submission of Written Offers After Mediation

At the conclusion of the mediation, the mediator shall require the parties to exchange written settlement offers that shall remain open for ten (10) days. If after the expiration of that ten (10) day period neither party accepts the other's written offer or the parties do not otherwise settle the matter, then the claimant may request binding or non-binding arbitration by sending to the PI Trust the appropriate signed Election Form and Agreement for either Binding or Non-binding Arbitration.

IV. RULES GOVERNING NON-BINDING AND BINDING ARBITRATION

A. Election by the Claimant

The Pi Trust shall review the Election Form and Agreement for Binding or Nonbinding Arbitration (Attachments D and E) and within five (5) business days of receipt the Pl Trust shall sign the Agreement and shall immediately send a fully signed Arbitration Agreement to the Private Adjudication Center.

B. Selection of the Arbitrator

- As soon as reasonably possible after the receipt of the signed Arbitration Agreement, but no more than fifteen (15) days after the receipt of the signed Arbitration Agreement, the Private Adjudication Center shall select three potential arbitrators from a rotating list kept by the Private Adjudication Center. Assignments of arbitrators will be made on a rotating basis nationally, by the Private Adjudication Center. The Private Adjudication Center shall promptly notify the arbitrators and the parties of the potential arbitrators' selection. If a potential arbitrator is unable or unwilling to serve, then a replacement selection will be made prior to notifying the PI Trust and the claimant of the potential arbitrators selected.
- Within seven (7) days of receipt of the list of potential arbitrators, the Pl Trust may select, and identify to the Private Adjudication Center, one potential arbitrator to be stricken from the list. The Private Adjudication Center shall then promptly notify the claimant of the Pl Trust's selection, whereupon, within seven (7) days of the receipt of such notification, the claimant may select, and identify to the Private Adjudication Center, a second potential arbitrator to be stricken from the list. The Private Adjudication Center shall then notify all parties which potential arbitrator remains and will conduct the arbitration. If either the Pl Trust or the claimant, or both, fails to exercise the right to strike an arbitrator from the list of potential arbitrators, the Private Adjudication Center shall appoint from those potential arbitrators remaining the arbitrator next in rotation on the Pl Trust's rotating list.
- 3. Any appointed arbitrator shall disclose to the Private Adjudication Center any circumstances likely to affect impartiality, including any bias or any financial or personal interest in the result of the arbitration or any past or present relationship with the parties or representatives. Upon receipt of such information from the arbitrator or another source, the Private Adjudication Center shall communicate the information to the parties and, if the administrator deems necessary, to the arbitrator and others. Upon objection of a party to the continued service, the Private Adjudication

Center shall determine whether the arbitrator should be disqualified and shall inform the parties of the decision, which shall be final.

C. Extraordinary Claims and Those Reviewed by the Extraordinary Claims Panel

In the event that the Extraordinary Claims Panel has deemed the claim worthy of extraordinary treatment, the Private Adjudication Center shall forward to the arbitrator the written decision of the Extraordinary Claims Panel, and the parties may submit a final request that exceeds the values ascribed to the type of injury in the TDP. In such circumstances, the arbitrator may issue an award in accordance with such a final offer/request.

In the event that the Extraordinary Claims Panel declined to give extraordinary treatment to the claim, the arbitrator shall not be informed of the Extraordinary Claims Panel's decision, and the claimant must confine his/her award to the values ascribed to the type of injury in the TDP because the arbitrator may not award an amount in excess of the Maximum Value assigned to the appropriate category for the injury in the TDP. The PI Trust will not engage in non-binding or binding arbitration, and reserves the unilateral right to withdraw from a signed non-binding or binding arbitration agreement at any time, where the claimant's final offer and award demand exceeds the Maximum Value assigned to the type of injury in the TDP and the Extraordinary Claims Panel has declined to give extraordinary treatment to the claim.

D. Final Offer or "Baseball Style" Binding Arbitration

All binding arbitration shall be conducted in the "final offer" format also known as "baseball style" arbitration. In the course of submitting the arbitration materials, as explained in these rules, the parties shall submit their final offer of settlement which shall also serve as the party's demand for arbitration award. The arbitrator must choose from one of these two demands in determining the amount of the arbitration award.

E. Submission of Pre-Hearing Statements

Within twenty (20) days of the appointment of an arbitrator each party shall submit to the opposing party and to the arbitrator a written statement (not to exceed ten (10) double spaced pages) containing that party's positions and arguments. Each party may then submit a supplement to its written statement (not to exceed five (5) double spaced pages) following the initial pre-hearing conference to respond to the opposing party's positions and arguments and addressing issues raised at the initial pre-hearing conference. Supplements must be sent to the opposing party and to the arbitrator within ten (10) days after the date of the pre-hearing conference.

The Private Adjudication Center will provide the arbitrator with a complete schedule of categories of injuries and Scheduled, Average and Maximum Values therefor in the TDP.

F. Initial Pre-Bearing Conference, Scheduling Hearing Date, Optional Video Conference for Arbitration Hearing

- Within fifteen (15) calendar days of the receipt of both party's briefs, the
 Private Adjudication Center shall contact the claimant, the arbitrator, and
 the PI Trust to schedule the initial pre-hearing conference. The prehearing conference shall be presided over by the arbitrator and held by
 telephone conference call.
- 2. During the initial pre-hearing conference, the arbitrator shall schedule the date and select the location of the arbitration hearing either at the location of the arbitrator or a location mutually agreeable to the parties. The arbitration hearing should be scheduled not less than forty-five (45) days, and not more than sixty (60) days, from the date of the initial pre-hearing conference. The Private Adjudication Center will mail a confirmation notice of this date to the claimant and the PI Trust.
- 3. At the election of the claimant, the arbitration hearing may be conducted by video conference. If the claimant so elects, then the claimant must state that election in writing prior to the initial pre-hearing conference. The Private Adjudication Center will make appropriate arrangements for the PI Trust and the arbitrator to participate by video conference. The PI Trust shall pay for its and the arbitrator's cost for use of video conference equipment and facilities. The claimant shall only be responsible for his/her costs (including participation by claimant's counsel).
- During the initial pre-trial conference, the arbitrator shall seek to achieve agreement between the parties on:
 - a. narrowing the issues (through methods including but not limited to stipulation of facts);
 - whether the claimant will appear at the hearing (at the claimant's sole discretion);
 - c. any legal issues;
 - d. and any other matters that will expedite the arbitration proceedings.

If appropriate or if the parties do not agree on these issues, then the arbitrator must issue orders governing the process.

G. No Discovery With Limited Exceptions

There shall be no discovery except as specifically provided below. The purpose of the arbitration is to resolve differences between the PI Trust and the claimant based only on the documents that have been previously submitted to the PI Trust by the claimant and any other documents relied upon by the PI Trust to make a settlement offer to the claimant or to disallow the claim. However, if the PI Trust commissions an independent medical examination or a third-party medical review upon which the PI Trust relies in evaluating the claimant's claim, then the claimant may depose the medical professional conducting the review or examination after having a reasonable opportunity to study any report or written opinion generated by the medical professional.

H. No Record of Proceedings Unless Requested by Arbitrator

There will be no record or transcript of the proceedings unless the arbitrator requests a transcript to assist him/her in reviewing the evidence or otherwise to aid in the decision making process. In the event an arbitrator requests a transcript prior to the arbitration, then the PI Trust shall arrange for a court reporter and shall pay all expenses associated with the preparation of the transcript. In no event, however, will the transcript be made available to the parties, nor shall any time required for preparation of the transcript affect the time for the arbitrator to render a decision.

I. Postponement of Hearing

The arbitrator for good cause may postpone any hearing upon the request of a party or upon the arbitrator's own initiative, and shall also grant such postponement when all of the parties agree.

J. Duration of Hearings

The arbitrator shall complete the hearing in one day except for good cause shown. The arbitrator shall set time limits on the respective presentations, and shall enforce those set limits. The parties shall request no more than three hours apiece for presentation of their cases.

K. Procedure at Arbitration Hearing

1. Testimony Under Oath or Affirmation

If the claimant or any other witness testifies, such testimony shall be under oath or affirmation administered by the arbitrator.

2. Conduct of Hearing

At the opening of the arbitration hearing, the arbitrator shall make a written record of the time, place, and date of the hearing, and the presence of the parties and counsel.

3. Evidence

- a. Rules of Evidence: The arbitrator is not required to apply the rules of evidence used in judicial proceedings, provided, however that the arbitrator shall apply the attorncy-client privilege and the work product privilege. The arbitrator shall determine the applicability of any privilege or immunity and the admissibility, relevance, materiality and weight of the evidence offered.
- b. Admission of Evidence: The evidence that the arbitrator may consider shall be limited to the following:
 - The documents supplied to the PI Trust prior to the execution of the Affidavit of Completeness;
 - (ii) Non-binding or binding arbitration election agreement;
 - (iii) Testimony of the claimant. The claimant may offer evidence regarding the nature and extent of compensable damages, including physical injuries, and/or the market share of OC and/or Fibreboard products, if there is a claim of greater than average market share. The PI Trust may cross-examine on these issues. At the claimant's option, a claimant's deposition, including videotaped testimony, shall be admissible into evidence in lieu of live testimony.
 - (iv) Any additional deposition testimony taken by the Pl Trust or the claimant, and provided to both sides, prior to the initiation of ADR.
 - (v) Any evidence submitted in mediation.
 - (vi) Closing arguments of the claimant and the Pl Trust. The arguments shall be limited to the evidence contained and the issues raised in the documents or testimony referred to above and shall be limited to ½ hour for each party. The arbitrator shall disregard any effort to introduce further evidence or issues in argument.

L. Arbitration in the Absence of a Party or Representative

The claimant may choose whether or not to attend the arbitration in person in his/her sole discretion. The arbitration may proceed in the absence of any party or representative who, after due notice, chooses not to be present, fails to be present or fails to obtain a postponement if he/she desires to be present but cannot. An award shall not be made against a party solely for the failure to appear. The arbitrator shall require the party who is present to submit such evidence as the arbitrator may require for the making of an award.

M. Conclusion of Hearing and Submission of Post-Hearing Briefs

When the parties state that they have no further evidence or witnesses to offer, and after the parties have made their closing arguments, if any, the arbitrator shall declare the hearing closed. Post-hearing briefs will be permitted only upon order of the arbitrator and shall be served upon the arbitrator no later than ten (10) days after the hearing is closed. Such briefs shall be no longer than five (5) double spaced pages. The time limit within which the arbitrator is required to make the award shall commence to run upon the closing of the hearing or the submission of post-hearing briefs, whichever is later.

N. Option to Waive Oral Hearings

The parties may request a waiver of oral hearings. Oral hearings will only be waived if all parties consent.

O. Arbitration Decision

- The arbitrator shall issue a decision no later than fifteen (15) calendar days
 after the date of the close of the hearing or submission of post-hearing
 briefs, whichever is later.
- 2. The decision shall state only the amount of the award, if any. The decision shall not state reasons for the award. An arbitrator shall not be permitted to award punitive, exemplary, trebled or other like damages or attorneys' fees, and prejudgment and post-judgment interest and costs shall not be sought or allowed. The award shall dispose of all monetary claims presented to the arbitrator and shall determine fully the only issue to be decided pursuant to the arbitration agreement: the amount, if any, at which the claim value should be fixed. To assist the arbitrator, the Private Adjudication Center will provide the arbitrator with a schedule setting forth the Disease Levels and the Scheduled, Average and Maximum Values associated with each category. Unless the Extraordinary Claims Panel has determined that a claim is entitled to extraordinary treatment during the claims review process, the arbitrator's award shall not exceed the Maximum Value amount for the appropriate Category in the TDP.

P. Payment of Award

Pursuant to the terms of the arbitration agreement, the Pl Trust will promptly send to the claimant the appropriate release. The Pl Trust will then pay the claim based upon the binding or, if accepted by both parties, the non-binding award, in accordance with the TDP in effect at that time.

Q. Rejection of Non-binding Award

1. A party in a non-binding arbitration proceeding that wishes to reject the award must notify the other party within thirty (30) days from the date a non-binding award is issued. If no rejection is received or sent by the PI Trust, then the decision will stand and the award will be deemed accepted by both parties and the PI Trust will promptly send to the claimant the appropriate release. The PI Trust will then pay the claim in accordance with the Claim Resolution Procedures in effect at that time.

2. Procedure for Rejected Award

a. Rejection by Claimant

If claimant has sent the PI Trust timely notification of rejection of a non-binding award and wishes to pursue the claim, then the claimant must notify the PI Trust through correspondence postmarked no later than sixty (60) days from the date of the non-binding award. If notification is received within the sixty (60) day deadline and claimant wishes to pursue the claim, then the PI Trust will within fifteen (15) days of receipt of this notification send the claimant an authorization to commence litigation.

b. Rejection by PI Trust

If the PI Trust rejects the non-binding award, then claimant may elect binding arbitration or request that the PI Trust forward the authorization to commence litigation.

V. GENERAL ADR PROCEDURES GOVERNING PRO BONO EVALUATION, MEDIATION, NON-BINDING ARBITRATION, AND BINDING ARBITRATION

A. ADR Submissions

The claimant's submissions (with the exception of the binding arbitration's written argument) will be reviewed by the ADR administrator before they are submitted to the pro bono evaluator, mediator or arbitrator. If they contain materials not previously submitted in support of the claim, then the PI Trust claims department will review the additional information and determine the effect, if any, it would have on the PI Trust's evaluation of the claim. In appropriate situations, a new offer may be made to the claimant.

If an attorney or other agent represents the claimant, both the attorney and the claimant must also sign the Election and Agreement for Binding Arbitration. The attorney or agent may not sign in place of, or for, the claimant unless the claimant is incapacitated, incompetent or deceased and the attorney or agent has been designated legally to act on the claimant's behalf. Documentation of this legal designation will be required.

B. No Grouping or Bundling of Claims

As a general matter, there shall be no grouping or bundling of claims by separate claimants at any stage of the ADR or arbitrations even if the claims are related and/or the claimants have the same counsel. Each claimant must proceed individually through the ADR and arbitration processes with all claims that claimant may have or represent. This provision is intended to separate claims of different exposed persons and has no effect upon multiple claims brought by a claimant's representative, such as heirs of a deceased worker. However, the PI Trust, in its sole discretion, may decide that it would be expeditious to allow the conduct of arbitration proceeding with respect to more than one claim of different exposed persons, provided that the arbitrator individually values each such claim in accordance with the valuation factors set forth in Section 5.3(b)(2) of the TDP, and the respective claimants' separate positions in the PI Trust's FIFO Processing and Payment Queues are maintained.

C. No Ex Parte Communication

There shall be no ex parte communication between the arbitrator or pro bono evaluator and any counsel or party in any matter. All correspondence between the arbitrator or pro bono evaluator and the parties will be facilitated by the Private Adjudication Center.

D. Claims and Defenses

All available claims and defenses which exist under the law subject to the claimant's election under the TDP shall be available to both sides.

E. Costs of ADR

ADR expenses

The PI Trust will pay the arbitrator's fee for non-binding or binding arbitration up to two thousand dollars (\$2000.00) per claim depending on the length of the hearing. The pro bono evaluator is a volunteer and thus no fee will be incurred. The PI Trust will assume costs of meeting and hearing facilities for arbitration. Claimants will pay their costs and attorney fees, including any expenses incurred should the claimant testify.

2. Filing Fee.

No filing fee is required of the claimant for any ADR selection, unless the PI Trust with the consent of the TAC and the Future Claimants' Representative decide that it would be in the best interests of the PI Trust and its beneficiaries to adopt such a fee.

F. Waiver of Objection to Rules Infraction

Either party who continues with the pro bono evaluation, mediation, non-binding arbitration, or binding arbitration proceeding after knowing that any provision or requirement of the applicable rules has not been complied with, and who fails to state a timely objection in writing to the arbitrator, mediator or pro bono evaluator, shall be deemed to have waived the right to object. A timely objection by a claimant must be stated in writing and mailed to the PI Trust with instructions to forward the objection to the Private Adjudication Center and to the arbitrator, mediator or pro bono evaluator. A timely objection by the PI Trust will be mailed to the claimant and to the Private Adjudication Center with instructions to forward to the arbitrator, mediator or pro bono evaluator.

G. Serving of Notices and Other Papers

Each party to the ADR and arbitration agreements shall be deemed to have consented that any papers, notices, or processes necessary or proper for the initiation or continuation of ADR and Arbitration proceedings under these rules may be served upon such party as follows:

- 1. By regular U.S. mail or overnight courier addressed to such party or their attorneys at their last known address;
- 2. By facsimile transmission or electronic mail, if a copy of the transmitted papers is mailed addressed to the party or their attorney at their last known address within twenty-four (24) hours of the facsimile transmission or electronic mail; or,
- By personal service, within or without the state where the pro bono
 evaluation, mediation or arbitration is to be held, whether the party is
 within or without the United States of America.

H. Time Limits Triggered Upon Receipt

 Documents sent by U.S. mail under these rules shall be deemed received three (3) business days after the date of postmark. Documents sent via overnight mail shall be deemed received on the next business day after mailing. Documents sent via facsimile transmission or electronic mail shall be deemed received on the business day that the transmission is sent.

I. Exclusion of Liability

Neither the Private Adjudication Center nor the mediator, nor the arbitrator nor pro bono evaluator shall be liable to any party for any act or omission in connection with any evaluation conducted under these rules.

J. Relationship of Rules to Election Form for Pro Bono Evaluation, Request for Mediation, Non-binding Arbitration Agreement or Binding Arbitration Agreement

These Rules shall be deemed a part of, and incorporated by reference in, every duly executed ADR agreement or arbitration agreement and shall be binding on all parties.

K. Arbitrator/Mediator/Pro Bono Evaluator Immunity

Arbitrators, mediators or pro bono evaluators who serve pursuant to these rules shall have the same immunity as judges for their official acts.

L. Jurisdiction

Any dispute under these rules shall be subject to the jurisdiction of the United States Bankruptcy Court for the District of Delaware.

M. Statement of Confidentiality

- 1. All ADR and arbitration proceedings and information relating to the proceeding will be confidential. Neither party shall disclose the information obtained during the proceedings, nor the valuation placed on the case by an arbitrator or pro bono evaluator, to anyone or use such information or valuation in any further proceeding except as necessary to maintain the PI Trust's obligation to report to the Bankruptcy Court and to provide ongoing evaluation by the PI Trust and TAC. Except for documents prepared by a non-party which are introduced as evidence before an arbitrator or pro bono evaluator, any document prepared by another party, attorney or other participant in anticipation of the ADR is privileged and shall not be disclosed to any court or arbitrator/pro bono evaluator or construed for any purpose as an admission against interest.
- 2. All ADR and arbitration proceedings shall be deemed a settlement conference pursuant to Rule 408 of the Federal Rules of Evidence. Except by agreement of the parties, the parties will not introduce into evidence in any other proceedings the fact that there was an arbitration, the nature or

amount of the award, and written submissions may not be used for purposes of showing accord and satisfaction or res judicata. In binding arbitration, the decision of the arbitrator may be admissible in the event the claimant improperly seeks to litigate the claim. The binding arbitration award shall be admissible in support of a motion to enjoin such litigation. No arbitrator or pro bono evaluator will ever be subpoensed or otherwise required by any party or any third party, to testify or produce records, notes or work product in any future proceedings.

N. Amendments

Except as otherwise ruled by the Bankruptcy Court, these rules, as they may from time to time be amended by the PI Trustees with the consent of the TAC and the Future Claimants' Representative, will be binding on all parties in the form in which they are in force on the date the claimant signs the election agreement.

O. Time Limits

The time limits included in these procedures are to be strictly enforced. Any time limit set forth herein may be extended by agreement of the parties or for cause shown to the neutral party presiding over the particular ADR or arbitration proceeding. Any request for extension, however, shall first be made to the opposing party and then if the parties cannot agree, shall be submitted to the Private Adjudication Center who will request a ruling from the pro bono evaluator, mediator, or arbitrator as the case may be.

Although the deadlines may be extended by agreement or for cause shown, failure to comply with a deadline without obtaining an extension may result in withdrawal of the claim. Promptly after a claimant fails to comply with a specified deadline without obtaining an extension, the PI Trust shall send the claimant written notice of the failure to comply. If the claimant does not take any action on the claim, then thirty (30) days thereafter the claim will be deemed withdrawn under Section 6.3 of the TDP.

EXHIBIT J

EXHIBIT D-1

OWENS CORNING/FIBREBOARD

ASBESTOS PERSONAL INJURY TRUST DISTRIBUTION PROCEDURES

Revised February 2, 2010

EXHIBIT D-1

OWENS CORNING/FIBREBOARD

ASBESTOS PERSONAL INJURY TRUST DISTRIBUTION PROCEDURES

TABLE OF CONTENTS

		<u>Page</u>
SECTION I	— Introduction	1
1.1	Purpose	1
1.2	Interpretation	1
SECTION I	I — Overview	2
2.1	PI Trust Goals	2 3
2.2	Claims Liquidation Procedures	3
2.3	Application of the Payment Percentage	6
2.4	Determination of the Maximum Annual Payment and	
	Maximum Available Payment	8
2.5	Claims Payment Ratio	9
2.6	Indemnity and Contribution Claims	12
SECTION I	II — TDP Administration	12
3.1	PI Trust Advisory Committee and Future Claimants'	
	Representative	12
3.2	Consent and Consultation Procedures	13
SECTION I	V — Payment Percentage; Periodic Estimates	13
		1.5
4.1	Uncertainty of OC's and Fibreboard's Total Personal Injury Asbestos Liabilities	13
4.2	Computation of Payment Percentage	13
4.3	Applicability of the Payment Percentage	
₩.5	Applicating of the Layment referitage	15

ECTION V	V — Resolution of PI Trust Claims
5.1	Ordering, Processing and Payment of Claims
	(a) Ordering of Claims
	(1) Establishment of the FIFO Processing Queues
	(2) Effect of Statutes of Limitations and Repose
	(b) Processing of Claims
	(c) Payment of Claims
5.2	Resolution of Pre-Petition Liquidated PI Trust Claims
	(a) Processing and Payment
	(b) Marshalling of Security
5.3	Resolution of Unliquidated PI Trust Claims
	(a) Expedited Review Process
	(1) In General
	(2) Claims Processing under Expedited Review
	(3) Disease Levels, Scheduled Values
	and Medical/Exposure Criteria
	(b) Individual Review Process
	(1) In General
	(A) Review of Medical/Exposure Criteria
	(B) Review of Liquidated Value
	(2) Valuation Factors to be Considered in
	Individual Review
	(3) Processing and Payment Limitations for Claims
	Involving Disease Levels III and II
	(A) Disease Level III Claims
	(B) Disease Level II Claims
5.4	(4) Scheduled, Average and Maximum Values
5.4	Categorizing Claims as Extraordinary
	and/or Exigent Hardship
	(a) Extraordinary Claims
~ ~	(b) Exigent Hardship Claims
5.5	Secondary Exposure Claims
5.6	Indirect PI Trust Claims
5.7	Evidentiary Requirements
	(a) Medical Evidence
	(1) In General
	(A) Disease Levels I – IV
	(B) Disease Levels V – VIII
	(C) Exception to the Exception for Certain
	Pre-Petition Claims
	(2) Credibility of Medical Evidence
	(b) Exposure Evidence
	(1) In General
	(2) Significant Occupational Exposure
	(3) OC or Fibreboard Exposure

	5.9	Second Disease (Malignancy) Claims
	5.10	Arbitration
		(a) Establishment of ADR Procedures
		(b) Claims Eligible for Arbitration
		(c) Limitations on and Payment of Arbitration Awards
	5.11	Litigation
SECT	ION VI	— Claims Materials
	6.1	Claims Materials
	6.2	Content of Claims Materials
	6.3	Withdrawal or Deferral of Claims
	6.4	Withdrawal or Deferral of Claims
	6.5	Filing Requirements and Fees
	0.3	Confidentiality of Claimants' Submissions
SECT	ION VI	I — General Guidelines for Liquidating and Paying Claims
	7.1	Showing Required
	7.2	Costs Considered
	7.3	Discretion to Vary Order and Amounts of Payments in
		Event of Limited Liquidity
	7.4	Punitive Damages
	7.5	Sequencing Adjustment
		(a) In General
		(b) Unliquidated PI Trust Claims
		(c) Liquidated Pre-Petition Claims
	7.6	Suits in the Tort System
	7.7	Payment of Judgments for Money Damages
	7.8	Releases
	7.9	Third-Party Services
	7.10	PI Trust Disclosure of Information
~		
SECT	ION VI	II — Miscellaneous
	8.1	Amendments
	8.2	Severability
	8.3	Governing Law

OWENS CORNING/FIBREBOARD

ASBESTOS PERSONAL INJURY TRUST DISTRIBUTION PROCEDURES

The Asbestos Personal Injury Trust Distribution Procedures ("TDP") contained herein provide for resolving all Asbestos Personal Injury Claims for which Owens Corning ("OC") and/or its wholly owned subsidiary, Fibreboard Corporation ("Fibreboard"), and their predecessors, successors, and assigns have legal responsibility (respectively, OC Asbestos Personal Injury Claims ("OC Claims") and FB Asbestos Personal Injury Claims ("Fibreboard Claims"), which terms are defined in the Sixth Amended Joint Plan of Reorganization for Owens Corning and its Affiliated Debtors and Debtors-in-Possession (As Modified) ("Plan") (hereinafter collectively referred to in this TDP as "PI Trust Claims"). The Plan and the Asbestos Personal Injury Trust Agreement ("PI Trust Agreement") establish the Owens Corning/Fibreboard Asbestos Personal Injury Trust (the "PI Trust"). The Trustees of the PI Trust ("Trustees") shall implement and administer this TDP in accordance with the PI Trust Agreement. Capitalized terms used herein and not otherwise defined shall have the meanings assigned to them in the Plan and the PI Trust Agreement.

SECTION I

Introduction

- 1.1 Purpose. This TDP has been adopted pursuant to the PI Trust Agreement. It is designed to provide fair, equitable, and substantially similar treatment for all PI Trust Claims that may presently exist or may arise in the future.
- 1.2 Interpretation. Except as may otherwise be provided below, nothing in this TDP shall be deemed to create a substantive right for any claimant. The rights and benefits, if

any, provided herein to holders of PI Trust Claims shall vest in such holders as of the Effective Date.

SECTION II

Overview

2.1 PI Trust Goals. The goal of the PI Trust is to treat all holders of PI Trust Claims equitably and in accordance with the requirements of Section 524(g) of the Bankruptcy Code. To achieve that goal, the PI Trust consists of two separate Sub-Accounts, an OC Sub-Account for payment of OC Claims and a Fibreboard Sub-Account for payment of Fibreboard Claims (together the "PI Trust Sub-Accounts").

A claimant may assert separate claims against the OC Sub-Account and the Fibreboard Sub-Account based on separate exposures to asbestos or asbestos-containing products manufactured or distributed by OC and Fibreboard, respectively ("Multiple Exposure Claims"); however, all such Multiple Exposure Claims must be filed by the claimant at the same time. To the extent that the OC Sub-Account and the Fibreboard Sub-Account each has separate liability to a claimant based on Multiple Exposure Claims, each Sub-Account shall pay the claimant the liquidated value of the separate claim for which it is liable, subject to applicable Payment Percentage, Maximum Annual Payment, Maximum Available Payment and Claims Payment Ratio limitations set forth below.

This TDP sets forth procedures for processing and paying all PI Trust Claims from the two Sub-Accounts generally on an impartial, first-in-first-out ("FIFO") basis, with the intention of paying all claimants over time as equivalent a share as possible of the value of their claims

based on historical values for substantially similar claims in the tort system. To this end, this TDP establishes a single schedule of eight asbestos-related diseases ("Disease Levels"), seven of which have presumptive medical and exposure requirements ("Medical/Exposure Criteria") that are applicable to both OC and Fibreboard Claims, as well as two separate schedules with liquidated values ("Scheduled Values"), anticipated average values ("Average Values"), and caps on liquidated values ("Maximum Values") that are applicable to OC Claims and Fibreboard Claims, respectively.

These Disease Levels, Medical/Exposure Criteria, Scheduled Values, Average Values and Maximum Values, which are set forth in Sections 5.3 and 5.4 below, have all been selected and derived with the intention of achieving a fair allocation of the assets held by the separate OC and Fibreboard Sub-Accounts as among their respective claimants suffering from different disease processes in light of the best available information considering the settlement histories of OC and Fibreboard, and the rights that OC and Fibreboard claimants would have in the tort system absent the bankruptcy.

2.2 Claims Liquidation Procedures. PI Trust Claims shall be processed based on their place in separate FIFO Processing Queues to be established for each of the two PI Trust Sub-Accounts pursuant to Section 5.1(a) below. The PI Trust shall take all reasonable steps to resolve OC and Fibreboard Claims as efficiently and expeditiously as possible at each stage of claims processing and arbitration. To this end, the PI Trust, in its sole discretion, may conduct settlement discussions with claimants' representatives with respect to more than one claim at a time, provided that the claimants' respective positions in the FIFO Processing Queues are

As used in this TDP, the phrase "in the tort system" shall not include claims asserted against a trust established pursuant to section 524(g) and/or section 105 of the Bankruptcy Code or any other applicable law.

maintained, and each claim is individually evaluated pursuant to the valuation factors set forth in Section 5.3(b)(2) below. The PI Trust shall also make every effort to resolve each year at least that number of PI Trust Claims required to exhaust the Maximum Annual Payment and the Maximum Available Payment for Category A and Category B claims, as those terms are defined below.

The PI Trust shall liquidate all OC and Fibreboard Claims except Foreign Claims (as defined in Section 5.3(b)(1) below) that meet the presumptive Medical/Exposure Criteria of Disease Levels I – V, VII and VIII under the Expedited Review Process described in Section 5.3(a) below. PI Trust Claims involving Disease Levels I – V, VII and VIII that do not meet the presumptive Medical/Exposure Criteria for the relevant Disease Level may undergo the PI Trust's Individual Review Process described in Section 5.3(b) below. In such a case, notwithstanding that the claim does not meet the presumptive Medical/Exposure Criteria for the relevant Disease Level, the PI Trust can offer the claimant an amount up to the Scheduled Value of that Disease Level if the PI Trust is satisfied that the claimant has presented a claim that would be cognizable and valid in the tort system.

In lieu of liquidating such claimant's claim under the Expedited Review Process, OC and Fibreboard claimants holding PI Trust Claims involving Disease Levels II - VIII may alternatively seek to establish liquidated values for their claims that are greater than their Scheduled Values by electing the PI Trust's Individual Review Process. However, the liquidated values of PI Trust Claims that undergo the Individual Review Process for valuation purposes may be determined to be less than the Scheduled Values, and in any event shall not exceed the respective Maximum Values for the Disease Levels set forth for OC and Fibreboard Claims in Section 5.3(b)(4) below, unless the claims qualify as Extraordinary Claims as defined

in Section 5.4(a) below, in which case their liquidated value cannot exceed the Maximum Values specified in that provision for such claims. OC and Fibreboard Level VI (Lung Cancer 2) Claims and all Foreign Claims may be liquidated² only pursuant to the PI Trust's Individual Review Process.

Based upon OC 's and Fibreboard's claims settlement history in light of applicable tort law, and current projections of present and future unliquidated claims, the Scheduled Values and Maximum Values set forth in Section 5.3(b)(4) for OC and Fibreboard Claims, respectively, have been established for each of the Disease Levels that are eligible for Individual Review of their liquidated values, with the expectation that the combination of settlements at the Scheduled Values and those resulting from the Individual Review Process should result in the Average Values also set forth in that provision.

All unresolved disputes over a claimant's medical condition, exposure history and/or the liquidated value of the claim shall be subject to mandatory *pro bono* evaluation and mediation and then to binding or non-binding arbitration pursuant to Section 5.10 below, at the election of the claimant, under the ADR Procedures that are provided in Attachment A hereto. PI Trust Claims that are the subject of a dispute with the PI Trust that cannot be resolved by non-binding arbitration may enter the tort system as provided in Sections 5.11 and 7.6 below. However, if and when an OC or Fibreboard claimant obtains a judgment in the tort system, the judgment will be payable (subject to the Payment Percentage, Maximum Available Payment, and Claims Payment Ratio provisions set forth below) as provided in Section 7.7 below.

2.3 Application of the Payment Percentage. After the liquidated value of an OC or Fibreboard Claim other than a claim involving Other Asbestos Disease (Disease Level I – Cash

² For purposes of this TDP, "liquidated" means approved and valued by the PI Trust.

Discount Payment), as defined in Section 5.3(a)(3) below, is determined pursuant to the procedures set forth herein for Expedited Review, Individual Review, arbitration, or litigation in the tort system, the claimant will ultimately receive a pro-rata share of that value based on the Payment Percentages separately set for OC and Fibreboard Claims pursuant to Section 4.2 below. These Payment Percentages shall also apply to all Pre-Petition Liquidated Claims as provided in Section 5.2 below and to all sequencing adjustments pursuant to Section 7.5 below.

The Initial Payment Percentage for the OC Sub-Account has been set at forty percent (40%), and the Initial Payment Percentage for the Fibreboard Sub-Account has been set at twenty-five percent (25%). These Initial Payment Percentages shall apply to all OC and Fibreboard PI Trust Voting Claims accepted as valid by the PI Trust, unless adjusted by the PI Trust with the consent of the PI Trust Advisory Committee ("TAC") and the Legal Representative for Future Asbestos Claimants ("Future Claimants' Representative") (who are described in Section 3.1 below) pursuant to Section 4.2 below, and except as provided in Section 4.3 below with respect to supplemental payments in the event an Initial Payment Percentage for a Sub-Account is changed.

The term "PI Trust Voting Claims" includes (i) Pre-Petition Liquidated Claims as provided in Section 5.2 below; (ii) OC and Fibreboard Claims filed against OC and/or Fibreboard in the tort system or actually submitted to OC and/or Fibreboard pursuant to an administrative settlement agreement prior to the Petition Date of October 5, 2000; and (iii) all asbestos claims filed against another defendant in the tort system prior to the date the Plan was first filed with the Bankruptcy Court (January 17, 2003 (the "Plan Filing Date")), provided, however, that (1) the holder of a claim described in subsection (i), (ii) or (iii) above, or his or her authorized agent, actually voted to accept or reject the Plan pursuant to the voting procedures

established by the Bankruptcy Court unless such holder certifies to the satisfaction of the Trustees that he or she was prevented from voting in this proceeding as a result of circumstances resulting in a state of emergency affecting, as the case may be, the holder's residence, principal place of business or legal representative's principal place of business at which the holder or his or her legal representative receives notice and/or maintains material records relating to his or her PI Trust Voting Claim, and (2) the claim was subsequently filed with the PI Trust pursuant to Section 6.1 below by the Initial Claims Filing Date as defined in Section 5.1(a) below.

The Initial Payment Percentages for the OC and Fibreboard Sub-Accounts set forth above have been calculated on the assumption that the Average Values set forth in Section 5.3(b)(4) below will be achieved with respect to existing present claims and projected future claims involving Disease Levels II – VIII. However, either or both of these Payment Percentages may be adjusted upwards or downwards from time to time pursuant to Section 4.2 below by the PI Trust with the consent of the TAC and the Future Claimants' Representative to reflect then-current estimates of the assets and liabilities allocable to OC and Fibreboard Claims, respectively, as well as the then-estimated value of pending and future OC and Fibreboard Claims. However, any adjustment to the Initial Payment Percentages shall be made only pursuant to Section 4.2 below. If the Payment Percentage for either the OC or Fibreboard Sub-Account is increased over time, claimants whose OC or Fibreboard Claims were liquidated and paid in prior periods under the TDP will receive additional payments only as provided in Section 4.3 below. Because there is uncertainty in the prediction of both the number and severity of future claims, and the amount of the PI Trust's assets, no guarantee can be made of any Payment Percentage for either OC or Fibreboard Claims.

Payment. For each of the OC and the Fibreboard Sub-Accounts, the PI Trust shall estimate or model the amount of cash flow anticipated to be necessary over the entire life of the Sub-Account to ensure that amounts will be available to treat all holders of OC and/or Fibreboard Claims as similarly as possible, given the assets and liabilities allocable to each of the two Sub-Accounts. In each year, for each Sub-Account, the PI Trust will be empowered to pay out all of the income earned during the year by the Sub-Account (net of taxes payable with respect thereto), together with a portion of the Sub-Account's principal, calculated so that the application of the Sub-Account's assets over its life shall correspond with the needs created by the anticipated flow of claims to the Sub-Account (the "Maximum Annual Payment"), taking into account the Payment Percentage provisions set forth in Section 2.3 above and Sections 4.2 and 4.3 below. The PI Trust's distributions from each Sub-Account to all holders of claims against the Sub-Account for that year shall not exceed the Maximum Annual Payment determined for that year.

In distributing the Maximum Annual Payment from each Sub-Account, the PI Trust shall first allocate the amount in question to outstanding Pre-Petition Liquidated Claims (as defined in Section 5.2(a) below) against the Sub-Account, and to liquidated claims against the Sub-Account involving Disease Level I (Cash Discount Payment), in proportion to the aggregate value of each group of claims. The remaining portion of the Maximum Annual Payment (the "Maximum Available Payment"), if any, shall then be allocated and used to satisfy all other previously liquidated PI Trust Claims against the Sub-Account, subject to the Claims Payment Ratio for the Sub-Account set forth in Section 2.5 below.

In the event there are insufficient amounts in any year to pay the total number of outstanding Pre-Petition Liquidated Claims and/or previously liquidated Disease Level I Claims against the Sub-Account, the available amounts allocated to that group of claims shall be paid to the maximum extent to claimants in the particular group based on their place in their Sub-Account's FIFO Payment Queue. Claims in either group for which there are insufficient amounts in the Sub-Account shall be carried over to the next year and placed at the head of the FIFO Payment Queue for that Sub-Account. If there is a decrease in the Payment Percentage prior to the payment of such claims, any such Pre-Petition Liquidated Claims shall nevertheless be entitled to be paid at the Payment Percentage that they would have been entitled to receive but for the application of the Maximum Annual Payment.

2.5 Claims Payment Ratio. Based upon OC 's and Fibreboard's claims settlement history and analysis of present and future claims, a single Claims Payment Ratio has been determined for both Sub-Accounts, which, as of the Effective Date, has been set at 65% for Category A claims, which consist of PI Trust Claims against OC and/or Fibreboard involving severe asbestosis and malignancies (Disease Levels IV – VIII) that were unliquidated as of the Petition Date, and at 35% for Category B claims, which are PI Trust Claims against OC and/or Fibreboard involving non-malignant Asbestosis or Pleural Disease (Disease Levels II and III) that were similarly unliquidated as of the Petition Date. However, the Claims Payment Ratio shall not apply to any Pre-Petition Liquidated Claims or to any claims for Other Asbestos Disease (Disease Level I - Cash Discount Payment) payable from either OC or Fibreboard Sub-Accounts.

In each year, after the determination of the Maximum Available Payment described in Section 2.4 above, 65% of that amount will be available to pay Category A claims and 35% will

be available to pay Category B claims that have been liquidated since the Petition Date. In the event there are insufficient amounts in either the OC or Fibreboard Sub-Accounts in any year to pay the liquidated claims within either or both of the Categories, the available amounts allocated to the particular Category within the Sub-Account shall be paid to the maximum extent to claimants in that Category based on their place in the Sub-Account's FIFO Payment Queue described in Section 5.1(c) below, which will be based upon the date of claim liquidation.

Claims for which there are insufficient amounts allocated to the relevant Category within a Sub-Account shall be carried over to the next year where they will be placed at the head of the Sub-Account's FIFO Payment Queue. If there is a decrease in the Payment Percentage prior to the payment of such claims, such claims shall nevertheless be entitled to be paid at the Payment Percentage that they would have been entitled to receive but for the application of the Claims Payment Ratio. If there are excess amounts in either or both Categories within a Sub-Account, because there is an insufficient amount of liquidated claims to exhaust the respective Sub-Account's Maximum Available Payment amount for that Category, then the excess amounts for either or both Categories will be rolled over and remain dedicated to the respective Category to which they were originally allocated.

The 65%/35% Claims Payment Ratio and its rollover provision shall apply to all OC and Fibreboard PI Trust Voting Claims as defined in Section 2.3 above (except Pre-Petition Liquidated Claims and Other Asbestos Claims (Disease Level I – Cash Discount Payment)) and shall not be amended until the third anniversary of the date the PI Trust first accepts for processing proof of claim forms and other materials required to file a claim with the PI Trust. Thereafter, the Sub-Account's Claims Payment Ratio and its rollover provision shall be continued absent circumstances, such as a significant change in law or medicine, necessitating

amendment to avoid a manifest injustice. However, the accumulation, rollover and subsequent delay of claims against one or both Sub-Accounts resulting from the application of the Claims Payment Ratio, shall not, in and of itself, constitute such circumstances. Nor may an increase in the numbers of Category B claims against a Sub-Account beyond those predicted or expected be considered as a factor in deciding whether to reduce the percentage allocated to Category A claims.

In considering whether to make any amendments to the Claims Payment Ratio and/or its rollover provisions for either Sub-Account, the Trustees should also consider the reasons for which the Claims Payment Ratio and its rollover provisions were adopted, the settlement histories of OC and Fibreboard that gave rise to its calculation, and the foreseeability or lack of the foreseeability of the reasons why there would be any need to make an amendment. In that regard, the Trustees should keep in mind the interplay between the Payment Percentage and the Claims Payment Ratio as it affects the net cash actually paid to claimants from either Sub-Account.

In any event, no amendment to the Claims Payment Ratio to reduce the percentage allocated to Category A claims may be made without the unanimous consent of the TAC members and the consent of the Future Claimants' Representative, and the percentage allocated to Category A claims may not be increased without the consent of the TAC and the Future Claimants' Representative. In case of any amendments to the Claims Payment Ratio, consents shall be governed by the consent process set forth in Sections 5.7(b) and 6.6(b) of the PI Trust Agreement. The Trustees, with the consent of the TAC and the Future Claimants' Representative, may offer the option of a reduced Payment Percentage to holders of claims in

either Category A or Category B against either Sub-Account in return for prompter payment by the Sub-Account (the "Reduced Payment Option").

2.6 Indemnity and Contribution Claims. As set forth in Section 5.6 below, PI Trust Claims for indemnity and contribution (defined in the Plan as OC Indirect Asbestos Personal Injury Claims and Fibreboard Indirect Asbestos Personal Injury Claims, and hereinafter referred to as "Indirect PI Trust Claims") against either the OC or the Fibreboard Sub-Accounts, if any, will be subject to the same categorization, evaluation, and payment provisions of this TDP as all other OC and Fibreboard Claims.

SECTION III

TDP Administration

3.1 PI Trust Advisory Committee and Future Claimants' Representative.

Pursuant to the Plan and the PI Trust Agreement, the PI Trust and this TDP shall be administered by the Trustees in consultation with the TAC, which represents the interests of holders of present PI Trust Claims against OC and Fibreboard, and the Future Claimants' Representative, who represents the interests of holders of PI Trust Claims against OC and/or Fibreboard that will be asserted in the future. The Trustees shall obtain the consent of the TAC and the Future Claimants' Representative on any amendments to these Procedures pursuant to Section 8.1 below, and on such other matters as are otherwise required below and in Section 2.2(f) of the PI Trust Agreement. The Trustees shall also consult with the TAC and the Future Claimants' Representative on such matters as are provided below and in Section 2.2(e) of the PI Trust Agreement. The initial members of the TAC and the initial Future Claimants' Representative are identified in the PI Trust Agreement.

3.2 Consent and Consultation Procedures. In those circumstances in which consultation or consent is required, the Trustees will provide written notice to the TAC and the Future Claimants' Representative of the specific amendment or other action that is proposed. The Trustees will not implement such amendment nor take such action unless and until the parties have engaged in the Consultation Process described in Sections 5.7(a) and 6.6(a), or the Consent Process described in Sections 5.7(b) and 6.6(b) of the PI Trust Agreement, respectively.

SECTION IV

Payment Percentage; Periodic Estimates

- 4.1 Uncertainty of OC's and Fibreboard's Total Personal Injury Asbestos

 Liabilities. As discussed above, there is inherent uncertainty regarding OC's and Fibreboard's total asbestos-related tort liabilities, as well as the total value of the assets available to the OC and Fibreboard Sub-Accounts to pay PI Trust Claims asserted against each Sub-Account.

 Consequently, there is inherent uncertainty regarding the amounts that holders of PI Trust Claims will receive. To seek to ensure substantially similar treatment of all present and future PI Trust Claims against either the OC or the Fibreboard Sub-Accounts, the Trustees must determine from time to time the percentage of full liquidated value that holders of PI Trust Claims against the Sub-Account will be likely to receive, *i.e.*, the "Payment Percentage" described in Section 2.3 above and Section 4.2 below.
- 4.2 Computation of Payment Percentage. As provided in Section 2.3 above, the Initial Payment Percentage for claims against the OC Sub-Account shall be forty percent (40%), and for claims against the Fibreboard Sub-Account twenty-five percent (25%). These percentages shall apply to all OC and Fibreboard PI Trust Voting Claims as defined in Section 2.3 above, unless the Trustees, with the consent of the TAC and the Future Claimants'

Representative, determine that the Initial Payment Percentage for one or both Sub-Accounts should be changed to assure that the PI Trust will be in a financial position to pay holders of unliquidated and/or unpaid PI Trust Voting Claims and present and future PI Trust Claims against the OC and Fibreboard Sub-Accounts, respectively, in substantially the same manner. In making any such adjustment, the Trustees, the TAC and the Future Claimants' Representative shall take into account the fact that the holders of PI Trust Voting Claims voted on the Plan relying on the findings of experts that the Initial Payment Percentage for each Sub-Account represented a reasonably reliable estimate of the PI Trust's total assets and liabilities over its life based on the best information available at the time, and shall thus give due consideration to the expectations of PI Trust Voting Claimants that the Initial Payment Percentage would be applied to their PI Trust Claims.

Except with respect to PI Trust Voting Claims to which the Initial Payment Percentage applies, the Payment Percentage for either the OC or the Fibreboard Sub-Accounts shall be subject to change pursuant to the terms of this TDP and the PI Trust Agreement if the Trustees determine that an adjustment is required. No less frequently than once every three years, commencing with the first day of January occurring after the Plan is consummated, the Trustees shall reconsider the then applicable Payment Percentage for each of the OC and Fibreboard Sub-Accounts to assure that the respective percentage is based on accurate, current information and may, after such reconsideration, change the Payment Percentage for either Sub-Account if necessary with the consent of the TAC and the Future Claimants' Representative.

The Trustees shall also reconsider the then applicable Payment Percentages for either or both Sub-Accounts at shorter intervals if they deem such reconsideration to be appropriate or if requested to do so by the TAC or the Future Claimants' Representative. The Trustees must base

their determination of the Payment Percentage on current estimates of the number, types, and values of present and future PI Trust Claims against the respective Sub-Accounts, the value of the assets then available to the respective Sub-Accounts for their payment, all anticipated administrative and legal expenses of the respective Sub-Accounts, and any other material matters that are reasonably likely to affect the sufficiency of the respective Sub-Accounts' assets to pay a comparable percentage of full value to all holders of claims against the Sub-Accounts. When making these determinations, the Trustees shall exercise common sense and flexibly evaluate all relevant factors. The Payment Percentage applicable to Category A or Category B claims against the respective Sub-Accounts may not be reduced to alleviate delays in payments of claims in the other Category; both Categories will receive the same Payment Percentage, but the payment from either or both Sub-Accounts may be deferred as needed pursuant to Section 7.3 below, and a Reduced Payment Option may be instituted for either Sub-Account as described in Section 2.5 above.

4.3 Applicability of the Payment Percentage. Except as set forth below in this Section 4.3 with respect to supplemental payments, no holder of a PI Trust Voting Claim other than a PI Trust Voting Claim for Other Asbestos Disease (Disease Level I - Cash Discount Payment) as defined in Section 5.3(a)(3) below shall receive a payment that exceeds the PI Trust's determination of the Initial Payment Percentage for the relevant Sub-Account of the liquidated value of the claim. Except as otherwise provided (a) in Section 5.1(c) below for PI Trust Claims involving deceased or incompetent claimants for which court or probate approval of the PI Trust's offer is required and (b) in the paragraph below with respect to Released Claims, no holder of any other PI Trust Claim shall receive a payment that exceeds the Payment Percentage for the respective Sub-Account in effect at the time of payment; provided, however,

that if there is a reduction in the Payment Percentage for a Sub-Account, the Trustees, in their sole discretion, may cause the PI Trust to pay a PI Trust Claim based on the Payment Percentage that was in effect prior to the reduction if such PI Trust Claim was filed and actionable with the PI Trust with respect to such Sub-Account ninety (90) days or more prior to the date the Trustees proposed the new Payment Percentage in writing to the TAC and the Future Claimants' Representative (the "Proposal Date") and the processing of such claim was unreasonably delayed due to circumstances beyond the control of the claimant or the claimant's counsel, but only if such claim had no deficiencies for the ninety (90) days prior to the Proposal Date. PI Trust Claims involving Other Asbestos Disease (Disease Level I - Cash Discount Payment) shall not be subject to such Sub-Account's Payment Percentage, but shall instead be paid the full amount of their Scheduled Value as set forth in Section 5.3(a)(3) below.

If a redetermination of the respective Sub-Account's Payment Percentage has been proposed in writing by the Trustees to the TAC and the Future Claimants' Representative but has not yet been adopted, the claimant shall receive the lower of such Sub-Account's current Payment Percentage or the proposed Payment Percentage. However, if the proposed Payment Percentage for such Sub-Account was the lower amount but was not subsequently adopted, the claimant shall thereafter receive the difference between the lower proposed amount and the higher current amount. Conversely, if the proposed Payment Percentage for such Sub-Account was the higher amount and was subsequently adopted, the claimant shall thereafter receive the difference between the lower current amount and the higher adopted amount. Notwithstanding anything contained herein, if the proposed Payment Percentage for a Sub-Account is lower than the current Payment Percentage for such Sub-Account, a claimant whose PI Trust Claim was

liquidated prior to the Proposal Date and who either (a) transmitted an executed release to the PI Trust prior to the Proposal Date or (b) with respect to those claimants who had received releases fewer than thirty (30) days prior to the Proposal Date, transmitted an executed release to the PI Trust within thirty (30) days of the claimant's receipt of the release (the claims described in (a) and (b) are collectively referred to here in as the "Released Claims") shall be paid based on the current Payment Percentage (the "Released Claims Payment Percentage"). For purposes hereof, (a) a claimant represented by counsel shall be deemed to have received a release on the date that the claimant's counsel receives the release, (b) if the PI Trust transmits a release electronically, the release shall be deemed to have been received on the date the PI Trust transmits the offer notification, and (c) if the PI Trust places the release in the U.S. mail, postage prepaid, the release shall be deemed to have been received three (3) business days after such mailing date. A delay in the payment of the Released Claims for any reason, including delays resulting from limitations on payment amounts in a given year pursuant to Sections 2.4 and 2.5 hereof, shall not affect the rights of the holders of the Released Claims to be paid based on the Released Claims Payment Percentage.

At least thirty (30) days prior to proposing in writing to the TAC and the Future Claimants' Representative a change in the Payment Percentage for a Sub-Account, the Trustees shall issue a written notice to claimants or claimants' counsel indicating that the Trustees are reconsidering such Payment Percentage. During the period of time when the Trustees are contemplating a change in the Payment Percentage for a Sub-Account, the PI Trust shall continuing processing claims and making offers in a manner consistent with its normal course of business.

³ For purposes of this sentence, "transmitted" is defined as the date/time postmarked if submitted by mail or the date/time uploaded if submitted electronically.

There is uncertainty surrounding the amount of the PI Trust's future assets. There is also uncertainty surrounding the totality of the PI Trust Claims to be paid over time as well as the extent to which changes in existing federal and state law could affect the PI Trust's liabilities under this TDP. If the value of the PI Trust's future assets increases significantly and/or if the value or volume of PI Trust Claims actually filed with the PI Trust is significantly lower than originally estimated, the PI Trust shall use those proceeds and/or claims savings, as the case may be, first to maintain the Payment Percentage then in effect.

If the Trustees, with the consent of the TAC and the Future Claimants' Representative, make a determination to increase the Payment Percentage due to a material change in the estimates of the PI Trust's future assets and/or liabilities, the Trustees shall also make supplemental payments to all claimants who previously liquidated their claims against the PI Trust and received payments based on a lower Payment Percentage. The amount of any such supplemental payment shall be the liquidated value of the claim in question times the newly adjusted Payment Percentage, less all amounts previously paid to the claimant with respect to the claim (excluding the portion of such previously paid amounts that was attributable to any sequencing adjustment paid pursuant to Section 7.5 below).

The Trustees' obligation to make a supplemental payment to a claimant shall be suspended in the event the payment in question would be less than \$100.00, and the amount of the suspended payment shall be added to the amount of any prior supplemental payment/payments that was/were also suspended because it/they would have been less than \$100.00. However, the Trustees' obligation shall resume and the Trustees shall pay any such aggregate supplemental payments due the claimant at such time that the total exceeds \$100.00.

SECTION V

Resolution of PI Trust Claims.

- 5.1 Ordering, Processing and Payment of Claims.
 - 5.1(a) Ordering of Claims.

5.1(a)(1) Establishment of FIFO Processing Queues. The PI Trust will order separately all OC and Fibreboard Claims that are sufficiently complete to be reviewed for processing purposes on a FIFO basis except as otherwise provided herein (the "FIFO Processing Queues"). For all claims filed on or before the date six months after the date that the PI Trust first makes available proof of claim forms and other claims materials required to file a claim with the PI Trust (the "Initial Claims Filing Date"), a claimant's position in either FIFO Processing Queue shall be determined as of the earliest of (i) the date prior to the Petition Date (if any) that the specific claim was either filed against OC or Fibreboard in the tort system or was actually submitted to OC or Fibreboard pursuant to an administrative settlement agreement; (ii) the date before the Petition Date that the asbestos claim was filed against another defendant in the tort system if at the time the claim was subject to a tolling agreement with OC or Fibreboard; (iii) the date after the Petition Date but before the date that the PI Trust first makes available the proof of claim forms and other claims materials required to file a claim with the PI Trust that the asbestos claim was filed against another defendant in the tort system; (iv) the date after the Petition Date but before the Effective Date the claimant filed a proof of claim form in OC's and/or Fibreboard's Chapter 11 proceeding; or (v) the date after the Petition Date the claimant submitted a ballot in OC's Chapter 11 proceeding for purposes of voting on the Plan pursuant to the voting procedures approved by the Bankruptcy Court.

Following the Initial Claims Filing Date, the claimant's position in one of the two FIFO Processing Queues shall be determined by the date the claim was filed with the PI Trust. If any claims are filed on the same date, the claimant's position in the FIFO Processing Queue shall be determined by date of the claimant's diagnosis of asbestos-related disease. If any claims are filed and diagnosed on the same date, the claimant's position in the FIFO Processing Queue shall be determined by the date of the claimant's birth, with older claimants given priority over younger claimants.

5.1(a)(2) Effect of Statutes of Limitations and Repose. All

unliquidated PI Trust Claims must meet either, (i) for claims first filed in the tort system against OC or Fibreboard, respectively prior to the Petition Date, the applicable federal, state and foreign statute of limitation and repose that was in effect at the time of the filing of the claim in the tort system, or (ii) for claims that were not filed against either OC or Fibreboard in the tort system prior to the Petition Date, the applicable federal, state or foreign statute of limitation that was in effect at the time of the filing with the PI Trust.

However, the running of the relevant statute of limitation shall be tolled as of the earliest of (A) the actual filing of the claim against OC or Fibreboard prior to the Petition Date, whether in the tort system or by submission of the claim to OC or Fibreboard pursuant to an administrative settlement agreement; (B) the tolling of the claim against OC or Fibreboard by an agreement or otherwise, provided such tolling is still in effect on the Petition Date; or (C) the Petition Date.

If a PI Trust Claim meets any of the tolling provisions described in the preceding sentence and the claim was not barred by the applicable federal, state or foreign statute of limitation at the time of the tolling event, it will be treated as timely filed if it is actually filed

with the PI Trust within three (3) years after the Initial Claims Filing Date. In addition, any claims that were first diagnosed after the Petition Date, irrespective of the application of any relevant federal, state or foreign statute of limitation or repose, may be filed with the PI Trust within three (3) years after the date of diagnosis or within three (3) years after the Initial Claims Filing Date, whichever occurs later. However, the processing of any PI Trust Claim by the PI Trust may be deferred at the election of the claimant pursuant to Section 6.3 below.

- **5.1(b) Processing of Claims.** As a general practice, the PI Trust will review its claims files on a regular basis and notify all claimants whose claims are likely to come up in either the OC or Fibreboard FIFO Processing Queue in the near future.
- Fibreboard Sub-Accounts that have been liquidated under the provisions of this TDP by the Expedited Review Process as provided in Section 5.3(a) below, by the Individual Review Process as provided in Section 5.3(b) below, by arbitration as provided in Section 5.10 below, or by litigation in the tort system provided in Section 5.11 below, shall be paid in FIFO order from the relevant Sub-Account based on the date their liquidation became final (the "FIFO Payment Queue"), all such payments being subject to the applicable Payment Percentage, the Maximum Annual Payment, the Maximum Available Payment, the Claims Payment Ratio, and the sequencing adjustment provided for in Section 7.5 below, except as otherwise provided herein. Pre-Petition Liquidated Claims, as defined in Section 5.2 below, shall be subject to the Maximum Annual Payment and Payment Percentage limitations, but not to the Maximum Available Payment and Claims Payment Ratio provisions set forth above.

Where the claimant is deceased or incompetent, and the settlement and payment of his or her claim must be approved by a court of competent jurisdiction or through a probate process prior to acceptance of the claim by the claimant's representative, an offer made by the PI Trust on the claim shall remain open so long as proceedings before that court or in that probate process remain pending, provided that the PI Trust has been furnished with evidence that the settlement offer has been submitted to such court or probate process for approval. If the offer is ultimately approved by the court or through the probate process and accepted by the claimant's representative, the PI Trust shall pay the claim from the relevant Sub-Account in the amount so offered, multiplied by the Payment Percentage in effect for such Sub-Account at the time the offer was first made.

If any claims are liquidated on the same date, the claimant's position in a Sub-Account's FIFO Payment Queue shall be determined by the date of the diagnosis of the claimant's asbestos-related disease. If any claims are liquidated on the same date and the respective holders' asbestos-related diseases were diagnosed on the same date, those claimants' positions in the Sub-Account's FIFO Payment Queue shall be determined by the PI Trust based on the dates of the claimants' birth, with older claimants given priority over younger claimants.

5.2 Resolution of Pre-Petition Liquidated PI Trust Claims.

Date, the PI Trust shall pay, upon submission by the claimant of the appropriate documentation, all PI Trust Claims that were liquidated (i) by a binding settlement agreement for the particular claim entered into prior to the Petition Date that is judicially enforceable by the claimant, (ii) after the Petition Date according to the terms of a binding settlement agreement entered into prior to the Petition Date (a "Pre-Petition Agreement"), (iii) by a jury verdict or non-final judgment in the tort system obtained prior to the Petition Date, or (iv) by a judgment that became final and non-appealable prior to the Petition Date (collectively "Pre-Petition Liquidated

Claims"). In order to receive payment from the PI Trust, the holder of a Pre-Petition Liquidated Claim must submit all documentation necessary to demonstrate to the PI Trust that the claim was liquidated in the manner described in the preceding sentence, which documentation shall include (A) a court authenticated copy of the jury verdict (if applicable), non-final judgment (if applicable) or final judgment (if applicable), and (B) the name, social security number and date of birth of the claimant and the name and address of the claimant's lawyer; provided, however, that such documentation shall not be required with respect to any Pre-Petition Liquidated Claim that OC or Fibreboard has identified to the PI Trust as a Pre-Petition Liquidated Claim as to which all conditions to payment under the applicable agreement, jury verdict or judgment have been satisfied. OC and Fibreboard shall deliver to the PI Trust a list of the Pre-Petition Liquidated Claims that OC and Fibreboard have approved for payment (the "Approved Pre-Petition Liquidated Claims"), which claims shall be entitled to rely upon the exception set forth in the preceding sentence.

The liquidated value of a Pre-Petition Liquidated Claim shall be the unpaid portion of the amount agreed to in the binding settlement agreement or Pre-Petition Agreement, the unpaid portion of the amount awarded by the jury verdict or non-final judgment, or the unpaid portion of the amount of the final judgment, as the case may be, plus interest, if any, that has accrued on that amount in accordance with the terms of a binding settlement agreement or Pre-Petition Agreement, if any, or under applicable state law for settlements or judgments as of the Petition Date; however, except as otherwise provided in Section 7.4 below, the liquidated value of a Pre-Petition Liquidated Claim shall not include any punitive or exemplary damages. In addition, the amounts payable with respect to such claims shall not be subject to or taken into account in consideration of the Claims Payment Ratio and the Maximum Available Payment limitations, but

shall be subject to the Maximum Annual Payment and Payment Percentage provisions. In the absence of a Final Order of the Bankruptcy Court determining whether a settlement agreement is binding and judicially enforceable, a dispute between a claimant and the PI Trust over this issue shall be resolved pursuant to the same procedures in this TDP that are provided for resolving the validity and/or liquidated value of a PI Trust Claim (*i.e.*, arbitration and litigation in the tort system as set forth in Sections 5.10 and 5.11 below).

The PI Trust shall pay the Approved Pre-Petition Liquidated Claims as expeditiously as possible. The other Pre-Petition Liquidated Claims shall be processed and paid from the OC and/or Fibreboard Sub-Accounts in accordance with their order in separate FIFO queues to be established for each Sub-Account by the PI Trust based on the date the PI Trust received all required documentation for the particular claim. If any Pre-Petition Liquidated Claims are filed with the PI Trust on the same date, the claimant's position in the Sub-Account's FIFO queue for such claims shall be determined by the date on which the claim was liquidated. If any Pre-Petition Liquidated Claims are both filed with the PI Trust and liquidated by a Sub-Account on the same dates, those claimants' positions in the FIFO queue shall be determined by the dates of the claimants' birth, with older claimants given priority over younger claimants.

- 5.2(b) Marshalling of Security. Holders of Pre-Petition Liquidated Claims that are secured by letters of credit, appeal bonds, or other security or sureties shall first exhaust their rights against any applicable security or surety before making a claim against the PI Trust. Only in the event that such security or surety is insufficient to pay the Pre-Petition Liquidated Claim in full shall the deficiency be processed and paid as a Pre-Petition Liquidated Claim.
- **5.3 Resolution of Unliquidated PI Trust Claims**. Within six months after the establishment of the PI Trust, the Trustees with the consent of the TAC and the Future

Claimants' Representative shall adopt procedures for reviewing and liquidating all unliquidated PI Trust Claims, which shall include deadlines for processing such claims. Such procedures shall also require claimants seeking resolution of unliquidated PI Trust claims to first file a proof of claim form, together with the required supporting documentation, in accordance with the provisions of Sections 6.1 and 6.2 below. It is anticipated that the PI Trust shall provide an initial response to the claimant within six months of receiving the proof of claim form.

The proof of claim form shall require the claimant to assert his or her OC and/or Fibreboard Claim for the highest Disease Level for which the claim qualifies at the time of filing. Irrespective of the Disease Level alleged on the proof of claim form, each OC and/or Fibreboard Claims shall be deemed to be a claim for the highest Disease Level for which the claim qualifies at the time of filing, and all lower Disease Levels for which the claim may also qualify at the time of filing or in the future shall be treated as subsumed into the higher Disease Level for both processing and payment purposes.

Upon filing of a valid proof of claim form with the required supporting documentation, the claim shall be placed in the relevant OC and/or Fibreboard FIFO Processing Queue in accordance with the ordering criteria described in Section 5.1(a) above. The PI Trust shall provide the claimant with six-months notice of the date by which it expects to reach the claim in the FIFO Processing Queue, following which the claimant shall promptly (i) advise the PI Trust whether the claim should be liquidated under the PI Trust's Expedited Review Process described in Section 5.3(a) below or, in certain circumstances, under the PI Trust's Individual Review Process described in Section 5.3(b) below; (ii) provide the PI Trust with any additional medical and/or exposure evidence that was not provided with the original claim submission; and (iii) advise the PI Trust of any change in the claimant's Disease Level. If a claimant fails to respond

to the PI Trust's notice prior to the reaching of the claim in the FIFO Processing Queue, the PI Trust will process and liquidate the claim under the Expedited Review Process based upon the medical/exposure evidence previously submitted by the claimant, although the claimant shall retain the right to request Individual Review as described in Section 5.3(b) below.

5.3(a) Expedited Review Process.

5.3(a)(1) In General. The PI Trust's Expedited Review Process is designed primarily to provide an expeditious, efficient and inexpensive method for liquidating all OC and Fibreboard Claims (except those involving Lung Cancer 2 - Disease Level VI and all Foreign Claims, which must be liquidated pursuant to the PI Trust's Individual Review process) where the claim can easily be verified by the PI Trust as meeting the presumptive Medical/Exposure Criteria for the relevant Disease Level. Expedited Review thus provides claimants with a substantially less burdensome process for pursuing PI Trust Claims than does the Individual Review Process described in Section 5.3(b) below. Expedited Review is also intended to provide qualifying claimants a fixed and certain claims payment.

Thus, claims that undergo Expedited Review and meet the presumptive

Medical/Exposure Criteria for the relevant Disease Level shall be paid the Scheduled Value (or

Values in the case of Multiple Exposure Claims) for such Disease Level set forth in Section

5.3(a)(3) below. However, except for claims involving Other Asbestos Disease (Disease Level I),
all claims liquidated by Expedited Review shall be subject to the applicable Payment Percentage,
the Maximum Annual Payment, the Maximum Available Payment, and the Claims Payment

Ratio limitations set forth herein. Claimants holding OC and/or Fibreboard Claims that cannot be
liquidated by Expedited Review because they do not meet the presumptive Medical/Exposure

Criteria for the relevant Disease Level may elect the PI Trust's Individual Review Process set forth in Section 5.3(b) below.

Subject to the provisions of Section 5.8, the claimant's eligibility to receive the Scheduled Value for his or her PI Trust Claim pursuant to the Expedited Review Process shall be determined solely by reference to the Medical/Exposure Criteria set forth below for each of the Disease Levels eligible for Expedited Review.

5.3(a)(2) Claims Processing under Expedited Review. All claimants seeking liquidation of an OC and/or Fibreboard Claim pursuant to Expedited Review shall file the PI Trust's proof of claim forms provided in Attachment B hereto. As a proof of claim form is reached in the OC or Fibreboard FIFO Processing Queue, the PI Trust shall determine whether the claim described therein meets the Medical/Exposure Criteria for one of the seven Disease Levels eligible for Expedited Review, and shall advise the claimant of its determination. If a Disease Level is determined, the PI Trust shall tender to the claimant an offer of payment from the relevant OC or Fibreboard Sub-Account of the Scheduled Value for the relevant Disease Level multiplied by the applicable Payment Percentage, together with a form of release approved by the PI Trust. If the claimant accepts the Scheduled Value and returns the release properly executed, the claim shall be placed in the Sub-Account's FIFO Payment Queue, following which the PI Trust shall disburse payment subject to the limitations of the Maximum Available Payment and Claims Payment Ratio, if any.

5.3(a)(3) Disease Levels, Scheduled Values and Medical/Exposure Criteria. The eight Disease Levels covered by this TDP, together with the Medical/Exposure Criteria for each, and the separate OC and Fibreboard Scheduled Values for the seven (7) Disease Levels eligible for Expedited Review, are set forth below. These Disease Levels,

Scheduled Values, and Medical/Exposure Criteria shall apply to all PI Trust Voting Claims (other than Pre-Petition Liquidated Claims) filed with the PI Trust on or before the Initial Claims Filing Date provided in Section 5.1 above for which the claimant elects the Expedited Review Process. Thereafter, for purposes of administering the Expedited Review Process and with the consent of the TAC and the Future Claimants' Representative, the Trustees may add to, change or eliminate Disease Levels, Scheduled Values, or Medical/Exposure Criteria; develop subcategories of Disease Levels, Scheduled Values or Medical/Exposure Criteria; or determine that a novel or exceptional asbestos personal injury claim is compensable even though it does not meet the Medical/Exposure Criteria for any of the then current Disease Levels.

Disease Level	OC/Fibreboard Scheduled Values	Medical/Exposure Criteria
Mesothelioma (Level V	VIII) \$215,000/\$135,000	(1) Diagnosis ⁴ of mesothelioma; and (2) credible evidence of OC or Fibreboard Exposure (as defined in Section 5.7(b)(3) below)
Lung Cancer 1 (Level	VII) \$40,000/\$27,000	(1) Diagnosis of a primary lung cancer plus evidence of an underlying Bilateral Asbestos- Related Nonmalignant Disease ⁵ , (2)

The requirements for a diagnosis of an asbestos-related disease that may be compensated under the provisions of this TDP are set forth in Section 5.7 below.

Evidence of "Bilateral Asbestos-Related Nonmalignant Disease" for purposes of meeting the criteria for establishing Disease Levels I, II, III, V, and VII, means either (i) a chest X-ray read by a qualified B reader of 1/0 or higher on the ILO scale or (ii)(x) a chest x-ray read by a qualified B reader or other Qualified Physician, (y) a CT scan read by a Qualified Physician, or (z) pathology, in each case showing either bilateral interstitial fibrosis, bilateral pleural plaques, bilateral pleural thickening, or bilateral pleural calcification. Evidence submitted to demonstrate (i) or (ii) above must be in the form of a written report stating the results (e.g., an ILO report, written radiology report or a pathology report). Solely for asbestos claims filed against OC or Fibreboard or another defendant in the tort system prior to the Petition Date, if an ILO reading is not available, either (i) a chest X-ray or a CT scan read by a Qualified Physician, or (ii) pathology, in each case showing either bilateral interstitial fibrosis, bilateral pleural plaques, bilateral pleural thickening, or bilateral pleural calcification, consistent with or compatible with a diagnosis of asbestos-related disease, shall be evidence of a "Bilateral Asbestos-Related"

Lung Cancer 2 (Level VI)

None

six months OC or Fibreboard Exposure prior to December 31, 1982, (3) Significant Occupational Exposure to asbestos, ⁶ and (4) supporting medical documentation establishing asbestos exposure as a contributing factor in causing the lung cancer in question.

(1) Diagnosis of a primary lung cancer; (2) OC or Fibreboard Exposure prior to December 31, 1982, and (3) supporting medical documentation establishing asbestos exposure as a contributing factor in causing the lung cancer in question.

Lung Cancer 2 (Level VI) claims are claims that do not meet the more stringent medical and/or exposure requirements of Lung Cancer (Level VII) claims. All claims in this Disease Level will be individually evaluated. The estimated likely Average Value of the individual evaluation awards for this category for OC Claims is \$20,000 and for Fibreboard Claims is \$12,000, with such awards capped at a Maximum Value of \$50,000 for OC Claims and \$30,000 for Fibreboard Claims,

Nonmalignant Disease" for purposes of meeting the presumptive medical requirements of Disease Levels I, II, III, V and VII. Pathological evidence of asbestosis may be based on the pathological grading system for asbestosis described in the Special Issue of the Archives of Pathology and Laboratory Medicine, "Asbestos-associated Diseases," Vol. 106, No. 11, App. 3 (October 8, 1982). For all purposes of this TDP, a "Qualified Physician" is a physician who is board certified (or in the case of Canadian claims or Foreign Claims, a physician who is certified or qualified under comparable medical standards or criteria of the jurisdiction in question) in one or more relevant specialized fields of medicine such as pulmonology, radiology, internal medicine or occupational medicine; provided, however, subject to the provisions of Section 5.8, that the requirement for board certification in this provision shall not apply to otherwise qualified physicians whose x-rays and/or CT scan readings are submitted for deceased holders of PI Trust Claims.

[&]quot;Significant Occupational Exposure" is defined in Section 5.7 below.

unless the claim qualifies for Extraordinary Claim treatment (discussed in Section 5.4 below).

Level VI claims that show no evidence of either an underlying Bilateral Asbestos-Related Nonmalignant Disease or Significant Occupational Exposure may be individually evaluated, although it is not expected that such claims will be treated as having any significant value, especially if the claimant is also a Smoker. In any event, no presumption of validity will be available for any claims in this category.

Other Cancer (Level V) \$ 22,000/\$12,000

(1) Diagnosis of a primary colorectal, laryngeal, esophageal, pharyngeal, or stomach cancer, plus evidence of an underlying Bilateral Asbestos-Related Nonmalignant Disease, (2) six months OC or Fibreboard Exposure prior to December 31, 1982, (3) Significant Occupational Exposure to asbestos, and (4) supporting medical documentation establishing asbestos exposure as a contributing factor in causing the other cancer in question.

Severe Asbestosis (Level IV) \$42,000/\$29,000

(1) Diagnosis of asbestosis with ILO of 2/1 or greater, or asbestosis determined by pathological evidence

There is no distinction between Non-Smokers and Smokers for either Lung Cancer (Level VII) or Lung Cancer (Level VI), although a claimant who meets the more stringent requirements of Lung Cancer (Level VII) (evidence of an underlying Bilateral Asbestos-Related Nonmalignant Disease plus Significant Occupational Exposure), and who is also a Non-Smoker, may wish to have his or her claim individually evaluated by the PI Trust. In such a case, absent circumstances that would otherwise reduce the value of the claim, it is anticipated that the liquidated value of the claim might well exceed the Scheduled Values for Lung Cancer (Level VII) claims against OC and Fibreboard, respectively, shown above. "Non-Smoker" means a claimant who either (a) never smoked or (b) has not smoked during any portion of the twelve (12) years immediately prior to the diagnosis of the lung cancer.

of asbestos, plus (a) TLC less than 65%, or (b) FVC less than 65% and FEV1/FVC ratio greater than 65%, (2) six months OC or Fibreboard Exposure prior to December 31, 1982, (3) Significant Occupational Exposure to asbestos, and (4) supporting medical documentation establishing asbestos exposure as a contributing factor in causing the pulmonary disease in question.

Asbestosis/

Pleural Disease (Level III) \$ 19,000/\$11,500

(1) Diagnosis of Bilateral Asbestos-Related Nonmalignant Disease plus (a) TLC less than 80%, or (b) FVC less than 80% and FEV1/FVC ratio greater than or equal to 65%, and (2) six months OC or Fibreboard Exposure prior to December 31, 1982, (3) Significant Occupational Exposure to asbestos, and (4) supporting medical documentation establishing asbestos exposure as a contributing factor in causing the pulmonary disease in question.

Asbestosis/
Pleural Disease (Level II) \$8,000/\$4,500

(1) Diagnosis of a Bilateral Asbestos-Related Nonmalignant Disease, and (2) six months OC or Fibreboard Exposure prior to December 31, 1982, and (3) five years cumulative occupational exposure to asbestos.

Other Asbestos Disease (Level I - Cash Discount Payment) \$ 400/\$240

(1) Diagnosis of a Bilateral Asbestos-Related Nonmalignant Disease or an asbestos-related malignancy other than mesothelioma, and (2) OC or Fibreboard Exposure prior to December 31, 1982.

5.3(b) Individual Review Process

5.3(b)(1) In General. Subject to the provisions set forth below, an OC or Fibreboard Claimant may elect to have his or her PI Trust Claim reviewed for purposes of determining whether the claim would be compensable in the tort system even though it does not meet the presumptive Medical/Exposure Criteria for any of the Disease Levels set forth in Section 5.3(a)(3) above. In addition or alternatively, an OC or Fibreboard claimant may elect to have a claim undergo the Individual Review Process for purposes of determining whether the liquidated value of the claim exceeds the Scheduled Value for the relevant Disease Level also set forth in said provision. However, until such time as the PI Trust has made an offer on a claim pursuant to Individual Review, the claimant may change his or her Individual Review election and have the claim liquidated pursuant to the PI Trust's Expedited Review Process. In the event of such a change in the processing election, the claimant shall nevertheless retain his or her place in the FIFO Processing Queue.

The liquidated value of all Foreign Claims shall be established pursuant to the PI Trust's Individual Review Process. PI Trust Claims of individuals exposed in Canada who were resident in Canada when such claims were filed ("Canadian Claims") shall not be considered Foreign Claims hereunder and shall be eligible for liquidation under the Expedited Review Process. Accordingly, a "Foreign Claim" is a PI Trust Claim with respect to which the claimant's exposure to an asbestos-containing product for which OC and or Fibreboard has legal responsibility occurred outside of the United States and its Territories and Possessions and outside of the Provinces and Territories of Canada.

In reviewing Foreign Claims, the PI Trust shall take into account all relevant procedural and substantive legal rules to which the claims would be subject in the Claimant's Jurisdiction as defined in Section 5.3(b)(2) below. The PI Trust shall determine the liquidated value of Foreign

Claims based on historical settlements and verdicts in the Claimant's Jurisdiction as well as the other valuation factors set forth in Section 5.3(b)(2) below.

For purposes of the Individual Review process, the Trustees, with the consent of the TAC and the Future Claimants' Representative, may develop separate Medical/Exposure Criteria and standards, as well as separate requirements for physician and other professional qualifications, which shall be applicable to Foreign Claims; provided, however, that such criteria, standards or requirements shall not effectuate substantive changes to the claims eligibility requirements under this TDP, but rather shall be made only for the purpose of adapting those requirements to the particular licensing provisions and/or medical customs or practices of the foreign country in question.

At such time as the PI Trust has sufficient historical settlement, verdict and other valuation data for claims from a particular foreign jurisdiction, the Trustees, with the consent of the TAC and the Future Claimants' Representative, may also establish a separate valuation matrix for such Foreign Claims based on that data.

5.3(b)(1)(A) Review of Medical/Exposure Criteria. The

PI Trust's Individual Review Process provides an OC or Fibreboard claimant with an opportunity for individual consideration and evaluation of a PI Trust Claim that fails to meet the presumptive Medical/Exposure Criteria for Disease Levels I – V, VII or VIII. In such a case, the PI Trust shall either deny the claim, or, if the PI Trust is satisfied that the claimant has presented a claim that would be cognizable and valid in the tort system, the PI Trust can offer the claimant a liquidated value amount up to the Scheduled Value for that Disease Level, unless the claim qualifies as an Extraordinary Claim as defined in Section 5.4(a) below, in which case its liquidated value cannot exceed the Maximum Value for such a claim.

5.3(b)(1)(B) Review of Liquidated Value. Claimants

holding claims involving Disease Levels II - VIII shall also be eligible to seek Individual Review of the liquidated value of their OC and Fibreboard Claims, as well as of their medical/exposure evidence. The Individual Review Process is intended to result in payments from the OC and/or Fibreboard Sub-Accounts equal to the full liquidated value for each claim multiplied by the Payment Percentage; however, the liquidated value of any OC or Fibreboard Claim that undergoes Individual Review may be determined to be less than the Scheduled Value the claimant would have received under Expedited Review. Moreover, the liquidated value for a claim involving Disease Levels II - VIII shall not exceed the Maximum Value for the relevant Disease Level set forth in Section 5.3(b)(4) below, unless the claim meets the requirements of an Extraordinary Claim described in Section 5.4(a) below, in which case its liquidated value cannot exceed the Maximum Value set forth in that provision for such claims. Because the detailed examination and valuation process pursuant to Individual Review requires substantial time and effort, claimants electing to undergo the Individual Review Process may be paid the liquidated value of their PI Trust Claims later than would have been the case had the claimant elected the Expedited Review Process. Subject to the provisions of Section 5.8, the PI Trust shall devote reasonable resources to the review of all claims to ensure that there is a reasonable balance maintained in reviewing all classes of claims.

5.3(b)(2) Valuation Factors to be Considered in Individual Review.

The PI Trust shall liquidate the value of each OC and Fibreboard Claim that undergoes

Individual Review based on the historic liquidated values of other similarly situated claims in the
tort system for the same Disease Level. The PI Trust will thus take into consideration all of the
factors that affect the severity of damages and values within the tort system including, but not

limited to credible evidence of (i) the degree to which the characteristics of a claim differ from the presumptive Medical/Exposure Criteria for the Disease Level in question; (ii) factors such as the claimant's age, disability, employment status, disruption of household, family or recreational activities, dependencies, special damages, and pain and suffering; (iii) whether the claimant's damages were (or were not) caused by asbestos exposure to an asbestos-containing product prior to December 31, 1982 for which OC or Fibreboard has legal responsibility (for example, alternative causes, and the strength of documentation of injuries); (iv) the industry of exposure; (v) settlements and verdict histories and other law firms' experience in the Claimant's Jurisdiction for similarly situated claims; and (vi) settlement and verdict histories for the claimant's law firm for similarly situated claims.

For these purposes, the "Claimant's Jurisdiction" is the jurisdiction in which the claim was filed (if at all) against OC or Fibreboard in the tort system prior to the Petition Date. If the claim was not filed against OC or Fibreboard in the tort system prior to the Petition Date, the claimant may elect as the Claimant's Jurisdiction either (i) the jurisdiction in which the claimant resides at the time of diagnosis or when the claim is filed with the PI Trust; or (ii) a jurisdiction in which the claimant experienced exposure to an asbestos-containing product for which OC or Fibreboard has legal responsibility.

With respect to the "Claimant's Jurisdiction" in the event a personal representative or authorized agent makes a claim under this TDP for wrongful death with respect to which the governing law of the Claimant's Jurisdiction could only be the Alabama Wrongful Death Statute, the Claimant's Jurisdiction for such claim shall be the Commonwealth of Pennsylvania, and such claimant's damages shall be determined pursuant to the statutory and common laws of the Commonwealth of Pennsylvania without regard to its choice of law principles. The choice of

law provision in Section 7.4 below applicable to any claim with respect to which, but for this choice of law provision, the applicable law of the Claimant's Jurisdiction pursuant to Section 5.3(b)(2) is determined to be the Alabama Wrongful Death Statute, shall only govern the rights between the PI Trust and the claimant, and, to the extent the PI Trust seeks recovery from any entity that provided insurance coverage to OC and or Fibreboard, the Alabama Wrongful Death Statute shall govern.

5.3(b)(3) Processing and Payment Limitations for Claims Involving

Disease Levels III and II. The PI Trust shall administer Individual Review for Disease Levels III and II so that Individual Review does not reduce payments to claimants electing the Scheduled Value for such PI Trust Claims under Expedited Review. As one means of implementing this requirement, the following shall apply for Disease Levels III and II claims:

5.3(b)(3)(A) Disease Level III Claims. No more than 13% or 9% of Disease Level III claims paid in any year from either the OC or the Fibreboard Sub-Account, respectively, shall be PI Trust Claims allowed under Individual Review, and the total payments to such Disease Level III claims allowed under Individual Review shall be no more than 17% or 13% of payments to all Disease Level III claimants from either the OC or Fibreboard Sub-Account, respectively, during any year.

5.3(b)(3)(B) Disease Level II Claims. No more than 15% or 20% of Disease Level II claims paid in any year from either the OC or the Fibreboard Sub-Account, respectively, shall be PI Trust Claims allowed under Individual Review, and the total payments to such Disease Level II claims allowed under Individual Review shall be no more than 24% or 33% of payments to all Disease Level II claimants from either the OC or Fibreboard Sub-Account, respectively, during any year.

5.3(b)(4) Scheduled, Average and Maximum Values. The Scheduled, Average and Maximum Values for the Disease Levels compensable under this TDP from the OC and Fibreboard Sub-Accounts are the following:

OC SUB-ACCOUNT

Scheduled Disease	Scheduled Value	Average Value	Maximum Value
Mesothelioma (Level VIII)	\$215,000	\$270,000	\$650,000
Lung Cancer 1 (Level VII)	\$ 40,000	\$ 50,000	\$150,000
Lung Cancer 2 (Level VI)	None	\$ 20,000	\$ 50,000
Other Cancer (Level V)	\$ 22,000	\$ 25,000	\$ 60,000
Severe Asbestosis (Level IV)	\$ 42,000	\$ 50,000	\$150,000
Asbestosis/Pleural Disease (Level III)	\$ 19,000	\$ 20,000	\$ 35,000
Asbestosis/Pleural Disease (Level II)	\$ 8,000	\$ 9,000	\$ 20,000
Other Asbestos Disease Cash Discount Payment (Level I)	\$ 400	None	None

FIBREBOARD SUB-ACCOUNT

Scheduled Disease	Scheduled Value	Average Value	Maximum Value
Mesothelioma (Level VIII)	\$135,000	\$180,000	\$450,000
Lung Cancerl (Level VII)	\$ 27,000	\$ 35,000	\$ 90,000
Lung Cancer 2 (Level VI)	None	\$ 12,000	\$ 30,000
Other Cancer (Level V)	\$ 12,000	\$ 15,000	\$ 36,000
Severe Asbestosis (Level IV)	\$ 29,000	\$ 30,000	\$ 90,000
Asbestosis/Pleural Disease (Level III)	\$ 11,500	\$ 12,000	\$ 21,000
Asbestosis/Pleural Disease (Level II)	\$ 4,500	\$ 5,400	\$ 12,000
Other Asbestos Disease Cash Discount Payment (Level I)	\$ 240	None	None

These OC and Fibreboard Scheduled Values, Average Values and Maximum Values shall apply to all PI Trust Voting Claims other than Pre-Petition Liquidated Claims filed with the PI Trust on or before the Initial Claims Filing Date as provided in Section 5.1 above. Thereafter, the PI Trust, with the consent of the TAC and the Future Claimants' Representative pursuant to Sections 5.7(b) and 6.6(b) of the PI Trust Agreement, may change these valuation amounts for good cause and consistent with other restrictions on the amendment power.

5.4 Categorizing Claims as Extraordinary and/or Exigent Hardship

5.4(a) Extraordinary Claims. "Extraordinary Claim" means a PI Trust Claim that otherwise satisfies the Medical Criteria for Disease Levels II - VIII, and that is held by a claimant whose exposure to asbestos (i) occurred predominately as the result of working in a manufacturing facility of OC or Fibreboard during a period in which OC or Fibreboard was manufacturing asbestos-containing products at that facility, or (ii) was at least 75% the result of

exposure to an asbestos-containing product for which OC or Fibreboard has legal responsibility, and in either case there is little likelihood of a substantial recovery elsewhere. All such Extraordinary Claims shall be presented for Individual Review and, if valid, shall be entitled to an award of up to a Maximum Value of five (5) times the Scheduled Value for claims qualifying for Disease Levels II – V, VII and VIII, and five (5) times the Average Value for claims in Disease Level VI, multiplied by the applicable Payment Percentage.

Any dispute as to Extraordinary Claim status shall be submitted to a special Extraordinary Claims Panel to be established by the PI Trust with the consent of the TAC and the Future Claimants' Representative. All decisions of the Extraordinary Claims Panel shall be final and not subject to any further administrative or judicial review. An Extraordinary Claim, following its liquidation, shall be placed in the Trust's FIFO Queue ahead of all other PI Trust Claims except Pre-Petition Liquidated Claims, Disease Level I (Other Asbestos Disease) Claims and Exigent Hardship Claims, which shall be paid first in that order in said Queue, based on its date of liquidation and shall be subject to the Maximum Available Payment and Claims Payment Ratio described above.

pay PI Trust Claims that qualify as Exigent Hardship Claims as defined below. Such claims may be considered separately no matter what the order of processing otherwise would have been under this TDP. An Exigent Hardship Claim, following its liquidation, shall be placed first in the relevant Sub-Account's FIFO Payment Queue ahead of all other liquidated claims except Pre-Petition Liquidated Claims and Disease Level I (Other Asbestos Disease) Claims, and shall be subject to the Maximum Available Payment and Claims Payment Ratio described above. A PI Trust Claim qualifies for payment as an Exigent Hardship Claim if the claim meets the

Medical/Exposure Criteria for Severe Asbestosis (Disease Level IV) or an asbestos-related malignancy (Disease Levels V-VIII), and the PI Trust, in its sole discretion, determines (i) that the claimant needs financial assistance on an immediate basis based on the claimant's expenses and all sources of available income, and (ii) that there is a causal connection between the claimant's dire financial condition and the claimant's asbestos-related disease.

- resulting solely from exposure to an occupationally exposed person, such as a family member, the claimant must seek Individual Review of his or her OC and/or Fibreboard Claim pursuant to Section 5.3(b) above. In such a case, the claimant must establish that the occupationally exposed person would have met the exposure requirements under this TDP that would have been applicable had that person filed a direct claim against the PI Trust. In addition, the claimant with secondary exposure must establish that he or she is suffering from one of the eight Disease Levels described in Section 5.3(a)(3) above or an asbestos-related disease otherwise compensable under this TDP, that his or her own exposure to the occupationally exposed person occurred within the same time frame as the occupationally exposed person was exposed to asbestos products produced by OC or Fibreboard, and that such secondary exposure to OC or Fibreboard products was a cause of the claimed disease. The proof of claim form included in Attachment B hereto contains an additional section for Secondary Exposure Claims. All other liquidation and payment rights and limitations under this TDP shall be applicable to such claims.
- 5.6 Indirect PI Trust Claims. Indirect PI Trust Claims asserted against either the OC or Fibreboard Sub-Accounts based upon theories of contribution or indemnification under applicable law, shall be treated as presumptively valid and paid by the PI Trust subject to the applicable Payment Percentage if (a) such claim satisfied the requirements of the Bar Date for

such claims established by the Bankruptcy Court, if applicable, and is not otherwise disallowed by Section 502(e) of the Code or subordinated under Section 509(c) of the Code, and (b) the holder of such claim (the "Indirect Claimant") establishes to the satisfaction of the Trustees that (i) the Indirect Claimant has paid in full the liability and obligation of the Trust to the individual claimant to whom the PI Trust would otherwise have had a liability or obligation under these Procedures (the "Direct Claimant"), (ii) the Direct Claimant and the Indirect Claimant have forever and fully released the Trust from all liability to the Direct Claimant, and (iii) the claim is not otherwise barred by a statute of limitation or repose or by other applicable law. In no event shall any Indirect Claimant have any rights against the PI Trust superior to the rights of the related Direct Claimant against the PI Trust, including any rights with respect to the timing, amount or manner of payment. In addition, no Indirect Claim may be liquidated and paid in an amount that exceeds what the Indirect Claimant has actually paid the related Direct Claimant.

To establish a presumptively valid Indirect PI Trust Claim, the Indirect Claimant's aggregate liability for the Direct Claimant's claim must also have been fixed, liquidated and paid fully by the Indirect Claimant by settlement (with an appropriate full release in favor of the PI Trust) or a Final Order (as defined in the Plan) provided that such claim is valid under the applicable state law. In any case where the Indirect Claimant has paid the claim of a Direct Claimant against the PI Trust under applicable law by way of a settlement, the Indirect Claimant shall obtain for the benefit of the PI Trust a release in form and substance satisfactory to the Trustees.

If an Indirect Claimant cannot meet the presumptive requirements set forth above, including the requirement that the Indirect Claimant provide the PI Trust with a full release of the Direct Claimant's claim, the Indirect Claimant may request that the PI Trust review the

Indirect PI Trust Claim individually to determine whether the Indirect Claimant can establish under applicable state law that the Indirect Claimant has paid all or a portion of a liability or obligation that the PI Trust had to the Direct Claimant as of the effective date of this TDP. If the Indirect Claimant can show that it has paid all or a portion of such a liability or obligation, the PI Trust shall reimburse the Indirect Claimant the amount of the liability or obligation so paid, times the then applicable Payment Percentage. However, in no event shall such reimbursement to the Indirect Claimant be greater than the amount to which the Direct Claimant would have otherwise been entitled. Further, the liquidated value of any Indirect PI Trust Claim paid by the PI Trust to an Indirect Claimant shall be treated as an offset to or reduction of the full liquidated value of any PI Trust Claim that might be subsequently asserted by the Direct Claimant against the PI Trust.

Any dispute between the PI Trust and an Indirect Claimant over whether the Indirect Claimant has a right to reimbursement for any amount paid to a Direct Claimant shall be subject to the ADR procedures provided in Section 5.10 below and set forth in Attachment A hereto. If such dispute is not resolved by said ADR procedures, the Indirect Claimant may litigate the dispute in the tort system pursuant to Sections 5.11 above and 7.6 below.

The Trustees may develop and approve a separate proof of claim form for Indirect PI
Trust Claims. Indirect PI Trust Claims that have not been disallowed, discharged, or otherwise
resolved by prior order of the Bankruptcy Court shall be processed in accordance with
procedures to be developed and implemented by the Trustees consistent with the provisions of
this Section 5.6, which procedures (a) shall determine the validity, allowability and
enforceability of such claims; and (b) shall otherwise provide the same liquidation and payment
procedures and rights to the holders of such claims as the PI Trust would have afforded the

holders of the underlying valid PI Trust Claims. Nothing in this TDP is intended to preclude a trust to which asbestos-related liabilities are channeled from asserting an Indirect PI Trust Claim against the PI Trust subject to the requirements set forth herein.

5.7 Evidentiary Requirements

5.7(a) Medical Evidence.

5.7(a)(1) In General. All diagnoses of a Disease Level shall be accompanied by either (i) a statement by the physician providing the diagnosis that at least ten (10) years have elapsed between the date of first exposure to asbestos or asbestos-containing products and the diagnosis, or (ii) a history of the claimant's exposure sufficient to establish a 10-year latency period. A finding by a physician after the Effective Date that a claimant's disease is "consistent with" or "compatible with" asbestosis will not alone be treated by the PI Trust as a diagnosis. 8

5.7(a)(1)(A). Disease Levels I-IV. Except for asbestos claims filed against OC, Fibreboard or another defendant in the tort system prior to the Petition Date, all diagnoses of a non-malignant asbestos-related disease (Disease Levels I-IV) shall be based in the case of a claimant who was living at the time the claim was filed, upon a physical examination of the claimant by the physician providing the diagnosis of the asbestos-related disease. In addition, all living claimants must provide (i) for Disease Levels I-III, evidence of Bilateral Asbestos-Related Nonmalignant Disease (as defined in Footnote 3 above); (ii) for

All diagnoses of Asbestosis/Pleural Disease (Disease Levels II and III) not based on pathology shall be presumed to be based on findings of bilateral asbestosis or pleural disease, and all diagnoses of Mesothelioma (Disease Level VIII) shall be presumed to be based on findings that the disease involves a malignancy. However, the PI Trust may refute such presumptions.

Disease Level IV, an ILO reading of 2/1 or greater or pathological evidence of asbestosis, and (iii) for Disease Levels III and IV, pulmonary function testing.⁹

In the case of a claimant who was deceased at the time the claim was filed, all diagnoses of a non-malignant asbestos-related disease (Disease Levels I-IV) shall be based upon either (i) a physical examination of the claimant by the physician providing the diagnosis of the asbestos-related disease; or (ii) pathological evidence of the non-malignant asbestos-related disease; or (iii) in the case of Disease Levels I-III, evidence of Bilateral Asbestos-Related Nonmalignant Disease (as defined in Footnote 3 above), and for Disease Level IV, either an ILO reading of 2/1 or greater or pathological evidence of asbestosis; and (iv) for either Disease Level III or IV, pulmonary function testing.

5.7(a)(1)(B). Disease Levels V – VIII. All diagnoses of an asbestos-related malignancy (Disease Levels V – VIII) shall be based upon either (i) a physical examination of the claimant by the physician providing the diagnosis of the asbestos-related disease, or (ii) a diagnosis of such a malignant Disease Level by a board-certified pathologist or by a pathology report prepared at or on behalf of a hospital accredited by the Joint Commission on Accreditation of Healthcare Organizations ("JCAHO").

[&]quot;Pulmonary function testing" or "PFT" shall mean testing that is in material compliance with the quality criteria established by the American Thoracic Society ("ATS") and is performed on equipment which is in material compliance with ATS standards for technical quality and calibration. PFT performed in a hospital accredited by the JCAHO, or performed, reviewed or supervised by a board certified pulmonologist or other Qualified Physician shall be presumed to comply with ATS standards, and the claimant may submit a summary report of the testing. If the PFT was not performed in a JCAHO accredited hospital, or performed, reviewed or supervised by a board certified pulmonologist or other Qualified Physician, the claimant must submit the full report of the testing (as opposed to a summary report); provided, however, that if the PFT was conducted prior to the Effective Date of the Plan and the full PFT report is not available, the claimant must submit a declaration signed by a Qualified Physician or other qualified party in the form provided by the PI Trust certifying that the PFT was conducted in material compliance with ATS standards.

5.7(a)(1)(C). Exception to the Exception for Certain

Pre-Petition Claims. If the holder of a PI Trust Claim that was filed against OC, Fibreboard or another defendant in the tort system prior to the Petition Date has available a report of a diagnosing physician engaged by the holder or his or her law firm who conducted a physical examination of the holder as described in Section 5.7(a)(1)(A), or if the holder has filed such medical evidence and/or a diagnosis of the asbestos-related disease by a physician not engaged by the holder or his or her law firm who conducted a physical examination of the claimant with another asbestos-related personal injury settlement trust that requires such evidence, without regard to whether the diagnosing physician was engaged by the holder or his or her law firm, the holder shall provide such medical evidence to the PI Trust notwithstanding the exception in Sections 5.7(a)(1)(A).

any payment to a claimant, the PI Trust must have reasonable confidence that the medical evidence provided in support of the claim is credible and consistent with recognized medical standards. The PI Trust may require the submission of X-rays, CT scans, detailed results of pulmonary function tests, laboratory tests, tissue samples, results of medical examination or reviews of other medical evidence, and may require that medical evidence submitted comply with recognized medical standards regarding equipment, testing methods and procedure to assure that such evidence is reliable. Medical evidence (i) that is of a kind shown to have been received in evidence by a state or federal judge at trial, (ii) that is consistent with evidence submitted to OC to settle for payment similar disease cases prior to OC 's bankruptcy, or (iii) that is a diagnosis by a physician shown to have previously qualified as a medical expert with respect to

the asbestos-related disease in question before a state or federal judge, is presumptively reliable, although the PI Trust may seek to rebut the presumption.

In addition, claimants who otherwise meet the requirements of this TDP for payment of a PI Trust Claim shall be paid irrespective of the results in any litigation at any time between the claimant and any other defendant in the tort system. However, any relevant evidence submitted in a proceeding in the tort system involving another defendant, other than any findings of fact, a verdict, or a judgment, may be introduced by either the claimant or the PI Trust in any Individual Review proceeding conducted pursuant to 5.3(b) or any Extraordinary Claim proceeding conducted pursuant to 5.4(a).

5.7(b) Exposure Evidence

qualify for any Disease Level, the claimant must demonstrate a minimum exposure to an asbestos-containing product manufactured or distributed by OC or Fibreboard. Claims based on conspiracy theories that involve no exposure to an asbestos-containing product produced by OC or Fibreboard are not compensable under this TDP. To meet the presumptive exposure requirements of Expedited Review set forth in Section 5.3(a)(3) above, the claimant must show (i) for all Disease Levels, OC or Fibreboard Exposure as defined in Section 5.7(b)(3) below prior to December 31, 1982; (ii) for Asbestos/Pleural Disease Level II, six months OC or Fibreboard Exposure prior to December 31, 1982, plus five (5) years cumulative occupational asbestos exposure; and (iii) for Asbestosis/Pleural Disease (Disease Level III), Severe Asbestosis (Disease Level IV), Other Cancer (Disease Level V) or Lung Cancer 1 (Disease Level VII), the claimant must show six months OC or Fibreboard Exposure prior to December 31, 1982, plus Significant Occupational Exposure to asbestos as defined below. If the claimant cannot meet the relevant

presumptive exposure requirements for a Disease Level eligible for Expedited Review, the claimant may seek Individual Review pursuant to Section 5.3(b) above of his or her exposure to an asbestos-containing product for which by OC or Fibreboard has legal responsibility.

5.7(b)(2) Significant Occupational Exposure. "Significant Occupational Exposure." Significant Occupational Exposure" means employment for a cumulative period of at least five (5) years, with a minimum of two (2) years prior to December 31, 1982 in an industry and an occupation in which the claimant (a) handled raw asbestos fibers on a regular basis; (b) fabricated asbestoscontaining products so that the claimant in the fabrication process was exposed on a regular basis to raw asbestos fibers; (c) altered, repaired or otherwise worked with an asbestos-containing product such that the claimant was exposed on a regular basis to asbestos fibers; or (d) was employed in an industry and occupation such that the claimant worked on a regular basis in close proximity to workers engaged in the activities described in (a), (b) and/or (c).

demonstrate meaningful and credible exposure, which occurred prior to December 31, 1982, to asbestos or asbestos-containing products supplied, specified, manufactured, installed, maintained, or repaired by either OC or Fibreboard, and/or any entity, including an OC or Fibreboard contracting unit, for which OC or Fibreboard has legal liability ("OC/Fibreboard Exposure"). That meaningful and credible exposure evidence may be established by an affidavit or sworn statement of the claimant, by an affidavit or sworn statement of a co-worker or the affidavit or sworn statement of a family member in the case of a deceased claimant (providing the PI Trust finds such evidence reasonably reliable), by invoices, employment, construction or similar records, or by other credible evidence. The specific exposure information required by the PI Trust to process a claim under either Expedited or Individual Review shall be set forth on the

proof of claim form to be used by the PI Trust. The PI Trust can also require submission of other or additional evidence of exposure when it deems such to be necessary.

Evidence submitted to establish proof of OC/Fibreboard Exposure is for the sole benefit of the PI Trust, not third parties or defendants in the tort system. The PI Trust has no need for, and therefore claimants are not required to furnish the PI Trust with evidence of, exposure to specific asbestos products other than those for which OC or Fibreboard has legal responsibility, except to the extent such evidence is required elsewhere in this TDP. Similarly, failure to identify OC or Fibreboard products in the claimant's underlying tort action, or to other bankruptcy trusts, does not preclude the claimant from recovering from the PI Trust, provided the claimant otherwise satisfies the medical and exposure requirements of this TDP.

5.8 Claims Audit Program. The PI Trust with the consent of the TAC and the Future Claimants' Representative may develop methods for auditing the reliability of medical evidence, including additional reading of X-rays, CT scans and verification of pulmonary function tests, as well as the reliability of evidence of exposure to asbestos, including exposure to asbestos-containing products manufactured or distributed by OC or Fibreboard prior to December 31, 1982. In the event that the PI Trust reasonably determines that any individual or entity has engaged in a pattern or practice of providing unreliable medical evidence to the Trust, it may decline to accept additional evidence from such provider in the future.

Further, in the event that an audit reveals that fraudulent information has been provided to the PI Trust, the PI Trust may penalize any claimant or claimant's attorney by disallowing the PI Trust Claim and/or by other means including, but not limited to, requiring the source of the fraudulent information to pay the costs associated with the audit and any future audit or audits, reordering the priority of payment of all affected claimants' PI Trust Claims, raising the level of

scrutiny of additional information submitted from the same source or sources, refusing to accept additional evidence from the same source or sources, seeking the prosecution of the claimant or claimant's attorney for presenting a fraudulent claim in violation of 18 U.S.C. §152, and seeking sanctions from the Bankruptcy Court.

Section 2.1 that provides that a claimant may not assert more than one PI Trust Claim hereunder, the holder of a PI Trust Claim involving a non-malignant asbestos-related disease (Disease Levels I through IV) may file a new PI Trust Claim against the PI Trust for a malignant disease (Disease Levels V – VIII) that is subsequently diagnosed. Any additional payments to which such claimant may be entitled with respect to such malignant asbestos-related disease shall not be reduced by the amount paid for the non-malignant asbestos-related disease, provided that the malignant disease had not been diagnosed at the time the claimant was paid with respect to his or her original claim involving the non-malignant disease.

5.10 Arbitration.

the TAC and the Future Claimants' Representative, shall institute binding and non-binding arbitration procedures in accordance with the ADR Procedures included in Attachment A hereto for resolving disputes concerning whether a Pre-Petition settlement agreement with OC or Fibreboard is binding and judicially enforceable in the absence of a Final Order of the Bankruptcy Court determining the issue, whether the PI Trust's outright rejection or denial of a claim was proper, or whether the claimant's medical condition or exposure history meets the requirements of this TDP for purposes of categorizing a claim involving Disease Levels I – VIII. Binding and non-binding arbitration shall also be available for resolving disputes over the liquidated value of a

claim involving Disease Levels II – VIII as well as disputes over OC's or Fibreboard's share of the unpaid portion of a Pre-Petition Liquidated Claim described in Section 5.2 above and disputes over the validity of an Indirect PI Trust Claim.

In all arbitrations, the arbitrator shall consider the same medical and exposure evidentiary requirements that are set forth in Section 5.7 above. In the case of an arbitration involving the liquidated value of a claim involving Disease Levels II – VIII, the arbitrator shall consider the same valuation factors that are set forth in Section 5.3(b)(2) above. In order to facilitate the Individual Review Process with respect to such claims, the PI Trust may from time to time develop a valuation model that enables the PI Trust to efficiently make initial liquidated value offers on these claims in the Individual Review setting. In an arbitration involving any such claim, the PI Trust shall neither offer into evidence or describe any such model nor assert that any information generated by the model has any evidentiary relevance or should be used by the arbitrator in determining the presumed correct liquidated value in the arbitration. The underlying data that was used to create the model may be relevant and may be made available to the arbitrator but only if provided to the claimant or his/her counsel ten (10) days prior to the arbitration proceeding. With respect to all claims eligible for arbitration, the claimant, but not the PI Trust, may elect either non-binding or binding arbitration. The ADR Procedures set forth in Attachment A hereto may be modified by the PI Trust with the consent of the TAC and the Future Claimants' Representative. Such amendments may include adoption of mediation procedures as well as establishment of an Extraordinary Claims Panel to review such claims pursuant to Section 5.4(a) above.

5.10(b) Claims Eligible for Arbitration. In order to be eligible for arbitration, the claimant must first complete the Individual Review Process set forth in Section 5.3(b) above,

as well as either the Pro-Bono Evaluation or the Mediation processes set forth in the ADR Procedures included in Attachment A, with respect to the disputed issue. Individual Review will be treated as completed for these purposes when the claim has been individually reviewed by the PI Trust, the PI Trust has made an offer on the claim, the claimant has rejected the liquidated value resulting from the Individual Review, and the claimant has notified the PI Trust of the rejection in writing. Individual Review will also be treated as completed if the PI Trust has rejected the claim.

5.10(c) Limitations on and Payment of Arbitration Awards. In the case of a non-Extraordinary Claim involving Disease Levels II – VIII, the arbitrator shall not return an award in excess of the Maximum Value for the appropriate Disease Level as set forth in Section 5.3(b)(4) above, and for an Extraordinary Claim involving one of those Disease Levels, the arbitrator shall not return an award greater than the maximum extraordinary value for such a claim as set forth in Section 5.4(a) above. A claimant who submits to arbitration and who accepts the arbitral award will receive payments in the same manner as one who accepts the PI Trust's original valuation of the claim.

5.11 Litigation. Claimants who elect non-binding arbitration and then reject their arbitral awards retain the right to institute a lawsuit in the tort system against the PI Trust pursuant to Section 7.6 below. However, a claimant shall be eligible for payment of a judgment for monetary damages obtained in the tort system from the PI Trust's available cash only as provided in Section 7.7 below.

SECTION VI

Claims Materials

6.1 Claims Materials. The PI Trust shall prepare suitable and efficient claims materials ("Claims Materials") for all PI Trust Claims, and shall provide such Claims Materials

upon a written request for such materials to the PI Trust. The proof of claim form to be submitted to the PI Trust shall require the claimant to assert the highest Disease Level for which the claim qualifies at the time of filing, and shall include a certification by the claimant or his or her attorney sufficient to meet the requirements of Rule 11(b) of the Federal Rules of Civil Procedure. In developing its claim filing procedures, the PI Trust shall make every reasonable effort to provide claimants with the opportunity to utilize currently available technology at their discretion, including filing claims and supporting documentation over the Internet and electronically by disk or CD-rom. The proof of claim forms to be used by the PI Trust shall be developed by the PI Trust and submitted to the TAC and the Future Claimants' Representative for approval. The proof of claim forms may be changed by the PI Trust with the consent of the TAC and the Future Claimants' Representative.

- 6.2 Content of Claims Materials. The Claims Materials shall include a copy of this TDP, such instructions as the Trustees shall approve, and a detailed proof of claim form. If feasible, the forms used by the PI Trust to obtain claims information shall be the same or substantially similar to those used by other asbestos claims resolution organizations. If requested by the claimant, the PI Trust shall accept information provided electronically. The claimant may, but will not be required to, provide the PI Trust with evidence of recovery from other asbestos defendants and claims resolution organizations.
- 6.3 Withdrawal or Deferral of Claims. A claimant can withdraw a PI Trust Claim at any time upon written notice to the PI Trust and file another claim subsequently without affecting the status of the claim for statute of limitations purposes, but any such claim filed after withdrawal shall be given a place in the FIFO Processing Queue based the date of such subsequent filing. A claimant can also request that the processing of his or her PI Trust Claim by

the PI Trust be deferred for a period not to exceed three (3) years without affecting the status of the claim for statute of limitation purposes, in which case the claimant shall also retain his or her original place in the FIFO Processing Queue. During the period of such deferral, a sequencing adjustment on such claimant's PI Trust Claim as provided in Section 7.5 hereunder shall not accrue and payment thereof shall be deemed waived by the claimant. Except for PI Trust Claims held by representatives of deceased or incompetent claimants for which court or probate approval of the PI Trust's offer is required, or a PI Trust Claim for which deferral status has been granted, a claim will be deemed to have been withdrawn if the claimant neither accepts, rejects, nor initiates arbitration within six months of the PI Trust's written offer of payment or rejection of the claim. Upon written request and good cause, the PI Trust may extend either the deferral or withdrawal period for an additional six (6) months.

- 6.4 Filing Requirements and Fees. The Trustees shall have the discretion to determine, with the consent of the TAC and the Futures Representative, (a) whether a claimant must have previously filed an asbestos-related personal injury claim in the tort system to be eligible to file the claim with the PI Trust and (b) whether a filing fee should be required for any PI Trust Claims.
- 6.5 Confidentiality of Claimants' Submissions. All submissions to the PI Trust by a holder of a PI Trust Claim or a proof of claim form and materials related thereto shall be treated as made in the course of settlement discussions between the holder and the PI Trust and intended by the parties to be confidential and to be protected by all applicable state and federal privileges, including, but not limited to, those directly applicable to settlement discussions. The PI Trust will preserve the confidentiality of such claimant submissions, and shall disclose the contents thereof only, with the permission of the holder, to another trust established for the

benefit of asbestos personal injury claimants pursuant to section 524(g) and/or section 105 of the Bankruptcy Code or other applicable law, to such other persons as authorized by the holder, or in response to a valid subpoena of such materials issued by the Bankruptcy Court. Furthermore, the PI Trust shall provide counsel for the holder a copy of any such subpoena immediately upon being served. The PI Trust shall on its own initiative or upon request of the claimant in question take all necessary and appropriate steps to preserve said privilege before the Bankruptcy Court and before those courts having appellate jurisdiction related thereto. Notwithstanding anything in the foregoing to the contrary, with the consent of the TAC and the Future Claimants' Representative, the PI Trust may, in specific limited instances, disclose information, documents, or other materials reasonably necessary in the PI Trust's judgment to preserve, litigate, resolve, or settle coverage, or to comply with an applicable obligation under an insurance policy or settlement agreement within the OC Asbestos Personal Injury Liability Insurance Assets; provided, however, that the PI Trust shall take any and all steps reasonably feasible in its judgment to preserve the further confidentiality of such information, documents and materials, and prior to the disclosure of such information, documents or materials to a third party, the PI Trust shall receive from such third party a written agreement of confidentiality that (a) ensures that the information, documents and materials provided by the PI Trust shall be used solely by the receiving party for the purpose stated in the agreement and (b) prohibits any other use or further dissemination of the information, documents and materials by the third party.

SECTION VII

General Guidelines for Liquidating and Paying Claims

7.1 Showing Required. To establish a valid PI Trust Claim, a claimant must meet the requirements set forth in this TDP. The PI Trust may require the submission of X-rays, CT

scans, laboratory tests, medical examinations or reviews, other medical evidence, or any other evidence to support or verify the PI Trust Claim, and may further require that medical evidence submitted comply with recognized medical standards regarding equipment, testing methods, and procedures to assure that such evidence is reliable.

- 7.2 Costs Considered. Notwithstanding any provisions of this TDP to the contrary, the Trustees shall always give appropriate consideration to the cost of investigating and uncovering invalid PI Trust Claims so that the payment of valid PI Trust Claims is not further impaired by such processes with respect to issues related to the validity of the medical evidence supporting a PI Trust Claim. The Trustees shall also have the latitude to make judgments regarding the amount of transaction costs to be expended by the PI Trust so that valid PI Trust Claims are not unduly further impaired by the costs of additional investigation. Nothing herein shall prevent the Trustees, in appropriate circumstances, from contesting the validity of any claim against the PI Trust whatever the costs, or declining to accept medical evidence from sources that the Trustees have determined to be unreliable pursuant to the Claims Audit Program described in Section 5.8 above.
- 7.3 Discretion to Vary the Order and Amounts of Payments in Event of Limited Liquidity. Consistent with the provisions hereof and subject to the FIFO Processing and Liquidation Queues, the Maximum Annual Payment, the Maximum Available Payment and the Claims Payment Ratio requirements set forth above, the Trustees shall proceed as quickly as possible to liquidate valid PI Trust Claims, and shall make payments to holders of such claims in accordance with this TDP from the OC and/or Fibreboard Sub-Accounts promptly as monies become available and as claims are liquidated, while maintaining sufficient assets within each Sub-Account to pay future valid claims in substantially the same manner.

Because the PI Trust's income over time remains uncertain, and decisions about payments must be based on estimates that cannot be done precisely, they may have to be revised in light of experiences over time, and there can be no guarantee of any specific level of payment for claims against either Sub-Account. However, the Trustees shall use their best efforts to treat similar claims in substantially the same manner, consistent with their duties as Trustees, the purposes of the PI Trust, the established allocation of monies to claims in Categories A and B, and the practical limitations imposed by the inability to predict the future with precision. In the event that either or both of the OC or the Fibreboard Sub-Accounts face temporary periods of limited liquidity, the Trustees may, with the consent of the TAC and the Future Claimants' Representative, (a) suspend the normal order of payment from such Sub-Account, (b) temporarily limit or suspend payments from such Sub-Account altogether, (c) offer a Reduced Payment Option for the Sub-Account as described in Section 2.5 above and/or (d) commence making payments on an installment basis.

Alabama Wrongful Death Statute, in determining the value of any liquidated or unliquidated PI Trust Claim, punitive or exemplary damages, *i.e.*, damages other than compensatory damages, shall not be considered or allowed, notwithstanding their availability in the tort system.

Similarly, no punitive or exemplary damages shall be payable with respect to any claim litigated against the PI Trust in the tort system pursuant to Sections 5.11 above and 7.6 below. The only damages that may be awarded pursuant to this TDP to Alabama Claimants who are deceased and whose personal representatives pursue their claims only under the Alabama Wrongful Death Statute shall be compensatory damages determined pursuant to the statutory and common law of the Commonwealth of Pennsylvania, without regard to its choice of law principles. The choice

of law provision in Section 7.4 herein applicable to any claim with respect to which, but for this choice of law provision, the applicable law of the Claimant's Jurisdiction pursuant to Section 5.3(b)(2) is determined to be the Alabama Wrongful Death Statute, shall only govern the rights between the PI Trust and the claimant including, but not limited to, suits in the tort system pursuant to Section 7.6, and to the extent the PI Trust seeks recovery from any entity that provided insurance to OC or Fibreboard, the Alabama Wrongful Death Statute shall govern.

7.5 Sequencing Adjustment.

7.5(a) In General. Except for PI Trust Claims involving Other Asbestos

Disease (Disease Level I – Cash Discount Payment) and subject to the limitations set forth

below, a sequencing adjustment shall be paid on all PI Trust Claims with respect to which the

claimant has had to wait a year or more for payment, provided, however, that no claimant shall

receive a sequencing adjustment for a period in excess of seven (7) years. The sequencing

adjustment factor for each year shall be the coupon issue yield equivalent (as determined by the

Secretary of the Treasury) of the average accepted auction price for the first auction of 5-year

Treasury Notes occurring in such year.

7.5(b) Unliquidated PI Trust Claims. A sequencing adjustment shall be payable on the Scheduled Value of any unliquidated PI Trust Claim that meets the requirements of Disease Levels II – V, VII and VIII, whether the claim is liquidated under Expedited Review, Individual Review, or by arbitration. No sequencing adjustment shall be paid on any claim involving Disease Level I, or on any claim liquidated in the tort system pursuant to Section 5.11 above and Section 7.6 below. The sequencing adjustment on an unliquidated PI Trust Claim that meets the requirements of Disease Level VI shall be based on the Average Value of such a claim. Sequencing adjustments on all such unliquidated claims shall be measured from the date of

payment back to the earliest of the date that is one (1) year after the date on which (a) the claim was filed against OC or Fibreboard prior to the Petition Date; (b) the claim was filed against another defendant in the tort system on or after the Petition Date but before the Effective Date; or (c) the claim was filed with the PI Trust after the Effective Date.

- 7.5(c) Liquidated Pre-Petition Claims. A sequencing adjustment shall also be payable on the liquidated value of all Pre-Petition Liquidated Claims described in Section 5.2(a) above. In the case of Pre-Petition Liquidated Claims liquidated by verdict or judgment, the sequencing adjustment shall be measured from the date of payment back to the date that is one (1) year after the date that the verdict or judgment was entered. In the case of Pre-Petition Liquidated Claims liquidated by a binding, judicially enforceable settlement, the sequencing adjustment shall be measured from the date of payment back to the date that is one (1) year after the Petition Date.
- Trust's determination regarding the Disease Level of the claim, the claimant's exposure history or the liquidated value of the claim, and if the holder has first submitted the claim to non-binding arbitration as provided in Section 5.10 above, the holder may file a lawsuit in the Claimant's Jurisdiction as defined in Section 5.3(b)(2) above. Any such lawsuit must be filed by the claimant in her or her own right and name and not as a member or representative of a class, and no such lawsuit may be consolidated with any other lawsuit. All defenses (including, with respect to the PI Trust, all defenses which could have been asserted by OC or Fibreboard) shall be available to both sides at trial; however, the PI Trust may waive any defense and/or concede any issue of fact or law. If the claimant was alive at the time the initial pre-petition complaint was filed or on the date the proof of claim was filed with the PI Trust, the case will be treated as

a personal injury case with all personal injury damages to be considered even if the claimant has died during the pendency of the claim.

7.7 Payment of Judgments for Money Damages. If and when an OC or Fibreboard claimant obtains a judgment in the tort system, the claim shall be placed in the relevant FIFO Payment Queue based on the date on which the judgment became final. Thereafter, the claimant shall receive from the OC or Fibreboard Sub-Account an initial payment (subject to the applicable Payment Percentage, the Maximum Available Payment, the Claims Payment Ratio, and the sequencing adjustment provisions set forth above) of an amount equal to one-hundred percent (100%) of the greater of (i) the PI Trust's last offer to the claimant, or (ii) the award that the claimant declined in non-binding arbitration. The claimant shall receive the balance of the judgment, if any, in five equal installments in years six (6) through ten (10) following the year of the initial payment (also subject to the applicable Payment Percentage, the Maximum Available Payment, the Claims Payment Ratio, and the sequencing adjustment provisions set forth above in effect on the date of the payment of the subject installment).

In the case of non-Extraordinary claims involving Disease Levels II - VIII, the total amounts paid with respect to such claims shall not exceed the Maximum Values for such Disease Levels set forth in Section 5.3(b)(4). In the case of Extraordinary Claims, the total amounts paid with respect to such claims shall not exceed the Maximum Value for such claims set forth in Section 5.4(a) above. In the case of claims involving Disease Level I, the total amounts paid shall not exceed the Scheduled Value of such claims. Under no circumstances shall either a sequencing adjustment be paid pursuant to Section 7.5 or interest be paid under any statute on any judgments obtained in the tort system pursuant to Sections 5.11 and 7.6 above.

- 7.8 Releases. The Trustees shall have the discretion to determine the form and substance of the releases to be provided to the PI Trust in order to maximize recovery for claimants against other tortfeasors without increasing the risk or amount of claims for indemnification or contribution from the PI Trust. As a condition to making any payment to a claimant, the PI Trust shall obtain a general, partial, or limited release as appropriate in accordance with the applicable state or other law. If allowed by state law, the endorsing of a check or draft for payment by or on behalf of a claimant may, in the discretion of the PI Trust, shall constitute such a release.
- 7.9 Third-Party Services. Nothing in this TDP shall preclude the PI Trust from contracting with another asbestos claims resolution organization to provide services to the PI Trust so long as decisions about the categorization and liquidated value of PI Trust Claims are based on the relevant provisions of this TDP, including the Disease Levels, Scheduled Values, Average Values, Maximum Values, and Medical/Exposure Criteria set forth above.
- 7.10 PI Trust Disclosure of Information. Periodically, but not less often than once a year, the PI Trust shall make available to claimants and other interested parties, the number of claims by Disease Levels that have been resolved both by the Individual Review Process and by arbitration as well as by litigation in the tort system, indicating the amounts of the awards and the averages of the awards by jurisdiction.

SECTION VIII

Miscellaneous

8.1 Amendments. Except as otherwise provided herein, the Trustees may amend, modify, delete, or add to any provisions of this TDP (including, without limitation, amendments to conform this TDP to advances in scientific or medical knowledge or other changes in

circumstances), provided they first obtain the consent of the TAC and the Future Claimants' Representative pursuant to the Consent Process set forth in Sections 5.7(b) and 6.6(b) of the PI Trust Agreement, except that the right to amend the Claims Payment Ratio is governed by the restrictions in Section 2.5 above, and the right to adjust the Payment Percentage is governed by Section 4.2 above. Nothing herein is intended to preclude the TAC or the Future Claimants' Representative from proposing to the Trustees, in writing, amendments to this TDP. Any amendment proposed by the TAC or the Future Claimants' Representative shall remain subject to Section 7.3 of the PI Trust Agreement.

- unenforceable, such determination shall in no way limit or affect the enforceability and operative effect of any and all other provisions of this TDP. Should any provision contained in this TDP be determined to be inconsistent with or contrary to OC's or Fibreboard's obligations to any insurance company providing insurance coverage to OC and/or Fibreboard in respect of claims for personal injury based on exposure to asbestos-containing products manufactured or produced by OC or Fibreboard, the PI Trust with the consent of the TAC and the Future Claimants' Representative may amend this TDP and/or the PI Trust Agreement to make the provision of either or both documents consistent with the duties and obligations of OC or Fibreboard to said insurance company.
- 8.3 Governing Law. Except for purposes of determining the liquidated value of any PI Trust Claim, administration of this TDP shall be governed by, and construed in accordance with, the laws of the State of Delaware. The law governing the liquidation of PI Trust Claims in the case of Individual Review, arbitration or litigation in the tort system shall be the law of the Claimant's Jurisdiction as described in Section 5.3(b)(2) above.

EXHIBIT K

UNITED STATES GYPSUM

ASBESTOS PERSONAL INJURY SETTLEMENT TRUST DISTRIBUTION PROCEDURES

Revised March 29, 2010

UNITED STATES GYPSUM

ASBESTOS PERSONAL INJURY SETTLEMENT TRUST DISTRIBUTION PROCEDURES

TABLE OF CONTENTS

		Page
SECTION I —	Introduction	2
1.1	Purpose	2
	Interpretation	2
SECTION II —	- Overview	. 2
2.1	PI Trust Goals	2
2.2	Claims Liquidation Procedures	3
2.3	Application of the Payment Percentage	5
	PI Trust's Determination of the Maximum Annual Payment	
	and Maximum Available Payment	. 7
2.5	Claims Payment Ratio	9
2.6	Indirect PI Trust Claims	11
SECTION III -	— TDP Administration	11
3.1	Trust Advisory Committee and Futures Representative	11
	Consent and Consultation Procedures	11
.*		
SECTION IV -	- Payment Percentage; Periodic Estimates	12
4.1	Uncertainty of USG's Personal Injury Asbestos Liabilities	12
	Computation of Payment Percentage	12
	Applicability of the Payment Percentage	14

1′
13
18
20
20
22
22
24
24
2:
0
29
32
p 34
36
36
39
39
39
40
40
41
42
43
43
44
ZI-Zi

	5.10	Arbitration	46
	·	(a) Establishment of ADR Procedures	46
		(b) Claims Eligible for Arbitration	47
,	•	(c) Limitations on and Payment of Arbitration Awards	47
,	5.11	Litigation	48
SEC'	TION V	I — Claims Materials	48
	6.1	Claims Materials	48
	6.2	Content of Claims Materials	48
	6.3	Withdrawal or Deferral of Claims	49
	6.4	Filing Requirements and Fees	49
	6.5	Confidentiality of Claimants' Submissions	50
SEC	TION V	II — General Guidelines for Liquidating and Paying Claims	51
	7.1	Showing Required	51
	7.2	Costs Considered	51
	7.3	Discretion to Vary the Order and Amounts of Payments in	31
•	7.5	Event of Limited Liquidity	52
	7.4	Punitive Damages	53
	7.5	Sequencing Adjustment	53
	1.5	(a) In General.	53
		(b) Unliquidated PI Trust Claims	54
		(c) Liquidated Pre-Petition Claims	54
	7.6	Suits in the Tort System	55
	7.0 7.7	Payment of Judgments for Money Damages	55 55
	7.7		56
	7.8 7.9	Releases	57
		Third-Party Services	
	7.10	PI Trust Disclosure of Information	57
SEC	TION V	III — Miscellaneous	57
	8.1	Amendments	57
•	8.2	Severability	57 58
	8.3	Governing Law	58
	0.2	CIOVERHIE LAW	.36

UNITED STATES GYPSUM

ASBESTOS PERSONAL INJURY SETTLEMENT TRUST DISTRIBUTION PROCEDURES

The United States Gypsum Asbestos Personal Injury Settlement Trust Distribution Procedures (the "TDP") contained herein provide for resolving all "Asbestos Personal Injury Claims" as defined in the Joint Plan of Reorganization of USG Corporation and its Debtor Subsidiaries, dated as of March 27, 2006 (as it may be amended or modified, the "Plan"), 1 including (a) all asbestos-related personal injury and death claims caused by conduct of, and/or exposure to products for which, USG Corporation and its subsidiaries, including United States Gypsum Company (collectively referred to as "USG"), and their predecessors, successors, and assigns, have legal responsibility and (b) all asbestos-related personal injury and death claims caused by conduct of, and/or exposure to products for which, A.P. Green Industries, Inc., A.P. Green Refractories Co. or any of their affiliates or predecessors to the extent such predecessors are listed in Exhibit I.A.96 of the Plan (collectively, "A.P. Green") have legal responsibility to the extent such conduct occurred prior to January 2, 1968 and/or the legal responsibility arose prior to such date, as provided in and required by the Plan and the United States Gypsum Asbestos Personal Injury Settlement Trust Agreement (the "PI Trust Agreement"). The Plan and PI Trust Agreement establish the United States Gypsum Asbestos Personal Injury Settlement Trust (the "PI Trust"). The Trustees of the PI Trust (the "Trustees") shall implement and administer this TDP in accordance with the PI Trust Agreement.

¹ Capitalized terms used herein and not otherwise defined shall have the meanings assigned to them in the Plan and the PI Trust Agreement; provided, however, that "Asbestos Personal Injury Claims" as defined in the Plan shall be referred to herein as "PI Trust Claims" and "Asbestos

SECTION I

Introduction

- 1.1 Purpose. This TDP has been adopted pursuant to the PI Trust Agreement. It is designed to provide fair, equitable and substantially similar treatment for all PI Trust Claims that may presently exist or may arise in the future.
- 1.2 Interpretation. Except as may otherwise be provided below, nothing in this TDP shall be deemed to create a substantive right for any claimant. The rights and benefits provided herein to holders of PI Trust Claims shall vest in such holders as of the Effective Date.

SECTION II

Overview

2.1 PI Trust Goals. The goal of the PI Trust is to treat all claimants equitably. This TDP furthers that goal by setting forth procedures for processing and paying USG's several share of the unpaid portion of the liquidated value of PI Trust Claims generally on an impartial, first-in-first-out ("FIFO") basis, with the intention of paying all claimants over time as equivalent a share as possible of the value of their claims based on historical values for substantially similar claims in the tort system.² To this end, the TDP establishes a schedule of eight asbestos-related diseases ("Disease Levels"), seven of which have presumptive medical and exposure requirements ("Medical/Exposure Criteria") and specific liquidated values ("Scheduled Values"), and five of which have both anticipated average values ("Average Values") and caps on their liquidated values ("Maximum Values"). The Disease Levels, Medical/Exposure

Personal Injury Indirect Claims" as defined in the Plan shall be referred to herein as "Indirect PI Trust Claims."

Criteria, Scheduled Values, Average Values and Maximum Values, which are set forth in Sections 5.3 and 5.4 below, have all been selected and derived with the intention of achieving a fair allocation of the PI Trust funds as among claimants suffering from different disease processes in light of the best available information considering the settlement histories of USG and A.P. Green and the rights claimants would have in the tort system absent the bankruptcy. A claimant may not assert more than one PI Trust Claim hereunder.

2.2 Claims Liquidation Procedures. PI Trust Claims shall be processed based on their place in the FIFO Processing Queue to be established pursuant to Section 5.1(a) below. The PI Trust shall take all reasonable steps to resolve PI Trust Claims as efficiently and expeditiously as possible at each stage of claims processing and arbitration, which steps may include, in the PI Trust's sole discretion, conducting settlement discussions with claimants' representatives with respect to more than one claim at a time, provided that the claimants' respective positions in the FIFO Processing Queue are maintained and each claim is individually evaluated pursuant to the valuation factors set forth in Section 5.3(b)(2) below. The PI Trust shall also make every effort to resolve each year at least that number of PI Trust Claims required to exhaust the Maximum Annual Payment and the Maximum Available Payment for Category A and Category B claims, as those terms are defined below.

The PI Trust shall liquidate all PI Trust Claims except Foreign Claims (as defined below) that meet the presumptive Medical/Exposure Criteria of Disease Levels I–V, VII and VIII under the Expedited Review Process described in Section 5.3(a) below. Claims involving Disease Levels I–V, VII and VIII that do not meet the presumptive Medical/Exposure Criteria for the

² As used in this TDP, the phrase "in the tort system" shall not include claims asserted against a trust established for the benefit of asbestos personal injury claimants pursuant to section 524(g) and/or section 105 of the Bankruptcy Code or any other applicable law.

relevant Disease Level may undergo the PI Trust's Individual Review Process described in Section 5.3(b) below. In such a case, notwithstanding that the claim does not meet the presumptive Medical/Exposure Criteria for the relevant Disease Level, the PI Trust can offer the claimant an amount up to the Scheduled Value of that Disease Level if the PI Trust is satisfied that the claimant has presented a claim that would be cognizable and valid in the tort system.

PI Trust Claims involving Disease Levels IV-VIII tend to raise more complex valuation issues than the PI Trust Claims in Disease Levels I-III. Accordingly, in lieu of liquidating such claimant's claim under the Expedited Review Process, claimants holding claims involving these Disease Levels may alternatively seek to establish a liquidated value for the claim that is greater than its Scheduled Value by electing the PI Trust's Individual Review Process. However, the liquidated value of a more serious Disease Level IV, V, VII or VIII claim that undergoes the Individual Review Process for valuation purposes may be determined to be less than its Scheduled Value, and in any event shall not exceed the Maximum Value for the relevant Disease Level set forth in Section 5.3(b)(3) below, unless the claim qualifies as an Extraordinary Claim as defined in Section 5.4(a) below, in which case its liquidated value cannot exceed the Maximum Value specified in that provision for such claims. Level VI (Lung Cancer 2) claims and all Foreign Claims may be liquidated³ only pursuant to the PI Trust's Individual Review Process.

Based upon USG's and A.P. Green's claims settlement histories in light of applicable tort law, and current projections of present and future unliquidated claims, the Scheduled Values and Maximum Values set forth in Section 5.3(b)(3) have been established for each of the five (5) more serious Disease Levels that are eligible for Individual Review of their liquidated values,

with the expectation that the combination of settlements at the Scheduled Values and those resulting from the Individual Review Process should result in the Average Values also set forth in that provision.

All unresolved disputes over a claimant's medical condition, exposure history and/or the liquidated value of the claim shall be subject to binding or non-binding arbitration as set forth in Section 5.10 below, at the election of the claimant, under the ADR Procedures that are provided in Attachment A hereto. PI Trust Claims that are the subject of a dispute with the PI Trust that cannot be resolved by non-binding arbitration may enter the tort system as provided in Sections 5.11 and 7.6 below. However, if and when a claimant obtains a judgment in the tort system, the judgment shall be payable (subject to the Payment Percentage, Maximum Available Payment, and Claims Payment Ratio provisions set forth below) as provided in Section 7.7 below.

2.3 Application of the Payment Percentage. After the liquidated value of a PI Trust Claim other than a claim involving Other Asbestos Disease (Disease Level I – Cash Discount Payment), as defined in Section 5.3(a)(3) below, is determined pursuant to the procedures set forth herein for Expedited Review, Individual Review, arbitration, or litigation in the tort system, the claimant shall ultimately receive a pro-rata share of that value based on a Payment Percentage described in Section 4.2 below. The Payment Percentage shall also apply to all Pre-Petition Liquidated Claims as provided in Section 5.2 below and to all sequencing adjustments pursuant to Section 7.5 below.

The Initial Payment Percentage has been set at 45% and shall apply to all PI Trust Voting Claims accepted as valid by the PI Trust, unless adjusted by the PI Trust pursuant to the consent of the PI Trust Advisory Committee (the "TAC") and the Legal Representative for Future

³ For purposes of this TDP, "liquidated" means approved and valued by the PI Trust.

Claimants (the "Futures Representative") (who are described in Section 3.1 below) pursuant to Section 4.2 below, and except as provided in Section 4.3 below with respect to supplemental payments in the event the Initial Payment Percentage is changed. The term "PI Trust Voting Claims" includes (i) Pre-Petition Liquidated Claims as defined in Section 5.2(a) below; (ii) claims filed against USG in the tort system or actually submitted to USG pursuant to an administrative settlement agreement prior to the Petition Date of June 25, 2001; and (iii) all asbestos claims filed against another defendant in the tort system prior to February 17, 2006, the date the Plan was filed with the Bankruptcy Court (the "Plan Filing Date"); provided, however, that (1) the holder of a claim described in subsection (i), (ii) or (iii) above, or his or her authorized agent, actually voted to accept or reject the Plan pursuant to the voting procedures established by the Bankruptcy Court, unless such holder certifies to the satisfaction of the Trustees that he or she was prevented from voting in this proceeding as a result of circumstances resulting in a state of emergency affecting, as the case may be, the holder's residence, principal place of business or legal representative's place of business at which the holder or his or her legal representative receives notice and/or maintains material records relating to his or her PI Trust Voting Claim; and provided further that (2) the claim was subsequently filed with the PI Trust pursuant to Section 6.1 below by the Initial Claims Filing Date defined in Section 5.1(a) below. The Initial Payment Percentage has been calculated on the assumption that the Average Values set forth in Section 5.3(b)(3) below shall be achieved with respect to existing present claims and projected future claims involving Disease Levels IV-VIII.

The Payment Percentage may thereafter be adjusted upwards or downwards from time to time by the PI Trust with the consent of the TAC and the Futures Representative to reflect then-current estimates of the PI Trust's assets and its liabilities, as well as then-estimated value of

then-pending and future claims. Any adjustment to the Initial Payment Percentage shall be made only pursuant to Section 4.2 below. If the Payment Percentage is increased over time, claimants whose claims were liquidated and paid in prior periods under the TDP shall receive additional payments only as provided in Section 4.2 below. Because there is uncertainty in the prediction of both the number and severity of future PI Trust Claims, and the amount of the PI Trust's assets, no guarantee can be made of any Payment Percentage of a PI Trust Claim's liquidated value.

Available Payment. The PI Trust shall estimate or model the amount of cash flow anticipated to be necessary over its entire life to ensure that funds shall be available to treat all present and future holders of PI Trust Claims as similarly as possible. In each year, the PI Trust shall be empowered to pay out all of the income earned during the year (net of taxes payable with respect thereto), together with a portion of its principal, calculated so that the application of PI Trust funds over its life shall correspond with the needs created by the estimated initial backlog of claims and the estimated anticipated future flow of claims (the "Maximum Annual Payment"), taking into account the Payment Percentage provisions set forth in Section 2.3 above and Sections 4.2 and 4.3 below. The PI Trust's distributions to all claimants for that year shall not exceed the Maximum Annual Payment determined for that year.

For the period from the Effective Date to the earlier of (a) the Trigger Date (as defined in the Plan) if the FAIR Act (as defined in the Plan) is not enacted and made law on or before the Trigger Date and (b) if the FAIR Act is enacted and made law on or before the Trigger Date and a Challenge Proceeding (as defined in the Plan) is commenced, the date a Final Order (as defined

in the Plan) is entered resolving the Challenge Proceeding, the Maximum Annual Payment shall not exceed \$90 million.

In distributing the Maximum Annual Payment, the PI Trust shall first allocate the amount in question to (a) outstanding Pre-Petition Liquidated Claims, (b) PI Trust Claims involving Disease Level I (Cash Discount Payment) which have been liquidated by the PI Trust, and (c) any claims (i) based upon a diagnosis dated prior to January 30, 2006, and (ii) subsequently filed with the PI Trust pursuant to Section 6.1 below within one (1) year following the date the PI Trust first accepts for processing the proof of claim forms and the other materials required to file a claim with the PI Trust⁴, which are liquidated by the PI Trust ("Existing Claims"). Should the Maximum Annual Payment be insufficient to pay all such claims in full, they shall be paid in proportion to the aggregate value of each group of claims and the available funds allocated to each group of claims shall be paid to the maximum extent to claimants in the particular group based on their place in their respective FIFO Payment Queue. Claims in any group for which there are insufficient funds shall be carried over to the next year, and placed at the head of their FIFO Payment Queue. If there is a decrease in the Payment Percentage prior to the payment of such claims, any such Pre-Petition Liquidated Claims and Existing Claims shall nevertheless be entitled to be paid at the Payment Percentage that they would have been entitled to receive but for the application of the Maximum Annual Payment. The remaining portion of the Maximum Annual Payment (the "Maximum Available Payment"), if any, shall then be allocated and used to satisfy all other liquidated PI Trust Claims, subject to the Claims Payment Ratio set forth in

⁴ Exceptions to the satisfaction of this one-year filing requirement will be made where a claimant can show inability to file prior to the Effective Date caused by extraneous factors including but not limited to the need to complete probate.

Section 2.5 below. Claims in the groups described in (a), (b) and (c) above shall not be subject to the Claims Payment Ratio.

2.5 Claims Payment Ratio. Based upon USG's and A.P. Green's claims settlement histories and analysis of present and future claims, a Claims Payment Ratio has been determined which, as of the Effective Date, has been set at 85% for Category A claims, which consist of PI Trust Claims involving severe asbestosis and malignancies (Disease Levels IV–VIII) and at 15% for Category B claims, which are PI Trust Claims involving non-malignant Asbestosis or Pleural Disease (Disease Levels II and III).

In each year, after the determination of the Maximum Available Payment described in Section 2.4 above, 85% of that amount shall be available to pay Category A claims and 15% shall be available to pay Category B claims that have been liquidated since the Petition Date except for claims liquidated which, pursuant to Section 2.4 above, are not subject to the Claims Payment Ratio. In the event there are insufficient funds in any year to pay the liquidated claims within either or both of the Categories, the available funds allocated to the particular Category shall be paid to the maximum extent to claimants in that Category based on their place in the FIFO Payment Queue described in Section 5.1(c) below, which shall be based upon the date of claim liquidation. Claims for which there are insufficient funds allocated to the relevant Category shall be carried over to the next year where they shall be placed at the head of the FIFO Payment Queue. If there is a decrease in the Payment Percentage prior to the payment of such claims, such claims shall nevertheless be entitled to be paid at the Payment Percentage that they would have been entitled to receive but for the application of the Claims Payment Ratio. If there are excess funds in either or both Categories, because there is an insufficient amount of liquidated claims to exhaust the respective Maximum Available Payment amount for that

Category, then the excess funds for either or both Categories shall be rolled over and remain dedicated to the respective Category to which they were originally allocated.

The 85%/15% Claims Payment Ratio and its rollover provision shall be continued absent circumstances, such as a significant change in law or medicine, necessitating amendment to avoid a manifest injustice. However, the accumulation, rollover and subsequent delay of claims resulting from the application of the Claims Payment Ratio shall not, in and of itself, constitute such circumstances. In addition, an increase in the numbers of Category B claims beyond those predicted or expected shall not be considered as a factor in deciding whether to reduce the percentage allocated to Category A claims.

In considering whether to make any amendments to the Claims Payment Ratio and/or its rollover provisions, the Trustees shall consider the reasons for which the Claims Payment Ratio and its rollover provisions were adopted, the settlement histories that gave rise to its calculation, and the foreseeability or lack of foreseeability of the reasons why there would be any need to make an amendment. In that regard, the Trustees should keep in mind the interplay between the Payment Percentage and the Claims Payment Ratio as it affects the net cash actually paid to claimants.

In any event, no amendment to the Claims Payment Ratio to reduce the percentage allocated to Category A claims may be made without the unanimous consent of the TAC members and the consent of the Futures Representative. The percentage allocated to Category A claims may be increased at any time after the first anniversary of the date the PI Trust first accepts for processing proof of claim forms and the other materials required to file a claim with the PI Trust, with the consent of the TAC and the Futures Representative. In case of any amendments to the Claims Payment Ratio, consents shall be governed by the consent process set

forth in Sections 5.7(b) and 6.6(b) of the PI Trust Agreement, however, failure to obtain consent shall not be subject to the Dispute Resolution provisions of Section 7.13 thereof. The Trustees, with the consent of the TAC and the Futures Representative, may offer the option of a reduced Payment Percentage to holders of claims in either Category A or Category B in return for prompter payment (the "Reduced Payment Option").

2.6 Indirect PI Trust Claims. As set forth in Section 5.6 below, Indirect PI Trust Claims, if any, shall be subject to the same categorization, evaluation, and payment provisions of this TDP as all other PI Trust Claims.

SECTION III

TDP Administration

- and the PI Trust Agreement, the PI Trust and this TDP shall be administered by the Trustees in consultation with the TAC, which represents the interests of holders of present PI Trust Claims, and the Futures Representative, who represents the interests of holders of PI Trust Claims that shall be asserted in the future. The Trustees shall obtain the consent of the TAC and the Futures Representative on any amendments to this TDP pursuant to Section 8.1 below, and on such other matters as are otherwise required below and in Section 2.2(f) of the PI Trust Agreement. The Trustees shall also consult with the TAC and the Futures Representative on such matters as are provided below and in Section 2.2(e) of the PI Trust Agreement. The initial Trustees, the initial members of the TAC and the initial Futures Representative are identified in the PI Trust Agreement.
- 3.2 Consent and Consultation Procedures. In those circumstances in which consultation or consent is required, the Trustees shall provide written notice to the TAC and the

Futures Representative of the specific amendment or other action that is proposed. The Trustees shall not implement such amendment nor take such action unless and until the parties have engaged in the Consultation Process described in Sections 5.7(a) and 6.6(a), or the Consent Process described in Sections 5.7(b) and 6.6(b), of the PI Trust Agreement, respectively.

SECTION IV

Payment Percentage; Periodic Estimates

- 4.1 Uncertainty of USG's Personal Injury Asbestos Liabilities. As discussed above, there is inherent uncertainty regarding USG's total asbestos-related tort liabilities, as well as the total value of the assets available to the PI Trust to pay PI Trust Claims. Consequently, there is inherent uncertainty regarding the amounts that holders of PI Trust Claims shall receive. To seek to ensure substantially equivalent treatment of all present and future PI Trust Claims, the Trustees must determine from time to time the percentage of full liquidated value that holders of present and future PI Trust Claims shall be likely to receive, *i.e.*, the "Payment Percentage" described in Section 2.3 above and Section 4.2 below.
- 4.2 Computation of Payment Percentage. As provided in Section 2.3 above, the Initial Payment Percentage shall be 45% and shall apply to all PI Trust Voting Claims as defined in Section 2.3 above, unless the Trustees, with the consent of the TAC and the Futures Representative, determine that the Initial Payment Percentage should be changed to assure that the PI Trust shall be in a financial position to pay holders of unliquidated and/or unpaid PI Trust Voting Claims and present and future PI Trust Claims in substantially the same manner.

In making any such adjustment, the Trustees, the TAC and the Futures Representative shall take into account the fact that the holders of PI Trust Voting Claims voted on the Plan relying on the findings of experts that the Initial Payment Percentage represented a reasonably

reliable estimate of the PI Trust's total assets and liabilities over its life based on the best information available at the time, and shall thus give due consideration to the expectations of PI Trust Voting Claimants that the Initial Payment Percentage would be applied to their PI Trust Claims.

Except with respect to PI Trust Voting Claims to which the Initial Payment Percentage applies, the Payment Percentage shall be subject to change pursuant to the terms of this TDP and the PI Trust Agreement if the Trustees with the consent of the TAC and Futures Representative determine that an adjustment is required. No less frequently than once every three (3) years, commencing with the first day of January occurring after the Plan is consummated, the Trustees shall reconsider the then applicable Payment Percentage to assure that it is based on accurate, current information and may, after such reconsideration, change the Payment Percentage if necessary with the consent of the TAC and the Futures Representative. The Trustees shall also reconsider the then applicable Payment Percentage at shorter intervals if they deem such reconsideration to be appropriate or if requested to do so by the TAC or the Futures Representative.

The Trustees must base their determination of the Payment Percentage on current estimates of the number, types, and values of present and future PI Trust Claims, the value of the assets then available to the PI Trust for their payment, all anticipated administrative and legal expenses, and any other material matters that are reasonably likely to affect the sufficiency of funds to pay a comparable percentage of full value to all holders of PI Trust Claims. When making these determinations, the Trustees shall exercise common sense and flexibly evaluate all relevant factors. The Payment Percentage applicable to Category A or Category B claims may not be reduced to alleviate delays in payments of claims in the other Category; both Categories

of claims shall receive the same Payment Percentage, but the payment may be deferred as needed, and a Reduced Payment Option may be instituted as described in Section 2.5 above.

4.3 Applicability of the Payment Percentage. Except as set forth below in this Section 4.3 with respect to supplemental payments, no holder of a PI Trust Voting Claim, other than a PI Trust Voting Claim for Other Asbestos Disease (Disease Level I – Cash Discount Payment) as defined in Section 5.3(a)(3) below, shall receive a payment that exceeds the Initial Payment Percentage times the liquidated value of the claim. Except as otherwise provided (a) in Section 5.1(c) below for PI Trust Claims involving deceased or incompetent claimants for which approval of the PI Trust's offer by a court or through a probate process is required and (b) in the paragraph below with respect to Released Claims, no holder of any other PI Trust Claim, other than a PI Trust Claim for Other Asbestos Disease (Disease Level I - Cash Discount Payment), shall receive a payment that exceeds the liquidated value of the claim times the Payment Percentage in effect at the time of payment; provided, however, that if there is a reduction in the Payment Percentage, the Trustees, in their sole discretion, may cause the PI Trust to pay a PI Trust Claim based on the Payment Percentage that was in effect prior to the reduction if such PI Trust Claim was filed and actionable with the PI Trust ninety (90) days or more prior to the date the Trustees proposed the new Payment Percentage in writing to the TAC and the Future Claimants' Representative (the "Proposal Date") and the processing of such claim was unreasonably delayed due to circumstances beyond the control of the claimant or the claimant's counsel, but only if such claim had no deficiencies for the ninety (90) days prior to the Proposal Date. PI Trust Claims involving Other Asbestos Disease (Disease Level I - Cash Discount Payment) shall not be subject to the Payment Percentage, but shall instead be paid the full amount of their Scheduled Value as set forth in Section 5.3(a)(3) below.

If a redetermination of the Payment Percentage has been proposed in writing by the Trustees to the TAC and the Futures Representative but has not yet been adopted, the claimant shall receive the lower of the current Payment Percentage or the proposed Payment Percentage. However, if the proposed Payment Percentage was the lower amount but was not subsequently adopted, the claimant shall thereafter receive the difference between the lower proposed amount and the higher current amount. Conversely, if the proposed Payment Percentage was the higher amount and was subsequently adopted, the claimant shall thereafter receive the difference between the lower current amount and the higher adopted amount.

Notwithstanding anything contained herein, if the proposed Payment Percentage is lower than the current Payment Percentage, a claimant whose PI Trust Claim was liquidated prior to the Proposal Date and who either (a) transmitted⁵ an executed release to the PI Trust prior to the Proposal Date or (b) with respect to those claimants who had received releases fewer than thirty (30) days prior to the Proposal Date, transmitted an executed release to the PI Trust within thirty (30) days of the claimant's receipt of the release (the claims described in (a) and (b) are collectively referred to here in as the "Released Claims") shall be paid based on the current Payment Percentage (the "Released Claims Payment Percentage"). For purposes hereof, (a) a claimant represented by counsel shall be deemed to have received a release on the date that the claimant's counsel receives the release, (b) if the PI Trust transmits a release electronically, the release shall be deemed to have been received on the date the PI Trust transmits the offer notification, and (c) if the PI Trust places the release in the U.S. mail, postage prepaid, the release shall be deemed to have been received three (3) business days after

such mailing date. A delay in the payment of the Released Claims for any reason, including delays resulting from limitations on payment amounts in a given year pursuant to Sections 2.4 and 2.5 hereof, shall not affect the rights of the holders of the Released Claims to be paid based on the Released Claims Payment Percentage.

At least thirty (30) days prior to proposing in writing to the TAC and the Future Claimants' Representative a change in the Payment Percentage, the Trustees shall issue a written notice to claimants or claimants' counsel indicating that the Trustees are reconsidering such Payment Percentage. During the period of time when the Trustees are contemplating a change in the Payment Percentage, the PI Trust shall continuing processing claims and making offers in a manner consistent with its normal course of business.

There is uncertainty surrounding the amount of the PI Trust's future assets. There is also uncertainty surrounding the totality of the PI Trust Claims to be paid over time, as well as the extent to which changes in existing federal and state law could affect the PI Trust's liabilities under this TDP. If the value of the PI Trust's future assets increases significantly and/or if the value or volume of PI Trust Claims actually filed with the PI Trust is significantly lower than originally estimated, the PI Trust shall use those proceeds and/or claims savings, as the case may be, first to maintain the Payment Percentage then in effect.

If the Trustees, with the consent of the TAC and the Futures Representative, make a determination to increase the Payment Percentage due to a material change in the estimates of the PI Trust's future assets and/or liabilities, the Trustees shall also make supplemental payments to all claimants who previously liquidated their claims against the PI Trust and received

⁵ For purposes of this sentence, "transmitted" is defined as the date/time postmarked if submitted by mail or the date/time uploaded if submitted electronically.

payments based on a lower Payment Percentage. The amount of any such supplemental payment shall be the liquidated value of the claim in question times the newly adjusted Payment Percentage, less all amounts previously paid to the claimant with respect to the claim (excluding the portion of such previously paid amounts that was attributable to any sequencing adjustment paid pursuant to Section 7.5 below).

The Trustees' obligation to make a supplemental payment to a claimant shall be suspended in the event the payment in question would be less than \$100.00, and the amount of the suspended payment shall be added to the amount of any prior supplemental payment/payments that was/were also suspended because it/they would have been less than \$100.00. However, the Trustees' obligation shall resume and the Trustees shall pay any such aggregate supplemental payments due the claimant at such time that the total exceeds \$100.00.

SECTION V

Resolution of PI Trust Claims.

- 5.1 Ordering, Processing and Payment of Claims.
 - 5.1(a) Ordering of Claims.

5.1(a)(1) Establishment of the FIFO Processing Queue. The PI Trust shall order claims that are sufficiently complete to be reviewed for processing purposes on a FIFO basis except as otherwise provided herein (the "FIFO Processing Queue"). For all claims filed on or before the date six (6) months after the date that the PI Trust first makes available the proof of claim forms and other claims materials required to file a claim with the PI Trust (the "Initial Claims Filing Date"), a claimant's position in the FIFO Processing Queue shall be determined as of the earliest of (i) the date prior to June 25, 2001 (the "Petition Date") that the specific claim was either filed against USG or A.P. Green in the tort system or was actually

submitted to USG or A.P. Green pursuant to an administrative settlement agreement; (ii) the date before the Petition Date that the asbestos claim was filed against another defendant in the tort system if at the time the claim was subject to a tolling agreement with USG or A.P. Green; (iii) the date after the Petition Date but before the date that the PI Trust first makes available the proof of claim forms and other claims materials required to file a claim with the PI Trust that the asbestos claim was filed against another defendant in the tort system; (iv) the date after the Petition Date but before January 30, 2006 that a proof of claim was filed by the claimant against USG in this Chapter 11 proceeding; or (v) the date a ballot was submitted on behalf of the claimant for purposes of voting to accept or reject the Plan pursuant to the voting procedures approved by the Bankruptcy Court.

Following the Initial Claims Filing Date, the claimant's position in the FIFO Processing Queue shall be determined by the date the claim is filed with the PI Trust. If any claims are filed on the same date, the claimant's position in the FIFO Processing Queue shall be determined by the date of the diagnosis of the asbestos-related disease. If any claims are filed and diagnosed on the same date, the claimant's position in the FIFO Processing Queue shall be determined by the claimant's date of birth, with older claimants given priority over younger claimants.

5.1(a)(2) Effect of Statutes of Limitation and Repose. All unliquidated PI Trust Claims must meet either (i) for claims first filed in the tort system against USG or A.P. Green prior to the Petition Date, the applicable federal, state and foreign statute of limitation and repose that was in effect at the time of the filing of the claim in the tort system, or (ii) for claims not filed against USG or A.P. Green in the tort system prior to the Petition Date, the applicable federal, state or foreign statute of limitation that was in effect at the time of the filing with the PI Trust. However, the running of the relevant statute of limitation shall be tolled as of the earliest

of (A) the actual filing of the claim against USG or A.P. Green prior to the Petition Date, whether in the tort system or by submission of the claim to USG or A.P. Green pursuant to an administrative settlement agreement; (B) the tolling of the claim against USG or A.P. Green prior to the Petition Date by an agreement or otherwise, provided such tolling is still in effect on the Petition Date; or (C) the Petition Date.

If a PI Trust Claim meets any of the tolling provisions described in the preceding sentence and the claim was not barred by the applicable federal, state or foreign statute of limitation at the time of the tolling event, it shall be treated as timely filed if it is actually filed with the PI Trust within three (3) years after the Initial Claims Filing Date. In addition, any claims that were first diagnosed after the Petition Date, irrespective of the application of any relevant federal, state or foreign statute of limitation or repose, may be filed with the PI Trust within three (3) years after the date of diagnosis or within three (3) years after the Initial Claims Filing Date, whichever occurs later. However, the processing of any PI Trust Claim by the PI Trust may be deferred at the election of the claimant pursuant to Section 6.3 below.

5.1(b) Processing of Claims. As a general practice, the PI Trust shall review its claims files on a regular basis and notify all claimants whose claims are likely to come up in the FIFO Processing Queue in the near future.

5.1(c) Payment of Claims. PI Trust Claims that have been liquidated by the Expedited Review Process as provided in Section 5.3(a) below, by the Individual Review Process as provided in Section 5.3(b) below, by arbitration as provided in Section 5.10 below, or by litigation in the tort system provided in Section 5.11 below, shall be paid in FIFO order based on the date their liquidation became final (the "FIFO Payment Queue"), all such payments being subject to the applicable Payment Percentage, the Maximum Available Payment, the

Claims Payment Ratio, and the sequencing adjustment provided for in Section 7.5 below, except as otherwise provided herein. Pre-Petition Liquidated Claims, as defined in Section 5.2 below, shall be subject to the Maximum Annual Payment and Payment Percentage limitations, but not to the Maximum Available Payment and Claims Payment Ratio provisions set forth above.

Where the claimant is deceased or incompetent, and the settlement and payment of his or her claim must be approved by a court of competent jurisdiction or through a probate process prior to acceptance of the claim by the claimant's representative, an offer made by the PI Trust on the claim shall remain open so long as proceedings before that court or in that probate process remain pending, provided that the PI Trust has been furnished with evidence that the settlement offer has been submitted to such court or in the probate process for approval. If the offer is ultimately approved by the court or through the probate process and accepted by the claimant's representative, the PI Trust shall pay the claim in the amount so offered, multiplied by the Payment Percentage in effect at the time the offer was first made.

If any claims are liquidated on the same date, the claimant's position in the FIFO Payment Queue shall be determined by the date of the diagnosis of the claimant's asbestos-related disease. If any claims are liquidated on the same date and the respective holders' asbestos-related diseases were diagnosed on the same date, the position of those claims in the FIFO Payment Queue shall be determined by the PI Trust based on the dates of the claimants' birth, with older claimants given priority over younger claimants.

5.2 Resolution of Pre-Petition Liquidated PI Trust Claims.

5.2(a) Processing and Payment. As soon as practicable after the Effective Date, the PI Trust shall pay, upon submission by the claimant of the appropriate documentation, all PI Trust Claims that were liquidated by (i) a binding settlement agreement for the particular claim

entered into prior to the Petition Date that is judicially enforceable by the claimant, (ii) a jury verdict or non-final judgment in the tort system obtained prior to the Petition Date, or (iii) by a judgment that became final and non-appealable prior to the Petition Date (collectively "Pre-Petition Liquidated Claims"). In order to receive payment from the Trust, the holder of a Pre-Petition Liquidated Claim must submit all documentation necessary to demonstrate to the Trust that the claim was liquidated in the manner described in the preceding sentence, which documentation shall include (A) a court authenticated copy of the jury verdict (if applicable), a non-final judgment (if applicable) or a final judgment (if applicable) and (B) the name, social security number and date of birth of the claimant and the name and address of the claimant's lawyer.

The liquidated value of a Pre-Petition Liquidated Claim shall be the unpaid portion of the amount agreed to in the binding settlement agreement, the unpaid portion of the amount awarded by the jury verdict or non-final judgment, or the unpaid portion of the amount of the final judgment, as the case may be, plus interest, if any, that has accrued on that amount in accordance with the terms of the agreement, if any, or under applicable state law for settlements or judgments as of the Petition Date; however, except as otherwise provided in Section 7.4 below, the liquidated value of a Pre-Petition Liquidated Claim shall not include any punitive or exemplary damages. In addition, the amounts payable with respect to such claims shall not be subject to or taken into account in consideration of the Claims Payment Ratio and the Maximum Available Payment limitations, but shall be subject to the Maximum Annual Payment and Payment Percentage provisions. In the absence of a Final Order of the Bankruptcy Court determining whether a settlement agreement is binding and judicially enforceable, a dispute between the claimant and the PI Trust over this issue shall be resolved pursuant to the same

procedures in this TDP that are provided for resolving the validity and/or liquidated value of a PI Trust Claim (i.e., arbitration and litigation in the tort system as set forth in Sections 5.10 and 5.11 below).

Pre-Petition Liquidated Claims shall be processed and paid in accordance with their order in a separate FIFO queue to be established by the PI Trust based on the date the PI Trust received all required documentation for the particular claim. If any Pre-Petition Liquidated Claims were filed on the same date, the claimants' position in the FIFO queue for such claims shall be determined by the date on which the claim was liquidated. If any Pre-Petition Liquidated Claims were both filed and liquidated on the same dates, the position of the claimants in the FIFO queue shall be determined by the dates of the claimants' birth, with older claimants given priority over younger claimants.

- 5.2(b) Marshalling of Security. Holders of Pre-Petition Liquidated Claims that are secured by letters of credit, appeal bonds, or other security or sureties shall first exhaust their rights against any applicable security or surety before making a claim against the PI Trust. Only in the event that such security or surety is insufficient to pay the Pre-Petition Liquidated Claim in full shall the deficiency be processed and paid as a Pre-Petition Liquidated Claim.
- 5.3 Resolution of Unliquidated PI Trust Claims. Within six (6) months after the establishment of the PI Trust, the Trustees, with the consent of the TAC and the Futures Representative, shall adopt procedures for reviewing and liquidating all unliquidated PI Trust Claims, which shall include deadlines for processing such claims. Such procedures shall also require that claimants seeking resolution of unliquidated PI Trust Claims must first file a proof of claim form, together with the required supporting documentation, in accordance with the

provisions of Sections 6.1 and 6.2 below. It is anticipated that the PI Trust shall provide an initial response to the claimant within six (6) months of receiving the proof of claim form.

The proof of claim form shall require the claimant to assert his or her claim for the highest Disease Level for which the claim qualifies at the time of filing. Irrespective of the Disease Level alleged on the proof of claim form, all claims shall be deemed to be a claim for the highest Disease Level for which the claim qualifies at the time of filing, and all lower Disease Levels for which the claim may also qualify at the time of filing or in the future shall be treated as subsumed into the higher Disease Level for both processing and payment purposes.

Upon filing of a valid proof of claim form with the required supporting documentation, the claimant shall be placed in the FIFO Processing Queue in accordance with the ordering criteria described in Section 5.1(a) above. The PI Trust shall provide the claimant with six (6) months notice of the date by which it expects to reach the claim in the FIFO Queue, following which the claimant shall promptly (i) advise the PI Trust whether the claim should be liquidated under the PI Trust's Expedited Review Process described in Section 5.3(a) below or, in certain circumstances, under the PI Trust's Individual Review Process described in Section 5.3(b) below; (ii) provide the PI Trust with any additional medical and/or exposure evidence that was not provided with the original claim submission; and (iii) advise the PI Trust of any change in the claimant's Disease Level. If a claimant fails to respond to the PI Trust's notice prior to the reaching of the claim in the FIFO Queue, the PI Trust shall process and liquidate the claim under the Expedited Review Process based upon the medical/exposure evidence previously submitted by the claimant, although the claimant shall retain the right to request Individual Review as described in Section 5.3(b) below.

5.3(a) Expedited Review Process.

5.3(a)(1) In General. The PI Trust's Expedited Review Process is designed primarily to provide an expeditious, efficient and inexpensive method for liquidating all PI Trust Claims (except those involving Lung Cancer 2 – Disease Level VI and all Foreign Claims (as defined below), which shall only be liquidated pursuant to the PI Trust's Individual Review Process), where the claim can easily be verified by the PI Trust as meeting the presumptive Medical/Exposure Criteria for the relevant Disease Level. Expedited Review thus provides claimants with a substantially less burdensome process for pursuing PI Trust Claims than does the Individual Review Process described in Section 5.3(b) below. Expedited Review is also intended to provide qualifying claimants a fixed and certain claims payment.

Thus, claims that undergo Expedited Review and meet the presumptive Medical/Exposure Criteria for the relevant Disease Level shall be paid the Scheduled Value for such Disease Level set forth in Section 5.3(a)(3) below. However, except for claims involving Other Asbestos Disease (Disease Level I), all claims liquidated by Expedited Review shall be subject to the applicable Payment Percentage, the Maximum Available Payment, and the Claims Payment Ratio limitations set forth above; provided, however, that Existing Claims shall not be subject to the Maximum Available Payment or the Claims Payment Ratio. Claimants holding claims that cannot be liquidated by Expedited Review because they do not meet the presumptive Medical/Exposure Criteria for the relevant Disease Level may elect the PI Trust's Individual Review Process set forth in Section 5.3(b) below.

Subject to the provisions of Section 5.8, the claimant's eligibility to receive the Scheduled Value for his or her PI Trust Claim pursuant to the Expedited Review Process shall be

determined solely by reference to the Medical/Exposure Criteria set forth below for each of the Disease Levels eligible for Expedited Review.

5.3(a)(2) Claims Processing Under Expedited Review. All claimants seeking liquidation of their claims pursuant to Expedited Review shall file the PI Trust's proof of claim form. As a proof of claim form is reached in the FIFO Processing Queue, the PI Trust shall determine whether the claim described therein meets the Medical/Exposure Criteria for one of the seven Disease Levels eligible for Expedited Review, and shall advise the claimant of its determination. If a Disease Level is determined, the PI Trust shall tender to the claimant an offer of payment of the Scheduled Value for the relevant Disease Level multiplied by the applicable Payment Percentage, together with a form of release of the PI Trust, the Debtors, the Debtors' Estates and the Reorganized Debtors approved by the PI Trust and reasonably acceptable in form and substance to the Reorganized Debtors, provided that the form of release included as Attachment B hereto is acceptable to the Reorganized Debtors. If the claimant accepts the Scheduled Value and returns the release properly executed, the claim shall be placed in the FIFO Payment Queue, following which the PI Trust shall disburse payment subject to the limitations of the Maximum Available Payment and Claims Payment Ratio, if any.

5.3(a)(3) Disease Levels, Scheduled Values and Medical/Exposure

Criteria. The eight Disease Levels covered by this TDP, together with the Medical/Exposure

Criteria for each and the Scheduled Values for the seven Disease Levels eligible for Expedited

Review, are set forth below. These Disease Levels, Scheduled Values, and Medical/Exposure

Criteria shall apply to all PI Trust Voting Claims filed with the PI Trust (except Pre-Petition

Liquidated Claims) on or before the Initial Claims Filing Date provided in Section 5.1 above for which the claimant elects the Expedited Review Process. Thereafter, for purposes of

administering the Expedited Review Process and with the consent of the TAC and the Futures Representative, the Trustees may add to, change, or eliminate Disease Levels, Scheduled Values, or Medical/Exposure Criteria; develop subcategories of Disease Levels, Scheduled Values or Medical/Exposure Criteria; or determine that a novel or exceptional asbestos personal injury claim is compensable even though it does not meet the Medical/Exposure Criteria for any of the then current Disease Levels.

<u>Disease Level</u> Mesothelioma (Level VIII)	Scheduled Value \$155,000	Medical/Exposure Criteria (1) Diagnosis ⁶ of mesothelioma; and (2) USG/A.P. Green Exposure as defined in Section 5.7(b)(3).
Lung Cancer 1 (Level VII)	\$45,000	(1) Diagnosis of a primary lung cancer plus evidence of an underlying Bilateral Asbestos-

⁶ The requirements for a diagnosis of an asbestos-related disease that may be compensated under the provisions of this TDP are set forth in Section 5.7 below.

⁷ Evidence of "Bilateral Asbestos-Related Nonmalignant Disease," for purposes of meeting the criteria for establishing Disease Levels I, II, III, V, and VII, means either (i) a chest X-ray read by a qualified B reader of 1/0 or higher on the ILO scale or (ii)(x) a chest X-ray read by a qualified B reader or other Qualified Physician, (y) a CT scan read by a Qualified Physician, or (z) pathology, in each case showing either bilateral interstitial fibrosis, bilateral pleural plaques, bilateral pleural thickening, or bilateral pleural calcification. Evidence submitted to demonstrate (i) or (ii) above must be in the form of a written report stating the results (e.g., in ILO report, written radiology report or a pathology report). Solely for asbestos claims filed against USG or another defendant in the tort system prior to the Petition Date, if an ILO reading is not available, either (i) a chest X-ray or a CT scan read by a Qualified Physician, or (ii) pathology, in each case showing bilateral interstitial fibrosis, bilateral pleural plaques, bilateral pleural thickening, or bilateral pleural calcification consistent with or compatible with a diagnosis of asbestos-related disease, shall be evidence of a Bilateral Asbestos-Related Nonmalignant Disease for purposes of meeting the presumptive medical requirements of Disease Levels I, II, III, V and VII. Pathological proof of asbestosis may be based on the pathological grading system for asbestosis described in the Special Issue of the Archives of Pathology and Laboratory Medicine, "Asbestosassociated Diseases," Vol. 106, No. 11, App. 3 (October 8, 1982). For all purposes of this TDP, a "Qualified Physician" is a physician who is board-certified (or in the case of Canadian Claims or Foreign Claims, a physician who is certified or qualified under comparable medical standards or criteria of the jurisdiction in question) in one or more relevant specialized fields of medicine such as pulmonology, radiology, internal medicine or occupational medicine; provided, however, subject to the provisions of Section 5.8, that the requirement for board certification in this

Disease Level

Scheduled Value

Medical/Exposure Criteria

Related Nonmalignant Disease⁷, (2) six months USG/A.P. Green Exposure prior to December 31, 1982 (in the case of USG) and January 2, 1968 (in the case of A.P. Green), (3) Significant Occupational Exposure⁸ to asbestos, and (4) supporting medical documentation establishing asbestos exposure as a contributing factor in causing the lung cancer in question.

Lung Cancer 2 (Level VI)

None

(1) Diagnosis of a primary lung cancer; (2) USG/A.P. Green Exposure prior to December 31, 1982 (in the case of USG) and January 2, 1968 (in the case of A.P. Green), and (3) supporting medical documentation establishing asbestos exposure as a contributing factor in causing the lung cancer in question.

Lung Cancer 2 (Level VI) claims are claims that do not meet the more stringent medical and/or exposure requirements of Lung Cancer 1 (Level VII) claims. All claims in this Disease Level shall be individually evaluated. The estimated likely average of the individual evaluation awards for this category is \$15,000, with such awards capped at \$35,000 unless the claim qualifies for Extraordinary Claim treatment.

Level VI claims that show no evidence of either an underlying Bilateral Asbestos-Related Nonmalignant Disease or Significant Occupational Exposure may be individually evaluated, although it is not expected that such claims shall be treated as having any significant value, especially if the claimant is also a Smoker. In any event, no presumption of

provision shall not apply to otherwise qualified physicians whose X-ray and/or CT scan readings are submitted for deceased holders of PI Trust Claims.

⁸ The term "Significant Occupational Exposure" is defined in Section 5.7(b)(2) below.

⁹ There is no distinction between Non-Smokers and Smokers for either Lung Cancer 1 (Level VII) or Lung Cancer 2 (Level VI), although a claimant who meets the more stringent requirements of Lung Cancer 1 (Level VII) (evidence of an underlying Bilateral Asbestos-Related Nonmalignant Disease plus Significant Occupational Exposure), and who is also a Non-

Disease Level	Scheduled Value	Medical/Exposure Criteria validity shall be available for any claims in this category.
Other Cancer (Level V)	\$15,000	(1) Diagnosis of a primary colo-rectal, laryngeal, esophageal, pharyngeal, or stomach cancer, plus evidence of an underlying Bilateral Asbestos-Related Nonmalignant Disease, (2) six months USG/A.P. Green Exposure prior to December 31, 1982 (in the case of USG) and January 2, 1968 (in the case of A.P. Green), (3) Significant Occupational Exposure to asbestos, and (4) supporting medical documentation establishing asbestos exposure as a contributing factor in causing the other cancer in question.
Severe Asbestosis (Level IV)	\$30,000	(1) Diagnosis of asbestosis with ILO of 2/1 or greater, or asbestosis determined by pathological evidence of asbestos, plus (a) TLC less than 65%, or (b) FVC less than 65% and FEV1/FVC ratio greater than 65%, (2) six months USG/A.P. Green Exposure prior to December 31, 1982 (in the case of USG) and January 2, 1968 (in the case of A.P. Green), (3) Significant Occupational Exposure to asbestos, and (4) supporting medical documentation establishing asbestos exposure as a contributing factor in causing the pulmonary disease in question.
Asbestosis/Pleural Disease (Level III)	\$8,300	(1) Diagnosis of Bilateral Asbestos-Related Nonmalignant Disease, plus (a) TLC less than 80%, or (b) FVC less than 80% and FEV1/FVC ratio greater than or equal to 65%, and (2) six months USG/A.P. Green Exposure prior to December 31, 1982 (in the case of USG) and January 2, 1968 (in the case of A.P. Green), (3) Significant Occupational Exposure to asbestos, and (4) supporting medical documentation establishing asbestos exposure as a contributing

Smoker, may wish to have his or her claim individually evaluated by the PI Trust. In such a case, absent circumstances that would otherwise reduce the value of the claim, it is anticipated that the liquidated value of the claim might well exceed the \$45,000 Scheduled Value for Lung Cancer 1 (Level VII) shown above. "Non-Smoker" means a claimant who either (a) never smoked or (b) has not smoked during any portion of the twelve (12) years immediately prior to the diagnosis of the lung cancer.

Disease Level	Scheduled Value	Medical/Exposure Criteria factor in causing the pulmonary disease in question.
Asbestosis/Pleural Disease (Level II)	\$2,625	(1) Diagnosis of a Bilateral Asbestos-Related Nonmalignant Disease, and (2) six months USG/A.P. Green Exposure prior to December 31, 1982 (in the case of USG) and January 2, 1968 (in the case of A.P. Green), and (3) five years cumulative occupational exposure to asbestos.
Other Asbestos Disease (Level I – Cash Discount Payment)	\$400	(1) Diagnosis of a Bilateral Asbestos-Related Nonmalignant Disease or an asbestos-related malignancy other than mesothelioma, and (2) USG/A.P. Green Exposure prior to December 31, 1982 (in the case of USG) and January 2, 1968 (in the case of A.P. Green).

5.3(b) Individual Review Process.

claimant may elect to have his or her PI Trust Claim reviewed for purposes of determining whether the claim would be compensable in the tort system even though it does not meet the presumptive Medical/Exposure Criteria for any of the Disease Levels set forth in Section 5.3(a)(3) above. In addition or alternatively, a claimant may elect to have a claim undergo the Individual Review Process for purposes of determining whether the liquidated value of claim involving Disease Levels IV, V, VII or VIII exceeds the Scheduled Value for the relevant Disease Level also set forth in said provision. However, until such time as the PI Trust has made an offer on a claim pursuant to Individual Review, the claimant may change his or her Individual Review election and have the claim liquidated pursuant to the PI Trust's Expedited Review Process. In the event of such a change in the processing election, the claimant shall nevertheless retain his or her place in the FIFO Processing Queue.

The liquidated value of all Foreign Claims payable under this TDP shall be established only under the PI Trust's Individual Review process. PI Trust Claims of individuals exposed in Canada who were resident in Canada when such claims were filed ("Canadian Claims") shall not be considered Foreign Claims hereunder and shall be eligible for liquidation under the Expedited Review Process. Accordingly, a "Foreign Claim" is a PI Trust Claim with respect to which the claimant's exposure to an asbestos-containing product or conduct for which USG or A.P. Green (prior to January 2, 1968) has legal responsibility occurred outside of the United States and its Territories and Possessions, and outside of the Provinces and Territories of Canada.

In reviewing Foreign Claims, the PI Trust shall take into account all relevant procedural and substantive legal rules to which the claims would be subject in the Claimant's Jurisdiction as defined in Section 5.3(b)(2) below. The PI Trust shall determine the liquidated value of Foreign Claims based on historical settlements and verdicts in the Claimant's Jurisdiction as well as the other valuation factors set forth in Section 5.3(b)(2) below.

For purposes of the Individual Review process for Foreign Claims, the Trustees, with the consent of the TAC and the Futures Representative, may develop separate Medical/Exposure Criteria and standards, as well as separate requirements for physician and other professional qualifications, which shall be applicable to all Foreign Claims channeled to the PI Trust; provided however, that such criteria, standards or requirements shall not effectuate substantive changes to the claims eligibility requirements under this TDP, but rather shall be made only for the purpose of adapting those requirements to the particular licensing provisions and/or medical customs or practices of the foreign country in question.

At such time as the PI Trust has sufficient historical settlement, verdict and other valuation data for claims from a particular foreign jurisdiction, the Asbestos Trustees, with the

consent of the TAC and the Futures Representative, may also establish a separate valuation matrix for any such Foreign Claims based on that data.

5.3(b)(1)(A) Review of Medical/Exposure Criteria. The PI

Trust's Individual Review Process provides a claimant with an opportunity for individual consideration and evaluation of a PI Trust Claim that fails to meet the presumptive Medical/Exposure Criteria for Disease Levels I–V, VII or VIII. In such a case, the PI Trust shall either deny the claim or, if the PI Trust is satisfied that the claimant has presented a claim that would be cognizable and valid in the tort system, the PI Trust can offer the claimant a liquidated value amount up to the Scheduled Value for that Disease Level.

claims in the five more serious Disease Levels IV-VIII shall also be eligible to seek Individual Review of the liquidated value of their claims, as well as of their medical/exposure evidence.

The Individual Review Process is intended to result in payments equal to the full liquidated value for each claim multiplied by the Payment Percentage; however, the liquidated value of any PI Trust Claim that undergoes Individual Review may be determined to be less than the Scheduled Value the claimant would have received under Expedited Review. Moreover, the liquidated value for a claim involving Disease Levels IV-VIII shall not exceed the Maximum Value for the relevant Disease Level set forth in Section 5.3(b)(3) below, unless the claim meets the requirements of an Extraordinary Claim described in Section 5.4(a) below, in which case its liquidated value cannot exceed the Maximum Value set forth in that provision for such claims. Because the detailed examination and valuation process pursuant to Individual Review requires substantial time and effort, claimants electing to undergo the Individual Review Process may be paid the liquidated value of their PI Trust Claims later than would have been the case had the

claimant elected the Expedited Review Process. Subject to the provisions of Section 5.8, the PI Trust shall devote reasonable resources to the review of all claims to ensure that there is a reasonable balance maintained in reviewing all classes of claims.

5.3(b)(2) Valuation Factors to Be Considered in Individual Review.

The PI Trust shall liquidate the value of each PI Trust Claim that undergoes Individual Review based on the historic liquidated values of other similarly situated claims in the tort system for the same Disease Level. The PI Trust shall thus take into consideration all of the factors that affect the severity of damages and values within the tort system including, but not limited to, credible evidence of (i) the degree to which the characteristics of a claim differ from the presumptive Medical/Exposure Criteria for the Disease Level in question; (ii) factors such as the claimant's age, disability, employment status, disruption of household, family or recreational activities, dependencies, special damages, and pain and suffering; (iii) whether the claimant's damages were (or were not) caused by asbestos exposure, including exposure to an asbestos-containing product or to conduct for which USG or A.P. Green has legal responsibility prior to December 31, 1982 (in the case of USG) and January 2, 1968 (in the case of A.P. Green) (for example, alternative causes, and the strength of documentation of injuries); (iv) the industry of exposure; (v) settlement and verdict histories and other law firms' experience in the Claimant's Jurisdiction for similarly situated claims; and (vi) settlement and verdict histories for the claimant's law firm for similarly situated claims.

For these purposes, the "Claimant's Jurisdiction" is the jurisdiction in which the claim was filed (if at all) against USG or A.P. Green in the tort system prior to the Petition Date. If the claim was not filed against USG or A.P. Green in the tort system prior to the Petition Date, the claimant may elect as the Claimant's Jurisdiction either (i) the jurisdiction in which the claimant

resides at the time of diagnosis or when the claim is filed with the PI Trust; or (ii) a jurisdiction in which the claimant experienced exposure to an asbestos-containing product or to conduct for which USG or A.P. Green has legal responsibility.

With respect to the "Claimant's Jurisdiction" in the event a personal representative or authorized agent makes a claim under this TDP for wrongful death with respect to which the governing law of the Claimant's Jurisdiction could only be the Alabama Wrongful Death Statute, the Claimant's Jurisdiction for such claim shall be the Commonwealth of Pennsylvania, and such claimant's damages shall be determined pursuant to the statutory and common laws of the Commonwealth of Pennsylvania without regard to its choice of law principles. The choice of law provision in Section 7.4 below applicable to any claim with respect to which, but for this choice of law provision, the applicable law of the Claimant's Jurisdiction pursuant to Section 5.3(b)(2) is determined to be the Alabama Wrongful Death Statute, shall only govern the rights between the PI Trust and the claimant, and, to the extent the PI Trust seeks recovery from any entity that provided insurance coverage to USG, the Alabama Wrongful Death Statute shall govern.

5.3(b)(3) Scheduled, Average and Maximum Values. The Scheduled, Average and Maximum Values for claims involving Disease Levels I-VIII are the following:

Scheduled Value	Average Value	Maximum Value
\$155,000	\$225,000	\$450,000
\$45,000	\$55,000	\$100,000
None	\$15,000	\$35,000
\$15,000	\$18,000	\$35,000
\$30,000	\$35,000	\$50,000
	\$155,000 \$45,000 None \$15,000	\$155,000 \$225,000 \$45,000 \$55,000 None \$15,000 \$15,000 \$18,000

Scheduled Disease	Scheduled Value	Average Value	Maximum Value
Asbestosis/Pleural Disease (Level III)	\$8,300	None	None
Asbestosis/Pleural Disease (Level II)	\$2,625	None	None
Other Asbestos Disease – Cash Discount Payment (Level I)	\$400	None	None

These Scheduled Values, Average Values and Maximum Values shall apply to all PI Trust Voting Claims other than Pre-Petition Liquidated Claims filed with the PI Trust on or before the Initial Claims Filing Date as provided in Section 5.1 above. Thereafter, the PI Trust, with the consent of the TAC and the Futures Representative pursuant to Sections 5.7(b) and 6.6(b) of the PI Trust Agreement, may change these valuation amounts for good cause and consistent with other restrictions on the amendment power.

5.4 Categorizing Claims as Extraordinary and/or Exigent Hardship.

5.4(a) Extraordinary Claims. "Extraordinary Claim" means a PI Trust Claim that otherwise satisfies the Medical Criteria for Disease Levels IV—VIII, and that is held by a claimant whose exposure to asbestos (i) occurred predominantly as a result of working in a manufacturing facility of USG or A.P. Green (prior to January 2, 1968) during a period in which USG or A.P. Green (prior to January 2, 1968) was manufacturing asbestos-containing products at that facility, or (ii) was at least 75% the result of exposure to an asbestos-containing product or to conduct for which USG or A.P. Green (prior to January 2, 1968) has legal responsibility, and in either case there is little likelihood of a substantial recovery elsewhere. All such Extraordinary Claims shall be presented for Individual Review and, if valid, shall be entitled to an award of up to a Maximum Value of five (5) times the Scheduled Value set forth in Section 5.3(b)(3) for

claims qualifying for Disease Levels IV–V, VII and VIII, and five (5) times the Average Value for claims in Disease Level VI, multiplied by the applicable Payment Percentage.

Any dispute as to Extraordinary Claim status shall be submitted to a special Extraordinary Claims Panel established by the PI Trust with the consent of the TAC and the Futures Representative. All decisions of the Extraordinary Claims Panel shall be final and not subject to any further administrative or judicial review. An Extraordinary Claim, following its liquidation, shall be placed in the FIFO Queue ahead of all other PI Trust Claims except Pre-Petition Liquidated Claims, Disease Level I Claims, Existing Claims and Exigent Hardship Claims, which shall be paid first in that order in said Queue, based on its date of liquidation, subject to the Maximum Available Payment and Claims Payment Ratio described above.

5.4(b) Exigent Hardship Claims. At any time the PI Trust may liquidate and pay PI Trust Claims that qualify as Exigent Hardship Claims as defined below. Such claims may be considered separately no matter what the order of processing otherwise would have been under this TDP. An Exigent Hardship Claim, following its liquidation, shall be placed first in the FIFO Payment Queue ahead of all other liquidated PI Trust Claims except Pre-Petition Liquidated Claims, Disease Level I Claims and Existing Claims and shall be subject to the Maximum Available Payment and Claims Payment Ratio described above. A PI Trust Claim qualifies for payment as an Exigent Hardship Claim if the claim meets the Medical/Exposure Criteria for Severe Asbestosis (Disease Level IV) or an asbestos-related malignancy (Disease Levels V–VIII), and the PI Trust, in its sole discretion, determines (i) that the claimant needs financial assistance on an immediate basis based on the claimant's expenses and all sources of available income, and (ii) that there is a causal connection between the claimant's dire financial condition and the claimant's asbestos-related disease.

- resulting solely from exposure to an occupationally exposed person, such as a family member, the claimant must seek Individual Review of his or her claim pursuant to Section 5.3(b) above. In such a case, the claimant must establish that the occupationally exposed person would have met the exposure requirements under this TDP that would have been applicable had that person filed a direct claim against the PI Trust. In addition, the claimant with secondary exposure must establish that he or she is suffering from one of the eight Disease Levels described in Section 5.3(a)(3) above or an asbestos-related disease otherwise compensable under this TDP, that his or her own exposure to the occupationally exposed person occurred within the same time frame as the occupationally exposed person was exposed to asbestos products manufactured, produced or distributed by USG or A.P. Green (prior to January 2, 1968) or to conduct for which USG or A.P. Green (prior to January 2, 1968) has legal responsibility, and that such secondary exposure was a cause of the claimed disease. All other liquidation and payment rights and limitations under this TDP shall be applicable to such claims.
- shall be treated as presumptively valid and paid by the PI Trust subject to the applicable Payment Percentage if (a) such claim satisfied the requirements of the Bar Date for such claims established by the Bankruptcy Court, if applicable, and is not otherwise disallowed by Section 502(e) of the Code or subordinated under Section 509(c) of the Code, and, and (b) the holder of such claim (the "Indirect Claimant") establishes to the satisfaction of the Trustees that (i) the Indirect Claimant has paid in full the liability and obligation of the PI Trust to the individual claimant to whom the PI Trust would otherwise have had a liability or obligation under these Procedures (the "Direct Claimant"), (ii) the Direct Claimant and the Indirect Claimant have

forever and fully released the PI Trust from all liability to the Direct Claimant, and (iii) the claim is not otherwise barred by a statute of limitation or repose or by other applicable law. In no event shall any Indirect Claimant have any rights against the PI Trust superior to the rights of the related Direct Claimant against the PI Trust, including any rights with respect to the timing, amount or manner of payment. In addition, no Indirect PI Trust Claim may be liquidated and paid in an amount that exceeds what the Indirect Claimant has actually paid the related Direct Claimant.

To establish a presumptively valid Indirect PI Trust Claim, the Indirect Claimant's aggregate liability for the Direct Claimant's claim must also have been fixed, liquidated and paid fully by the Indirect Claimant by settlement (with an appropriate full release in favor of the PI Trust) or a Final Order (as defined in the Plan) provided that such claim is valid under the applicable state law. In any case where the Indirect Claimant has satisfied the claim of a Direct Claimant against the PI Trust under applicable law by way of a settlement, the Indirect Claimant shall obtain for the benefit of the PI Trust a release in form and substance satisfactory to the Trustees.

If an Indirect Claimant cannot meet the presumptive requirements set forth above, including the requirement that the Indirect Claimant provide the PI Trust with a full release of the Direct Claimant's claim, the Indirect Claimant may request that the PI Trust review the Indirect PI Trust Claim individually to determine whether the Indirect Claimant can establish under applicable state law that the Indirect Claimant has paid all or a portion of a liability or obligation that the PI Trust had to the Direct Claimant as of the Effective Date of this TDP. If the Indirect Claimant can show that it has paid all or a portion of such a liability or obligation, the PI Trust shall reimburse the Indirect Claimant the amount of the liability or obligation so

paid, times the then applicable Payment Percentage. However, in no event shall such reimbursement to the Indirect Claimant be greater than the amount to which the Direct Claimant would have otherwise been entitled. Further, the liquidated value of any Indirect PI Trust Claim paid by the PI Trust to an Indirect Claimant shall be treated as an offset to or reduction of the full liquidated value of any PI Trust Claim that might be subsequently asserted by the Direct Claimant against the PI Trust.

Any dispute between the PI Trust and an Indirect Claimant over whether the Indirect Claimant has a right to reimbursement for any amount paid to a Direct Claimant shall be subject to the ADR Procedures provided in Section 5.10 below and set forth in Attachment A hereto. If such dispute is not resolved by said ADR Procedures, the Indirect Claimant may litigate the dispute in the tort system pursuant to Sections 5.11 and 7.6 below.

The Trust ees may develop and approve a separate proof of claim form for Indirect PI Trust Claims. Indirect PI Trust Claims that have not been disallowed, discharged, or otherwise resolved by prior order of the Bankruptcy Court shall be processed in accordance with procedures to be developed and implemented by the Trustees consistent with the provisions of this Section 5.6, which procedures (a) shall determine the validity, allowability and enforceability of such claims; and (b) shall otherwise provide the same liquidation and payment procedures and rights to the holders of such claims as the PI Trust would have afforded the holders of the underlying valid PI Trust Claims. Nothing in this TDP is intended to preclude a trust to which asbestos-related liabilities are channeled from asserting an Indirect PI Trust Claim against the PI Trust subject to the requirements set forth herein.

5.7 Evidentiary Requirements.

5.7(a) Medical Evidence.

5.7(a)(1) In General. All diagnoses of a Disease Level shall be accompanied by either (i) a statement by the physician providing the diagnosis that at least ten (10) years have elapsed between the date of first exposure to asbestos or asbestos-containing products and the diagnosis, or (ii) a history of the claimant's exposure sufficient to establish a 10-year latency period. A finding by a physician after the Effective Date that a claimant's disease is "consistent with" or "compatible with" asbestosis shall not alone be treated by the PI Trust as a diagnosis.

5.7(a)(1)(A) Disease Levels I–IV. Except for asbestos claims filed against USG or any other defendant in the tort system prior to the Petition Date, all diagnoses of a non-malignant asbestos-related disease (Disease Levels I–IV) shall be based in the case of a claimant who was living at the time the claim was filed, upon a physical examination of the claimant by the physician providing the diagnosis of the asbestos-related disease. All living claimants must also provide (i) for Disease Levels I–III, evidence of Bilateral Asbestos-Related Nonmalignant Disease (as defined in Footnote 3 above); (ii) for Disease Level IV, ¹⁰ an ILO reading of 2/1 or greater or pathological evidence of asbestosis, and (iii) for Disease Levels III and IV, pulmonary function testing. ¹¹

¹⁰ All diagnoses of Asbestos/Pleural Disease (Disease Levels II and III) not based on pathology shall be presumed to be based on findings of bilateral asbestosis or pleural disease, and all diagnoses of Mesothelioma (Disease Level VIII) shall be presumed to be based on findings that the disease involves a malignancy. However, the PI Trust may rebut such presumptions.

¹¹ "Pulmonary function testing" or "PFT" shall mean testing that is in material compliance with the quality criteria established by the American Thoracic Society ("ATS") and is performed on equipment which is in material compliance with ATS standards for technical quality and

In the case of a claimant who was deceased at the time the claim was filed, all diagnoses of a non-malignant asbestos-related disease (Disease Levels I–IV) shall be based upon either (i) a physical examination of the claimant by the physician providing the diagnosis of the asbestos-related disease; or (ii) pathological evidence of the non-malignant asbestos-related disease; or (iii) in the case of Disease Levels I–III, evidence of Bilateral Asbestos-Related Nonmalignant Disease (as defined in Footnote 3 above), and for Disease Level IV, either an ILO reading of 2/1 or greater or pathological evidence of asbestosis; and (iv) for either Disease Level III or IV, pulmonary function testing.

5.7(a)(1)(B) Disease Levels V-VIII. All diagnoses of an asbestos-related malignancy (Disease Levels V-VIII) shall be based upon either (i) a physical examination of the claimant by the physician providing the diagnosis of the asbestos-related disease, or (ii) a diagnosis of such a malignant Disease Level by a board-certified pathologist or by a pathology report prepared at or on behalf of a hospital accredited by the Joint Commission on Accreditation of Healthcare Organizations ("JCAHO").

5.7(a)(1)(C) Exception to the Exception for Certain Pre-Petition Claims. If the holder of a PI Trust Claim that was filed against USG or any other defendant in the tort system prior to the Petition Date has available a report of a diagnosing physician engaged by the holder or his or her law firm who conducted a physical examination of

calibration. PFT performed in a hospital accredited by the JCAHO, or performed, reviewed or supervised by a board certified pulmonologist or other Qualified Physician shall be presumed to comply with ATS standards, and the claimant may submit a summary report of the testing. If the PFT was not performed in an JCAHO-accredited hospital, or performed, reviewed or supervised by a board certified pulmonologist or other Qualified Physician, the claimant must submit the full report of the testing (as opposed to a summary report); provided, however, that if the PFT was conducted prior to the Effective Date of the Plan and the full PFT report is not available, the claimant must submit a declaration signed by a Qualified Physician or other qualified party, in

the holder as described in Sections 5.7(a)(1)(A), or if the holder has filed such medical evidence and/or a diagnosis of the asbestos-related disease by a physician not engaged by the holder or his or her law firm who conducted a physical examination of the holder with another asbestos-related personal injury settlement trust that requires such evidence, without regard to whether the claimant or the law firm engaged the diagnosing physician, the holder shall provide such medical evidence to the PI Trust notwithstanding the exception in Section 5.7(a)(1)(A).

5.7(a)(2) Credibility of Medical Evidence. Before making any payment to a claimant, the PI Trust must have reasonable confidence that the medical evidence provided in support of the claim is credible and consistent with recognized medical standards. The PI Trust may require the submission of X-rays, CT scans, detailed results of pulmonary function tests, laboratory tests, tissue samples, results of medical examination or reviews of other medical evidence, and may require that medical evidence submitted comply with recognized medical standards regarding equipment, testing methods and procedures to assure that such evidence is reliable. Medical evidence (i) that is of a kind shown to have been received in evidence by a state or federal judge at trial, (ii) that is consistent with evidence submitted to USG to settle for payment similar disease cases prior to USG's bankruptcy, or (iii) that is a diagnosis by a physician shown to have previously qualified as a medical expert with respect to the asbestosrelated disease in question before a state or federal judge, is presumptively reliable, although the PI Trust may seek to rebut the presumption. In addition, claimants who otherwise meet the requirements of this TDP for payment of a PI Trust Claim shall be paid irrespective of the results in any litigation at any time between the claimant and any other defendant in the tort system. However, any relevant evidence submitted in a proceeding in the tort system, other than any

the form provided by the PI Trust, certifying that the PFT was conducted in material compliance

findings of fact, a verdict, or a judgment, involving another defendant may be introduced by either the claimant or the PI Trust in any Individual Review proceeding conducted pursuant to 5.3(b) or any Extraordinary Claim proceeding conducted pursuant to 5.4(a).

5.7(b) Exposure Evidence.

5.7(b)(1) In General. As set forth above in Section 5.3(a)(3), to qualify for any Disease Level, the claimant must demonstrate a minimum exposure to an asbestoscontaining product manufactured, produced or distributed by USG or A.P. Green (prior to January 2, 1968) or to conduct for which USG or A.P. Green (prior to January 2, 1968) has legal responsibility. Claims based on conspiracy theories that involve no exposure to an asbestoscontaining product manufactured, produced or distributed by USG or A.P. Green are not compensable under this TDP. To meet the presumptive exposure requirements of Expedited Review set forth in Section 5.3(a)(3) above, the claimant must show (i) for all Disease Levels, USG/A.P. Green Exposure as defined in Section 5.7(b)(3) below prior to December 31, 1982 (in the case of USG) and January 2, 1968 (in the case of A.P. Green); (ii) for Asbestos/Pleural Disease Level II, six (6) months USG/A.P. Green Exposure prior to December 31, 1982 (in the case of USG) and January 2, 1968 (in the case of A.P. Green), plus five (5) years cumulative occupational asbestos exposure; and (iii) for Asbestosis/Pleural Disease (Disease Level III), Severe Asbestosis (Disease Level IV), Other Cancer (Disease Level V) or Lung Cancer 1 (Disease Level VII), the claimant must show six (6) months USG/A.P. Green Exposure prior to December 31, 1982 (in the case of USG) and January 2, 1968 (in the case of A.P. Green), plus Significant Occupational Exposure to asbestos. If the claimant cannot meet the relevant presumptive exposure requirements for a Disease Level eligible for Expedited Review, the

with ATS standards.

claimant may seek Individual Review pursuant to Section 5.3(b) of his or her claim based on exposure to an asbestos-containing product or to conduct for which USG or A.P. Green (prior to January 2, 1968) has legal responsibility.

5.7(b)(2) Significant Occupational Exposure. "Significant

Occupational Exposure" means employment for a cumulative period of at least five (5) years with a minimum of two (2) years prior to December 31, 1982, in an industry and an occupation in which the claimant (a) handled raw asbestos fibers on a regular basis; (b) fabricated asbestos-containing products so that the claimant in the fabrication process was exposed on a regular basis to raw asbestos fibers; (c) altered, repaired or otherwise worked with an asbestos-containing product such that the claimant was exposed on a regular basis to asbestos fibers; or (d) was employed in an industry and occupation such that the claimant worked on a regular basis in close proximity to workers engaged in the activities described in (a), (b) and/or (c).

5.7(b)(3) USG/A.P. Green Exposure. The claimant must demonstrate (i) meaningful and credible exposure, which occurred prior to December 31, 1982, to asbestos or asbestos-containing products supplied, specified, manufactured, installed, maintained, or repaired by USG and/or any entity, including a USG contracting unit, for which USG has legal responsibility or (ii) meaningful and credible exposure, which occurred prior to January 2, 1968, to asbestos or asbestos-containing products supplied, specified, manufactured, installed, maintained, or repaired by A.P. Green and/or any entity, including an A.P. Green contracting unit, for which A.P. Green has legal responsibility ("USG/A.P. Green Exposure"). That

¹² If a claimant's alleged exposure to A.P. Green asbestos or asbestos-containing products occurred only on or after January 2, 1968, but the claimant believes the claim qualifies as a PI Trust Claim because legal responsibility for the exposure arose prior to that date, the claimant

meaningful and credible exposure evidence may be established by an affidavit or sworn statement of the claimant, by an affidavit or sworn statement of a co-worker or the affidavit or sworn statement of a family member in the case of a deceased claimant (providing the PI Trust finds such evidence reasonably reliable), by invoices, employment, construction or similar records, or by other credible evidence. The specific exposure information required by the PI Trust to process a claim under either Expedited or Individual Review shall be set forth on the proof of claim form to be used by the PI Trust. The PI Trust can also require submission of other or additional evidence of exposure when it deems such to be necessary.

Evidence submitted to establish proof of USG/A.P. Green Exposure is for the sole benefit of the PI Trust, not third parties or defendants in the tort system. The PI Trust has no need for, and therefore claimants are not required to furnish the PI Trust with evidence of, exposure to specific asbestos products other than those for which USG or A.P. Green has legal responsibility, except to the extent such evidence is required elsewhere in this TDP. Similarly, failure to identify USG or A.P. Green products in the claimant's underlying tort action, or to other bankruptcy trusts, does not preclude the claimant from recovering from the PI Trust, provided the claimant otherwise satisfies the medical and exposure requirements of this TDP.

5.8 Claims Audit Program. The PI Trust, with the consent of the TAC and the Futures Representative, may develop methods for auditing the reliability of medical evidence, including additional reading of X-rays, CT scans and verification of pulmonary function tests, as well as the reliability of evidence of exposure to asbestos, including exposure to asbestoscontaining products manufactured or distributed by USG prior to December 31, 1982 or A.P.

must submit his or her PI Trust Claim for Individual Review on this issue pursuant to Section 5.3(b) above.

Green prior to January 2, 1968. In the event that the PI Trust reasonably determines that any individual or entity has engaged in a pattern or practice of providing unreliable medical evidence to the PI Trust, it may decline to accept additional evidence from such provider in the future.

Further, in the event that an audit reveals that fraudulent information has been provided to the PI Trust, the PI Trust may penalize any claimant or claimant's attorney by disallowing the PI Trust Claim or by other means including, but not limited to, requiring the source of the fraudulent information to pay the costs associated with the audit and any future audit or audits, reordering the priority of payment of all affected claimants' PI Trust Claims, raising the level of scrutiny of additional information submitted from the same source or sources, refusing to accept additional evidence from the same source or sources, seeking the prosecution of the claimant or claimant's attorney for presenting a fraudulent claim in violation of 18 U.S.C. § 152, and seeking sanctions from the Bankruptcy Court.

Section 2.1 that a claimant may not assert more than one (1) PI Trust Claim hereunder, the holder of a PI Trust Claim involving a non-malignant asbestos-related disease (Disease Levels I–IV) may assert a new PI Trust Claim against the PI Trust for a malignant disease (Disease Levels V–VIII) that is subsequently diagnosed. Any additional payments to which such claimant may be entitled with respect to such malignant asbestos-related disease shall not be reduced by the amount paid for the non-malignant asbestos-related disease, provided that the malignant disease had not been diagnosed by the time the claimant was paid with respect to the original claim involving the non-malignant disease.

5.10 Arbitration.

TAC and the Futures Representative, shall institute binding and non-binding arbitration procedures in accordance with the Alternative Dispute Resolution ("ADR") Procedures included in Attachment A hereto for resolving disputes concerning whether a Pre-Petition settlement agreement with USG is binding and judicially enforceable in the absence of a Final Order of the Bankruptcy Court determining the issue, whether the PI Trust's outright rejection or denial of a claim was proper, or whether the claimant's medical condition or exposure history meets the requirements of this TDP for purposes of categorizing a claim involving Disease Levels I–VIII. Binding and non-binding arbitration shall also be available for resolving disputes over the liquidated value of a claim involving Disease Levels IV–VIII, as well as disputes over USG's share of the unpaid portion of a Pre-Petition Liquidated PI Trust Claim described in Section 5.2 above and disputes over the validity of an Indirect PI Trust Claim.

In all arbitrations, the arbitrator shall consider the same medical and exposure evidentiary requirements that are set forth in Section 5.7 above. In the case of an arbitration involving the liquidated value of a claim involving Disease Levels IV—VIII, the arbitrator shall consider the same valuation factors that are set forth in Section 5.3(b)(2) above. In order to facilitate the Individual Review Process with respect to such claims, the PI Trust may from time to time develop a valuation model that enables the PI Trust to efficiently make initial liquidated value offers on these claims in the Individual Review setting. In an arbitration involving any such claim, the PI Trust shall neither offer into evidence or describe any such model nor assert that any information generated by the model has any evidentiary relevance or should be used by the arbitrator in determining the presumed correct liquidated value in the arbitration. The

underlying data that was used to create the model may be relevant and may be made available to the arbitrator but only if provided to the claimant or his/her counsel ten (10) days prior to the arbitration proceeding. With respect to all claims eligible for arbitration, the claimant, but not the PI Trust, may elect either non-binding or binding arbitration. The ADR Procedures set forth in Attachment A hereto may be modified by the PI Trust with the consent of the TAC and the Futures Representative.

5.10(b) Claims Eligible for Arbitration. In order to be eligible for arbitration, the claimant must first complete the Individual Review Process with respect to the disputed issue as well as either the Pro Bono Evaluation or the Mediation processes set forth in the ADR Procedures. Individual Review shall be treated as completed for these purposes when the claim has been individually reviewed by the PI Trust, the PI Trust has made an offer on the claim, the claimant has rejected the liquidated value resulting from the Individual Review, and the claimant has notified the PI Trust of the rejection in writing. Individual Review shall also be treated as completed if the PI Trust has rejected the claim.

5.10(c) Limitations on and Payment of Arbitration Awards. In the case of a non-Extraordinary claim involving Disease Levels I–III, the arbitrator shall not return an award in excess of the Scheduled Value for such claim. In the case of a non-Extraordinary Claim involving Disease Levels IV–VIII, the arbitrator shall not return an award in excess of the Maximum Value for the appropriate Disease Level as set forth in Section 5.3(a)(3) above, and for an Extraordinary Claim involving one of those Disease Levels, the arbitrator shall not return an award greater than the maximum value for such a claim as set forth in Section 5.4(a) above. A claimant who submits to arbitration and who accepts the arbitral award shall receive payments in the same manner as one who accepts the PI Trust's original valuation of the claim.

5.11 Litigation. Claimants who elect non-binding arbitration and then reject their arbitral awards retain the right to institute a lawsuit in the tort system against the PI Trust pursuant to Section 7.6 below. However, a claimant shall be eligible for payment of a judgment for monetary damages obtained in the tort system from the PI Trust's available cash only as provided in Section 7.7 below.

SECTION VI

Claims Materials

- 6.1 Claims Materials. The PI Trust shall prepare suitable and efficient claims materials ("Claims Materials") for all PI Trust Claims, and shall provide such Claims Materials upon a written request for such materials to the PI Trust. The proof of claim form to be submitted to the PI Trust shall require the claimant to assert the highest Disease Level for which the claim qualifies at the time of filing. The proof of claim form shall also include a certification by the claimant or his or her attorney sufficient to meet the requirements of Rule 11(b) of the Federal Rules of Civil Procedure. In developing its claim filing procedures, the PI Trust shall make every effort to provide claimants with the opportunity to utilize currently available technology at their discretion, including filing claims and supporting documentation over the Internet and electronically by disk or CD-rom. The proof of claim form to be used by the Trust shall be developed by the PI Trust and submitted to the TAC and the Futures Representatives for approval; it may be changed by the PI Trust with the consent of the TAC and the Futures Representative.
- 6.2 Content of Claims Materials. The Claims Materials shall include a copy of this TDP, such instructions as the Trustees shall approve, and a detailed proof of claim form. If feasible, the forms used by the PI Trust to obtain claims information shall be the same or

substantially similar to those used by other asbestos claims resolution organizations. If requested by the claimant, the PI Trust shall accept information provided electronically. The claimant may, but shall not be required to, provide the PI Trust with evidence of recovery from other defendants and claims resolution organizations.

- 6.3 Withdrawal or Deferral of Claims. A claimant can withdraw a PI Trust Claim at any time upon written notice to the PI Trust and file another claim subsequently without affecting the status of the claim for statute of limitations purposes, but any such claim filed after withdrawal shall be given a place in the FIFO Processing Queue based on the date of such subsequent filing. A claimant can also request that the processing of his or her PI Trust Claim by the PI Trust be deferred for a period not to exceed three (3) years without affecting the status of the claim for statute of limitation purposes, in which case the claimant shall also retain his or her original place in the FIFO Processing Queue. During the period of such deferral, a sequencing adjustment on such claimant's PI Trust Claim as provided in Section 7.5 hereunder shall not accrue and payment thereof shall be deemed waived by the claimant. Except for PI Trust Claims held by representatives of deceased or incompetent claimants for which court or probate approval of the PI Trust's offer is required, or a PI Trust Claim for which deferral status has been granted, a claim shall be deemed to have been withdrawn if the claimant neither accepts, rejects, nor initiates arbitration within six (6) months of the PI Trust's written offer of payment or rejection of the claim. Upon written request and good cause, the PI Trust may extend the withdrawal or deferral period for an additional six (6) months.
- 6.4 Filing Requirements and Fees. The Trustees shall have the discretion to determine, with the consent of the TAC and the Futures Representative, (a) whether a claimant must have previously filed an asbestos-related personal injury claim in the tort system to be

eligible to file the claim with the PI Trust and (b) whether a filing fee should be required for any PI Trust Claims.

6.5 Confidentiality of Claimants' Submissions. All submissions to the PI Trust by a holder of a PI Trust Claim or a proof of claim form and materials related thereto shall be treated as made in the course of settlement discussions between the holder and the PI Trust, and intended by the parties to be confidential and to be protected by all applicable state and federal privileges, including but not limited to those directly applicable to settlement discussions. The PI Trust will preserve the confidentiality of such claimant submissions, and shall disclose the contents thereof only, with the permission of the holder, to another trust established for the benefit of asbestos personal injury claimants pursuant to section 524(g) and/or section 105 of the Bankruptcy Code or other applicable law, to such other persons as authorized by the holder, or in response to a valid subpoena of such materials issued by the Bankruptcy Court. Furthermore, the PI Trust shall provide counsel for the holder a copy of any such subpoena immediately upon being served. The PI Trust shall on its own initiative or upon request of the claimant in question take all necessary and appropriate steps to preserve said privileges before the Bankruptcy Court and before those courts having appellate jurisdiction related thereto. Notwithstanding anything in the foregoing to the contrary, with the consent of the TAC and the Future Claimants' Representative, the PI Trust may, in specific limited instances, disclose information, documents, or other materials reasonably necessary in the PI Trust's judgment to preserve. litigate, resolve, or settle coverage, or to comply with an applicable obligation under an insurance policy or settlement agreement within the USG Asbestos Personal Injury Liability Insurance Assets; provided, however, that the PI Trust shall take any and all steps reasonably feasible in its judgment to preserve the further confidentiality of such information, documents

and materials, and prior to the disclosure of such information, documents or materials to a third party, the PI Trust shall receive from such third party a written agreement of confidentiality that (a) ensures that the information, documents and materials provided by the PI Trust shall be used solely by the receiving party for the purpose stated in the agreement and (b) prohibits any other use or further dissemination of the information, documents and materials by the third party.

SECTION VII

General Guidelines for Liquidating and Paying Claims

- 7.1 Showing Required. To establish a valid PI Trust Claim, a claimant must meet the requirements set forth in this TDP. The PI Trust may require the submission of X-rays, CT scans, laboratory tests, medical examinations or reviews, other medical evidence, or any other evidence to support or verify the PI Trust Claim, and may further require that medical evidence submitted comply with recognized medical standards regarding equipment, testing methods, and procedures to assure that such evidence is reliable.
- 7.2 Costs Considered. Notwithstanding any provisions of this TDP to the contrary, the Trustees shall always give appropriate consideration to the cost of investigating and uncovering invalid PI Trust Claims so that the payment of valid PI Trust Claims is not further impaired by such processes with respect to issues related to the validity of the medical evidence supporting a PI Trust Claim. The Trustees shall also have the latitude to make judgments regarding the amount of transaction costs to be expended by the PI Trust so that valid PI Trust Claims are not unduly further impaired by the costs of additional investigation. Nothing herein shall prevent the Trustees, in appropriate circumstances, from contesting the validity of any claim against the PI Trust whatever the costs, or to decline to accept medical evidence from

sources that the Trustees have determined to be unreliable pursuant to the Claims Audit Program described in Section 5.8 above.

Discretion to Vary the Order and Amounts of Payments in Event of Limited Liquidity. Consistent with the provisions hereof and subject to the FIFO Processing and Liquidation Queues, the Maximum Annual Payment, the Maximum Available Payment and the Claims Payment Ratio requirements set forth above, the Trustees shall proceed as quickly as possible to liquidate valid PI Trust Claims, and shall make payments to holders of such claims in accordance with this TDP promptly as funds become available and as claims are liquidated, while maintaining sufficient resources to pay future valid claims in substantially the same manner.

Because the PI Trust's income over time remains uncertain, and decisions about payments must be based on estimates that cannot be done precisely, they may have to be revised in light of experiences over time, and there can be no guarantee of any specific level of payment to claimants. However, the Trustees shall use their best efforts to treat similar claims in substantially the same manner, consistent with their duties as Trustees, the purposes of the PI Trust, the established allocation of funds to claims in Categories A and B, and the practical limitations imposed by the inability to predict the future with precision.

In the event that the PI Trust faces temporary periods of limited liquidity, the Trustees may, with the consent of the TAC and the Futures Representative, (a) suspend the normal order of payment, (b) temporarily limit or suspend payments altogether, (c) offer a Reduced Payment Option as described in Section 2.5 above and/or (d) commence making payments on an installment basis.

7.4 Punitive Damages. Except as provided below for claims asserted under the Alabama Wrongful Death Statute, in determining the value of any liquidated or unliquidated PI Trust Claim, punitive or exemplary damages, *i.e.*, damages other than compensatory damages, shall not be considered or allowed, notwithstanding their availability in the tort system.

Similarly, no punitive or exemplary damages shall be payable with respect to any claim litigated against the PI Trust in the tort system pursuant to Sections 5.11 above and 7.6 below. The only damages that may be awarded pursuant to this TDP to Alabama Claimants who are deceased and whose personal representatives pursue their claims only under the Alabama Wrongful Death Statute shall be compensatory damages determined pursuant to the statutory and common law of the Commonwealth of Pennsylvania, without regard to its choice of law principles. The choice of law provision in Section 7.4 herein applicable to any claim with respect to which, but for this choice of law provision, the applicable law of the Claimant's Jurisdiction pursuant to Section 5.3(b)(2) is determined to be the Alabama Wrongful Death Statute, shall only govern the rights between the PI Trust and the claimant including, but not limited to, suits in the tort system pursuant to Section 7.6, and to the extent the PI Trust seeks recovery from any entity that provided insurance to USG, the Alabama Wrongful Death Statute shall govern.

7.5 Sequencing Adjustment.

7.5(a) In General. Except for any PI Trust Claim involving Other Asbestos

Disease (Disease Level I – Cash Discount Payment) and subject to the limitations set forth

below, a sequencing adjustment shall be paid on all PI Trust Claims with respect to which the

claimant has had to wait a year or more for payment, provided, however, that no claimant shall

receive a sequencing adjustment for a period in excess of seven (7) years. The sequencing

adjustment factor for each year shall be the coupon issue yield equivalent (as determined by the Secretary of the Treasury) of the average accepted auction price for the first auction of five-year Treasury Notes occurring in such year.

7.5(b) Unliquidated PI Trust Claims. A sequencing adjustment shall be payable on the Scheduled Value of any unliquidated PI Trust Claim that meets the requirements of Disease Levels II–V, VII and VIII, whether the claim is liquidated under Expedited Review, Individual Review, or by arbitration. No sequencing adjustment shall be paid on any claim involving Disease Level I or on any claim liquidated in the tort system pursuant to Section 5.11 above and Section 7.6 below. The sequencing adjustment on an unliquidated PI Trust Claim that meets the requirements of Disease Level VI shall be based on the Average Value of such a claim. Sequencing adjustments on all such unliquidated claims shall be measured from the date of payment back to the earliest of the date that is one year after the date on which (a) the claim was filed against USG or A.P. Green prior to the Petition Date; (b) the claim was filed against another defendant in the tort system on or after the Petition Date but before the Effective Date; (c) the claim was filed with the Bankruptcy Court during the pendency of the Chapter 11 proceeding if the claim was filed prior to January 30, 2006; or (d) the claim was filed with the PI Trust after the Effective Date.

7.5(c) Liquidated Pre-Petition Claims. A sequencing adjustment shall also be payable on the liquidated value of all Pre-Petition Liquidated Claims described in Section 5.2(a) above. In the case of Pre-Petition Liquidated Claims liquidated by verdict or judgment, the sequencing adjustment shall be measured from the date of payment back to the date that is one (1) year after the date that the verdict or judgment was entered. In the case of Pre-Petition Liquidated Claims liquidated by a binding, judicially enforceable settlement, the sequencing

adjustment shall be measured from the date of payment back to the date that is one (1) year after the Petition Date.

- Trust's determination regarding the Disease Level of the claim, the claimant's exposure history or the liquidated value of the claim, and if the holder has first submitted the claim to non-binding arbitration as provided in Section 5.10 above, the holder may file a lawsuit in the Claimant's Jurisdiction as defined in Section 5.3(b)(2) above. Any such lawsuit must be filed by the claimant in her or her own right and name and not as a member or representative of a class, and no such lawsuit may be consolidated with any other lawsuit. All defenses (including, with respect to the PI Trust, all defenses which could have been asserted by USG or A.P. Green) shall be available to both sides at trial; however, the PI Trust may waive any defense and/or concede any issue of fact or law. If the claimant was alive at the time the initial pre-petition complaint was filed or on the date the proof of claim form was filed with the PI Trust, the case shall be treated as a personal injury case with all personal injury damages to be considered even if the claimant has died during the pendency of the claim.
- 7.7 Payment of Judgments for Money Damages. If and when a claimant obtains a judgment in the tort system, the claim shall be placed in the FIFO Payment Queue based on the date on which the judgment became final. Thereafter, the claimant shall receive from the PI Trust an initial payment (subject to the applicable Payment Percentage, the Maximum Available Payment, the Claims Payment Ratio, and the sequencing adjustment provisions set forth above) of an amount equal to the greater of (i) the PI Trust's last offer to the claimant or (ii) the award that the claimant declined in non-binding arbitration. The claimant shall receive the balance of the judgment, if any, in five (5) equal installments in years six (6) through ten (10) following the year

of the initial payment (also subject to the applicable Payment Percentage, the Maximum Available Payment, the Claims Payment Ratio, and the sequencing adjustment provisions above in effect on the date of the payment of the subject installment).

In the case of non-Extraordinary claims involving Disease Levels I, II and III, the total amounts paid with respect to such claims shall not exceed the relevant Scheduled Value for such Disease Levels as set forth in Section 5.3(b)(3) above. In the case of claims involving a non-malignant asbestos-related disease that does not attain classification under Disease Levels I, II or III, the amount payable shall not exceed the Scheduled Value for the Disease Level most comparable to the disease proven. In the case of non-Extraordinary claims involving severe asbestosis and malignancies (Disease Levels IV–VIII), the total amounts paid with respect to such claims shall not exceed the Maximum Values for such Disease Levels set forth in Section 5.3(b)(3). In the case of Extraordinary Claims, the total amounts paid with respect to such claims shall not exceed the Maximum Value for such claims set forth in Section 5.4(a) above. Under no circumstances shall either a sequencing adjustment be paid pursuant to Section 7.5 or interest be paid under any statute on any judgments obtained in the tort system.

7.8 Releases. The Trustees shall have the discretion to determine the form and substance of the releases to be provided to the PI Trust in order to maximize recovery for claimants against other tortfeasors without increasing the risk or amount of claims for indemnification or contribution from the PI Trust, which releases shall be reasonably acceptable in form and substance to the Reorganized Debtors. As a condition to making any payment to a claimant, the PI Trust shall obtain a general, partial, or limited release reasonably acceptable in form and substance to the Reorganized Debtors as appropriate in accordance with the applicable state or other law. If allowed by state law, the endorsing of a check or draft for payment by or on

behalf of a claimant may, in the discretion of the PI Trust, constitute such a release. The form of release included as Attachment B hereto is acceptable to the Reorganized Debtors. In no event shall the form of release extend to any parties other than the PI Trust, the Debtors, the Debtors' Estates and the Reorganized Debtors.

- 7.9 Third-Party Services. Nothing in this TDP shall preclude the PI Trust from contracting with another asbestos claims resolution organization to provide services to the PI Trust so long as decisions about the categorization and liquidated value of PI Trust Claims are based on the relevant provisions of this TDP, including the Disease Levels, Scheduled Values, Average Values, Maximum Values, and Medical/Exposure Criteria set forth above.
- 7.10 PI Trust Disclosure of Information. Periodically, but not less often than once a year, the PI Trust shall make available to claimants and other interested parties, the number of claims by Disease Levels that have been resolved both by the Individual Review Process and by arbitration as well as by litigation in the tort system indicating the amounts of the awards and the averages of the awards by jurisdiction.

SECTION VIII

Miscellaneous

8.1 Amendments. Except as otherwise provided herein, the Trustees may amend, modify, delete, or add to any provisions of this TDP (including, without limitation, amendments to conform this TDP to advances in scientific or medical knowledge or other changes in circumstances), provided they first obtain the consent of the TAC and the Futures Representative pursuant to the Consent Process set forth in Sections 5.7(b) and 6.6(b) of the PI Trust Agreement, except that the right to amend the Claims Payment Ratio is governed by the restrictions in Section 2.5 above, and the right to adjust the Payment Percentage is governed by

Section 4.2 above. Nothing herein is intended to preclude the TAC or the Futures Representatives from proposing to the Trustees, in writing, amendments to this TDP. Any amendment proposed by the TAC or the Futures Representatives shall remain subject to Section 7.3 of the PI Trust Agreement. Notwithstanding anything contained in this TDP to the contrary, neither this TDP nor any document annexed to this TDP shall be modified or amended in any way that could: (a) alter this Section 8.1; (b) jeopardize, impair, or modify the applicability of section 524(g) of the Bankruptcy Code to the Plan and the Confirmation Order, the efficacy or enforceability of the injunction entered thereunder, or the PI Trust's qualified settlement fund status under the QSF Regulations; or (c) otherwise adversely affect the rights of the Reorganized Debtors under Sections 5.3(a)(2), 7.8 or 8.1 of this TDP.

- 8.2 Severability. Should any provision contained in this TDP be determined to be unenforceable, such determination shall in no way limit or affect the enforceability and operative effect of any and all other provisions of this TDP. Should any provision contained in this TDP be determined to be inconsistent with or contrary to USG's obligations to any insurance company providing insurance coverage to USG in respect of claims for personal injury based on exposure to an asbestos-containing product or to conduct for which USG has legal responsibility, the PI Trust with the consent of the TAC and the Futures Representative may amend this TDP and/or the PI Trust Agreement to make the provisions of either or both documents consistent with the duties and obligations of USG to said insurance company.
- 8.3 Governing Law. Except for purposes of determining the liquidated value of any PI Trust Claim, administration of this TDP shall be governed by, and construed in accordance with, the laws of the State of Delaware. The law governing the liquidation of PI Trust Claims in

the case of Individual Review, arbitration or litigation in the tort system shall be the law of the Claimant's Jurisdiction as described in Section 5.3(b)(2) above.

EXHIBIT L

IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

In re:) Chapter 11	
OWENS CORNING, et al.,) Case No. 00-3837 (JKF)
Debtors.) Jointly Administered	

ANNUAL REPORT AND ACCOUNT OF THE OWENS CORNING/FIBREBOARD ASBESTOS PERSONAL INJURY TRUST FOR THE FISCAL YEAR ENDING DECEMBER 31, 2009

Harry Huge, D. LeAnne Jackson, and Dean M. Trafelet, the Trustees of the Owens Corning/Fibreboard Asbestos Personal Injury Trust (the "Trust"), created pursuant to the Sixth Amended Joint Plan of Reorganization for Owens Corning and Its Affiliated Debtors and Debtors-in-Possession, as modified (the "Plan") submit this Annual Report, Financial Statements, and Claims Summary for the fiscal year ended December 31, 2009.

I. INTRODUCTION

The purpose of this Annual Report is to comply with the reporting requirements of the Owens Corning/Fibreboard Asbestos Personal Injury Trust Agreement (the "Trust Agreement") and to report to the Court on the actions taken by the Trustees on behalf of the Trust during the period from January 1, 2009 to December 31, 2009 (the "Reporting Period").

Section 2.2(c) of the Trust Agreement provides:

The Trustees shall timely account to the Bankruptcy Court as follows:

(i) The Trustees shall cause to be prepared and filed with the Bankruptcy Court, as soon as available, and in any event within one hundred and twenty (120) days following the end of each fiscal year, an annual report (the "Annual Report") containing financial statements of the PI Trust (including, without limitation, a balance sheet of the PI Trust as of the end of such fiscal year and

a statement of operations for such fiscal year) audited by a firm of independent certified public accountants selected by the Trustees and accompanied by an opinion of such firm as to the fairness of the financial statements' presentation of the cash and investments available for the payment of claims and as to the conformity of the financial statements with generally accepted accounting principles. The Trustees shall provide a copy of such report to the TAC, the Future Claimants' Representative, and Reorganized OC when such reports are filed with the Bankruptcy Court.

- (ii) Simultaneously with the filing of the Annual Report, the Trustees shall cause to be prepared and filed with the Bankruptcy Court a report containing a summary regarding the number and type of claims disposed of during the period covered by the financial statements. The Trustees shall provide a copy of such report to the TAC, the Future Claimants' Representative, and Reorganized OC when such report is filed.
- (iii) All materials required to be filed with the Bankruptcy Court by this Section 2.2(c) shall be available for inspection by the public in accordance with procedures established by the Bankruptcy Court and shall be filed with the Office of the United States Trustee for the District of Delaware.

The Trust Agreement further provides for the inclusion of a description of the amounts paid to the Trustees, TAC, and FCR in the accounts filed with the Bankruptcy Court. See Sections 4.5(c), 5.6, and 6.5.

II. BACKGROUND

On October 5, 2000, Owens Corning and certain of its subsidiaries, debtors and debtors-in-possession in the above-captioned cases (collectively, the "Debtors"), filed voluntary petitions for relief under Chapter 11 of Title 11 of the United States Code. On September 26, 2006, the Court entered an order in these cases confirming the Plan (the "Plan"). On September 28, 2006, the United States District Court for the District of Delaware affirmed this Court's confirmation order. On October 31, 2006 (the "Effective Date"), the Plan became effective and the Trustees began to operate the Trust.

¹ Trust Advisory Committee

On or after the Effective Date, the OC and FB Sub-Accounts of the Trust were funded as provided for in Sections 3.3(f)(iii) and 3.4(d)(iii) of the Plan respectively. The purpose of the Trust is to assume the liabilities of OC and Fibreboard, their predecessors and successors in interest, for all Asbestos Personal Injury Claims, as defined in the Plan, and to use the assets and income of the Trust to pay both present and future asbestos claimants in accordance with the Trust Agreement and the Trust Distribution Procedures in such a way that holders of Asbestos Personal Injury Claims are treated fairly, equitably, and reasonably in light of the limited assets available to satisfy such claims. See Trust Agreement, Section 1.2

III. TRUST ADMINISTRATION

A. Trustees

Harry Huge, D. LeAnne Jackson, and Dean M. Trafelet served as the Trustees of the Trust during the Reporting Period. Dean M. Trafelet served as the Managing Trustee during the Reporting Period.

1. Meetings

During the Reporting Period, the Trustees held formal meetings in accordance with the requirements of the Owens Corning/Fibreboard Asbestos Personal Injury Trust Agreement. These meetings were designated as "regular meetings" under the Trust's By-laws. Each meeting was in person and was attended by representatives of the Trust Advisory Committee and the Future Claimants' Representative. In addition to these formal meetings, the Trustees held regularly scheduled weekly teleconferences, met individually with Trust advisors, held executive session and special purpose meetings, including meetings to address Trust policies and claims processing matters and meetings with the Trust's investment advisor, investment managers and investment bankers as well as with the management of Owens Corning, and devoted

considerable time to Trust matters outside of scheduled meetings. Activities included exercising oversight over the investment and liquidation of Trust assets to pay claims and the Trust's processes to receive, process, and pay claims pursuant to the Trust Distribution Procedures, communicating with claimants regarding the processing of claims, continual monitoring of the claims and investment processes and supervising various insurance related matters.

2. Compensation

The compensation and expenses paid to the Trustees during the Reporting Period is set forth in the Trust's financial statements attached hereto as Exhibit "A." The inclusion of this information in the Trust's financial statements satisfies the requirements of Section 4.5 of the Trust Agreement. All distributions related to Trustee compensation and expense reimbursements were made in accordance with the Trust Agreement guidelines and applicable By-laws.

B. Trust Officers

Analysis Research Planning Consulting ("ARPC") served as the Trust's Executive Director during the Reporting Period.

C. Trust Advisory Committee

Matthew P. Bergman, Russell W. Budd, John D. Cooney, James L. Ferraro, Steven Kazan, Joseph F. Rice, Armand J. Volta, Jr., and Perry Weitz served as Members of the Trust Advisory Committee (the "TAC Members") during the Reporting Period. Theodore Goldberg served as a TAC Member during the Reporting Period until appointing his successor, Mark C. Meyer who served as a TAC Member through the remainder of the Reporting Period. The compensation and expenses paid to the TAC Members during the Reporting Period is set forth in the Trust's financial statements. The inclusion of this information in the Trust's financial statements satisfies the requirements of Section 5.6 of the Trust Agreement. All distributions

related to TAC Member compensation and expense reimbursements were made in accordance with the Trust Agreement guidelines and applicable By-laws.

D. Future Claimants' Representative

Michael J. Crames served as the Future Claimants' Representative (the "FCR") during the Reporting Period. The compensation paid to the FCR during the Reporting Period is set forth in the Trust's financial statements. The inclusion of this information in the Trust's financial statements satisfies the requirements of Section 6.5 of the Trust Agreement. All distributions related to FCR compensation and expense reimbursements were made in accordance with the Trust Agreement guidelines and applicable By-laws.

E. Claims Processing

The Delaware Claims Processing Facility, LLC continues to process the Owens Corning/Fibreboard Asbestos Personal Injury Trust Claims. The terms of the Trust's ownership interest in the Delaware Claims Processing Facility are detailed in the Audited Financial Statements.² In addition, on March 1, 2007 the Trust engaged MFR Claims Processing, Inc, a Pennsylvania corporation to process Pre-Petition Liquidated Claims.

F. Investment Management

During the Reporting Period, the Trust continued to engage Cambridge Associates, LLC of Boston, Massachusetts as the Trust's investment advisor. Cambridge Associates represents other asbestos settlement trusts and has experience advising such trusts regarding investment management. Cambridge Associates advises the Trust on, among other things, asset allocation, monetizing assets to provide funds to meet liquidity needs for the payment of claims and Trust expenses, and the selection and oversight of individual investment managers for the investable portions of the Trust's portfolio.

² Audited Financial Statements at Note 3.

The Trust Agreement contains certain general investment guidelines that primarily address credit quality and asset concentration. Additionally, the Trust's investments are invested in fixed income and equity investments in accordance with a targeted allocation policy recommended by Cambridge Associates after consideration of the currently forecasted timing of the liquidity needs of the Trust to pay claims and expenses, diversification, the Trust's concentrated investment in Owens Corning common stock, the Trust's status as a federal tax paying "qualified settlement fund," and such other factors as Cambridge Associates considers relevant to forming its recommendation. Due to fluctuations in market prices of the Trust's investments, particularly its equity investments, and the regular liquidation of fixed income investments to meet the cash needs of the Trust's two sub-accounts to pay claims and operating expenses, actual allocations may vary from target allocations at any single point in time.

The Trust engages multiple investment managers for each of the asset classes in which it invests in order to achieve additional diversification. The investment managers are selected by the Trust with the assistance, and upon the recommendation, of Cambridge Associates. At December 31, 2009, the following investment managers were retained by the Trust:

Equity

Northern Trust Global Investments Grisanti Brown & Partners LLC Sanderson Asset Management, Inc. Gryphon International Investment Corporation

Tax Exempt Municipal Bonds

Eaton Vance Tax Advantaged Bond Strategy Schroders Investment Management T. Rowe Price Brown Brothers Harriman Goldman Sachs

For investment and tax purposes each of the Trust's two sub-accounts, the Owens Corning sub-account and the Fibreboard sub-account, are treated as separate "qualified settlement funds," although they have similar investment target allocations and employ the same mix of asset managers.

During the Reporting Period, the Trustees regularly met with Cambridge Associates to review investment performance, economic conditions and investment strategy. In addition, during March 2009, as part of an annual review process, the Trustees and Cambridge Associates met with each of the Trust's investment managers to review changes to the manager's organization, the manager's investment strategy and investment performance, and the manager's market outlook and expectations for the upcoming year.

The Bank of New York Mellon serves as custodian for the Trust's separately-managed investment accounts. The Trust's actively managed non-U.S. equity portfolios are invested through commingled vehicles that have a separate custodian (selected by the investment manager) for each of the respective vehicles. The Trust utilizes custodian accounts, wire transfers and zero-balance banking accounts to reduce its exposure to systemic banking system risk.

In February 2009 the Trust retained Rothschild Inc. as its financial advisor with respect to its ownership of Owens Corning common stock. Prior to this date, the Trust had engaged Blackstone Advisory Services, LLP to advise it with respect to the Owens Corning common stock. During the Reporting Period, the Trustees met regularly with Rothschild Inc. to review matters relating to its equity interest in Owens Corning. The Trustees and their advisors also met with the management of Owens Corning to discuss matters of interest to shareholders.

G. Insurance Coverage Litigation

Upon the Effective Date, pursuant to the Plan, the Trust received the right to pursue certain unsettled insurance coverage and the rights to certain proceeds paid pursuant to Asbestos PI Insurance Settlement Agreements. The Trust retained the firm of Anderson Kill & Olick, P.C. as Special Insurance Counsel to the Trust. Prior to the Effective Date of the Plan of

Reorganization, the Debtors and other Plan Proponents reached settlement agreements with many of the Debtor's historical insurance companies that required the settling insurance companies to make cash payments to or for the benefit of the Trust. Certain of those agreements required the settling insurance companies to make payments into escrow accounts during the pendency of the bankruptcy case, and those escrow accounts were released to the Trust in the first quarter of 2006. Certain other settlement agreements required the insurance companies to pay settlement proceeds to the Trust after the Plan's Effective Date in 2006, and the Trust received from insurance companies substantial payments in 2007, 2008 and 2009. In addition, some of these settlement agreements require the settling insurance companies to make payments to the Trust on agreed-to payment schedules that extend beyond 2009. As a result, the Trust anticipates receiving additional payments through 2015 under these settlement agreements. Although the Plan Proponents settled with many of the Debtors' insurance companies prior to the Plan's Effective Date, they did not settle with all of the Debtors' insurance companies. Accordingly, in 2007, the Trust initiated settlement discussions with some unsettled solvent and insolvent domestic insurance companies (and state insurance guaranty associations) and continued in 2009 to participate in mediation and arbitration pursuant to the Wellington Agreement with others. The Trust anticipates that its efforts to secure coverage from unsettled insurance companies will continue at least through 2010.

H. General Counsel

Campbell & Levine, LLC of Pittsburgh, Pennsylvania and Wilmington, Delaware continues to serve as general counsel to the Trust. Kaplan, Strangis & Kaplan, P.A. of Minneapolis, Minnesota continues to serve as Special Corporate Counsel and Anderson Kill & Olick, P.C. of New York, New York continues to serve as Special Insurance Counsel to the Trust

IV. ANNUAL REPORT AND ACCOUNT

A. Summary of Claims

As of December 31, 2009, the Trust had paid a total of 47,206 Owens Corning Pre-Petition Liquidated PI Trust Claims. Of the claims paid, there were 4,883 malignancy claims and 42,323 non-malignancy claims. After application of the payment percentage and applicable sequencing adjustment, the Trust had paid approximately \$345.28 million to asbestos victims in settlement of their Owens Corning Pre-Petition Liquidated PI Trust Claims. The malignant to nonmalignant ratio of the Owens Corning Pre-Petition Liquidated PI Trust Claims paid in number was 1/9 and the malignant to nonmalignant ratio of the Owens Corning Pre-Petition Liquidated Trust Claims in dollars paid was 1.3/1.

As of December 31, 2009, the Trust had received 324,270 Owens Corning Unliquidated PI Trust Claims and paid a total of 113,340 claims. Of the claims paid, there were 20,724 malignancy claims and 92,616 non-malignancy claims. After application of the payment percentage and applicable sequencing adjustment, the Trust had paid approximately \$1,400.13 million to asbestos victims in settlement of their Owens Corning Unliquidated PI Trust Claims. The malignant to nonmalignant ratio of the Owens Corning Unliquidated PI Trust Claims paid in number was 1/5 and the malignant to nonmalignant ratio of the Owens Corning Unliquidated Trust Claims in dollars paid was 1.6/1.

During the reporting period, the Trust paid a total of 8,887 Owens Corning Pre-Petition Liquidated PI Trust Claims. Of the claims paid, there were 720 malignancy claims and 8,167 non-malignancy claims. After application of the payment percentage and applicable sequencing adjustment, the Trust paid approximately \$15.84 million to asbestos victims in settlement of their Owens Corning Pre-Petition Liquidated PI Trust Claims. The malignant to nonmalignant ratio of

the Owens Corning Pre-Petition Liquidated PI Trust Claims paid in number was 1/12 and the malignant to nonmalignant ratio of the Owens Corning Pre-Petition Liquidated Trust Claims in dollars paid was 4/5.

During the reporting period, the Trust received 76,063 Owens Corning Unliquidated PI Trust Claims and paid a total of 44,294 claims. Of the claims paid, there were 13,674 malignancy claims and 30,620 non-malignancy claims. After application of the payment percentage and applicable sequencing adjustment, the Trust paid approximately \$647.67 million to asbestos victims in settlement of their Owens Corning Unliquidated PI Trust Claims. The malignant to nonmalignant ratio of the Owens Corning Unliquidated PI Trust Claims paid in number was 4/9 and the malignant to nonmalignant ratio of the Owens Corning Unliquidated Trust Claims in dollars paid was 3.1/1.

As of December 31, 2009, the Trust had paid a total of 42,200 Fibreboard Pre-Petition Liquidated PI Trust Claims. Of the claims paid, there were 3,864 malignancy claims and 38,336 non-malignancy claims. After application of the payment percentage and applicable sequencing adjustment, the Trust had paid approximately \$133.58 million to asbestos victims in settlement of their Fibreboard Pre-Petition Liquidated PI Trust Claims. The malignant to nonmalignant ratio of the Fibreboard Pre-Petition Liquidated PI Trust Claims paid in number was 1/10 and the malignant to nonmalignant ratio of the Fibreboard Pre-Petition Liquidated Trust Claims in dollars paid was 1.3/1.

As of December 31, 2009, the Trust had received 300,857 Fibreboard Unliquidated PI Trust Claims and paid a total of 124,839 claims. Of the claims paid, there were 19,367 malignancy claims and 105,472 non-malignancy claims. After application of the payment percentage and applicable sequencing adjustment, the Trust had paid approximately \$586.92

million to asbestos victims in settlement of their Fibreboard Unliquidated PI Trust Claims. The malignant to nonmalignant ratio of the Fibreboard Unliquidated PI Trust Claims paid in number was 1/5 and the malignant to nonmalignant ratio of the Fibreboard Unliquidated Trust Claims in dollars paid was 2/1.

During the reporting period, the Trust paid a total of 10,821 Fibreboard Pre-Petition Liquidated PI Trust Claims. Of the claims paid, there were 597 malignancy claims and 10,224 non-malignancy claims. After application of the payment percentage and applicable sequencing adjustment, the Trust paid approximately \$9.42 million to asbestos victims in settlement of their Fibreboard Pre-Petition Liquidated PI Trust Claims. The malignant to nonmalignant ratio of the Fibreboard Pre-Petition Liquidated PI Trust Claims paid in number was 1/17 and the malignant to nonmalignant ratio of the Fibreboard Pre-Petition Liquidated Trust Claims in dollars paid was 3/4.

During the reporting period, the Trust received 81,614 Fibreboard Unliquidated PI Trust Claims and paid a total of 72,419 claims. Of the claims paid, there were 13,714 malignancy claims and 58,705 non-malignancy claims. After application of the payment percentage and applicable sequencing adjustment, the Trust paid approximately \$350.48 million to asbestos victims in settlement of their Fibreboard Unliquidated PI Trust Claims. The malignant to nonmalignant ratio of the Fibreboard Unliquidated PI Trust Claims paid in number was 1/5 and the malignant to nonmalignant ratio of the Fibreboard Unliquidated Trust Claims in dollars paid was 3/1.

A summary of the claims processing procedures and policies may be found at the Trust website at www.ocfbasbestostrust.com.

B. Reconsideration of Payment Percentage

1. Owens-Corning Sub-Account

Section 4.2 of the Trust Distribution Procedures requires that the Trustees reconsider the payment percentage for a Sub-Account if they deem such reconsideration to be appropriate in order, among other things, to assure that the Trust will be able to pay similarly situated current and future claimants against such Sub-Account as equitably as reasonably possible. After such a reconsideration, on June 4, 2009, the Trustees proposed to the Trust Advisory Committee and the Future Claimants' Representative of the Trust, whose consent to the payment percentage reduction is required by Section 2.2 (f)(iii) of the Trust's Trust Agreement, a reduction to the Payment Percentage applicable to the Owens Corning Sub-Account to 10%. In accordance with Section 4.3 of the Trust's Trust Distribution Procedures, during the pendency of the consent process, the payment percentage for the Owens Corning Sub-Account will be 10%.

As required by Section 4.2 of the Trust Distribution Procedures, the Trustees have based their decision to propose a reduction in the payment percentage for the Owens Corning Sub-Account on current estimates of the number, types and values of present and future claims against the respective Sub-Accounts, the value of assets currently available to the respective Sub-Accounts for claims payments, anticipated processing, administrative and legal expenses of the respective Sub-Accounts, and other material matters that the Trustees and their advisors considered to be reasonably likely to affect the sufficiency of the respective Sub-Account's assets to pay a comparable percentage of full value to all holders of claims against the Sub-Accounts.

2. Fibreboard Sub-Account

Section 4.2 of the Trust Distribution Procedures requires that the Trustees reconsider the payment percentage for a Sub-Account if they deem such reconsideration to be appropriate in order, among other things, to assure that the Trust will be able to pay similarly situated current and future claimants against such Sub-Account as equitably as reasonably possible. After such a reconsideration, on August 7, 2009, the Trustees proposed to the Trust Advisory Committee and the Future Claimants' Representative of the Trust, whose consent to the payment percentage reduction is required by Section 2.2 (f)(iii) of the Trust's Trust Agreement, a reduction to the Payment Percentage applicable to the Fibreboard Sub-Account to 11%. In accordance with Section 4.3 of the Trust's Trust Distribution Procedures, during the pendency of the consent process the payment percentage for the Fibreboard Sub-Account will be 11%.

As required by Section 4.2 of the Trust Distribution Procedures, the Trustees have based their decision to propose a reduction in the payment percentage for the Fibreboard Sub-Account on current estimates of the number, types and values of present and future claims against the respective Sub-Accounts, the value of assets currently available to the respective Sub-Accounts for claims payments, anticipated processing, administrative and legal expenses of the respective Sub-Accounts, and other material matters that the Trustees and their advisors considered to be reasonably likely to affect the sufficiency of the respective Sub-Account's assets to pay a comparable percentage of full value to all holders of claims against the Sub-Accounts

C. Financial Information

The Trust's audited financial statements for the Reporting Period are attached hereto as Exhibit "A." The financial statements were audited by Argy, Wiltse & Robinson, LLC.

Date: April 30, 2010

CAMPBELL & LEVINE, LLC

/s/ Kathleen Campbell Davis

Marla Rosoff Eskin (No. 2989) Kathleen Campbell Davis (No. 4229) 800 N. King Street, Suite 300 Wilmington, DE 19801 Tel. 302.426.1900 Fax 302.426.9947

-and-

Douglas A. Campbell Stanley E. Levine Philip E. Milch 1700 Grant Building Pittsburgh, PA 15219 Tel. 412.261.0310 Fax 412.261.5066

Counsel to Owens Corning/Fibreboard Asbestos Personal Injury Trust

EXHIBIT M

ELECTRONIC FILER AGREEMENT

Th	nis Electronic F	liler Agreeme	ent (the "A	greement	") is	made by	and be	tween (Celotex
Asbestos	Settlement Tru	st ("Celotex	Settlement	Trust"),	with	offices a	it 1007	North	Orange
Street,	Wilmington,	Delaware	19801-	-3023,	and	the	law	firm	of
				,		with	off	fices	at
						•	(collect	ively,	"the
Parties.")		is a partner,	member or	other prin	cipal	of the lav	v firm o	n whose	behalf
this Agree	ement is execute	ed and has the	authorizati	on of the	law f	irm to ent	er into	this Agr	eement
and shall	be designated as	the "Primary	Counsel",						

Recitals

Celotex Settlement Trust is a qualified settlement Trust, established in part to process, settle and pay asbestos personal injury claims to persons injured by exposure to asbestos containing materials manufactured or sold by Celotex Corp. or Carey Canada, Inc.; and

the Primary Counsel is a lawyer who files asbestos personal injury claims on behalf of the Primary Counsel's clients ("Claimants"); and

it is mutually beneficial to Celotex Settlement Trust and the Primary Counsel to settle asbestos personal injury claims by communicating information to each other electronically, to speed claim processing and lower transactional costs,

NOW THEREFORE, in consideration of the promises and undertakings described herein, the sufficiency of which consideration is hereby acknowledged, the undersigned Parties agree, with each other, as follows:

1. Electronic Claims Filing

- 1.1 Access to Celotex Online. Celotex Settlement Trust will provide the Primary Counsel with access to Celotex Settlement Trust's on-line claim system (Celotex Online), through which Primary Counsel may view, submit and modify asbestos injury claims submitted to Celotex Settlement Trust in electronic format. The Primary Counsel will have access to Celotex Online for the sole purpose of filing and settling asbestos claims utilizers on-line claim review. The Primary Counsel will have access to information through Celotex Online only regarding the Primary Counsel's own on-line claim review, and will be authorized to act through Celotex Online only in regard to the Primary Counsel's own on-line claim review.
- Filing Methods, Media, and Format. Celotex Settlement Trust will accept claim data from the Primary Counsel using one or more methods and electronic media that Celotex Settlement Trust will from time to time specify, in formats that Celotex Settlement Trust will from time to time specify. The methods, media, and formats which Celotex Settlement Trust will specify will be among those then in general use among businesses transferring information electronically.

Proprietary System. The Primary Counsel acknowledges that the Celotex Online concept, including all enhancements thereto and all screens and formats used in connection therewith, are the exclusive proprietary property of Celotex Settlement Trust, and the Primary Counsel shall not publish, disclose, display, provide access to or otherwise make available any software, hardware or any other products associated with Celotex Online, or any screens, formats, reports or printouts used, provided, produced from or in connection therewith, to any person or entity other than an employee or principal of the Primary Counsel, without the prior written consent of Celotex Settlement Trust, with the exception that the Primary Counsel may publish, disclose, display, provide access to or otherwise make available to a Claimant represented by the Primary Counsel any screens, reports or printouts which contain information relating solely to that Claimant's claim.

2. <u>User Identification</u>,

- 2.1 Firm Administrator. The Primary Counsel's access to Celotex Online will be managed by an individual under the Primary Counsel's employ, management or control whom the Primary Counsel identifies to serve as the Firm Administrator on the Primary Counsel's behalf. The Primary Counsel must be a licensed, practicing attorney who is partner or other principal in the law firm on whose behalf he or she executes this Agreement. The Firm Administrator, however, need not be an attorney. A Firm Administrator will be permitted to identify additional individuals under the Primary Counsel's employ, management or control as authorized to access Celotex Online on the Primary Counsel's behalf. Some Celotex Online users will be limited in the functions they can perform on the system; a Firm Administrator will have the power to designate the level of authority which each of the Primary Counsel's other Celotex Online users will have on the system. At all times the Primary Counsel and the Firm Administrator will remain responsible for oversight and supervision of the additional individuals authorized to access the system, and will at all times remain responsible for the accuracy and non-fraudulent nature of the information submitted.
- Limited Access. Celotex Settlement Trust will assign a unique Celotex Online password to the Firm Administrator. The Firm Administrator will then create accounts for users within the Primary Counsel's employ, management or control. Celotex Settlement Trust will provide access to Celotex Online only upon entry of Celotex Online password. Celotex Settlement Trust may deny access to its system following failed log-ins. Individuals with access to Celotex Online may not share their passwords with others, and the Primary Counsel will not permit any individual to use a Celotex Online password that is not assigned to that individual. Celotex Settlement Trust will limit each user's access to the system to correspond with the level of authority specified by the Primary Counsel's Firm Administrator. Celotex Settlement Trust also retains the right to temporarily deny access to proper passwords for any reason, including, but not limited to, system resources, bandwith constraints, and fraud or security concerns.

- 2.3 <u>Notice of Changes.</u> When a change is made to any user account within Primary Counsel's organization, the system will send an email to the Primary Counsel.
- 2.4 <u>Reliance on Primary Counsel Communications.</u> Celotex Settlement Trust is entitled to rely on communications and instructions it receives from persons using Celotex Online user accounts and passwords assigned by the Firm Administrator and purporting to act on behalf of the Primary Counsel and, except if Celotex Settlement Trust has acted with gross negligence, will not be held liable for such reliance.
- Acceptance of Terms and Conditions. Each time the Firm Administrator, or his approved designees, enters the Celotex Online system, he will be required to review a pop-up screen that will affirm that all on-line submissions to Celotex Settlement Trust will conform to the terms of this agreement and will be true and accurate to the best of the individual user's, Firm Administrator's, and Primary Counsel's knowledge, information and belief. The pop-up screen will have a hyperlink to an electronic copy of this Agreement. The pop-up screen will also require the user to assent to these terms and conditions by clicking "I Agree" or other similar language. The requirement of assenting to the terms and conditions before entering the system shall not in any way affect the validity or binding effect of this Agreement once executed.

3. <u>Technical Capabilities</u>

- 3.1 <u>Compatible Equipment.</u> In order to file claims through Celotex Online, the Primary Counsel will provide its own compatible computer equipment that meets technical standards that Celotex Settlement Trust will from time to time announce. The standards that Celotex Settlement Trust will set for this purpose will be among those then in general use among businesses transferring information electronically.
- 3.2 <u>Reports.</u> Celotex Settlement Trust will make available to the Primary Counsel through Celotex Settlement Trust's web site the ability to download specified data and to generate reports summarizing information regarding the Primary Counsel's on-line claim review.
- Continuous Access. Celotex Online system will be available to the Primary Counsel seven days per week, with the exception that availability will be interrupted nightly for approximately one hour to synchronize data with Celotex Settlement Trust processing systems and at any time as a result of system failure or when necessary for system upgrades, adjustments, maintenance or other operational considerations. When reasonably possible, Celotex Settlement Trust will notify the Primary Counsel in advance of any foreseen interruption of availability, and at all times Celotex Settlement Trust will use its best efforts to minimize the length and frequency of interruptions in Celotex Online availability. Notwithstanding any other provision of this Agreement, Celotex Settlement Trust

- shall not be liable for any damages resulting directly or indirectly from system unavailability.
- 3.4 <u>Security.</u> Celotex Settlement Trust will maintain physical, electronic, and procedural safeguards that will protect the information the Primary Counsel transmits though on-line claim review.

4. Rules and Procedures

- 4.1 Adherence to CRP. The Primary Counsel will assure that any submissions made to Celotex Settlement Trust through Celotex Online will conform to the rules and procedures established by Celotex Settlement Trust and contained in its Asbestos Personal Injury Claims Resolution Procedures as may be amended from time to time.
- 4.2 <u>Payment Criteria Unchanged.</u> This Agreement is not intended to alter and does not supersede the claim payment criteria contained in the Asbestos Personal Injury Claims Resolution Procedures. This Agreement does not create any rights to claim payment beyond those set forth in the Asbestos Personal Injury Claims Resolution Procedures.

5. Claim Information

- 5.1 <u>Complete and Accurate Information.</u> Consistent with the Primary Counsel's legal and professional responsibilities and the terms of this Agreement, the Primary Counsel, the Firm Administrator and individual users will provide complete and accurate information in any on-line claim review. Celotex Settlement Trust is not obligated to take any action on a claim until it has received all the complete information as required by its Asbestos Personal Injury Claims Resolution Procedures.
- Maintenance of Supporting Documents. The Primary Counsel will timely provide to Celotex Settlement Trust hard copies of documents relied upon in Celotex Online whenever instructed to do so by Celotex Settlement Trust. Whenever Celotex Settlement Trust does not instruct the Primary Counsel to provide copies of documents relied upon as part of the Celotex Online claim review, the Primary Counsel will maintain a copy of each document so relied upon, in either paper or electronic format, and thereafter will timely provide copies of such documents to Celotex Settlement Trust upon request.
- Pursuant to Section VII of the Claims Resolution Procedures, "all materials, records and information submitted by claimants, including that provided with medical audits under Section 7.3, are confidential, submitted solely for settlement purposes." Accordingly, any electronic submissions to the Celotex Settlement Trust under this Agreement shall remain confidential and will not be submitted to any third-party except in response to a valid subpoena or upon prior written consent of the claimant.

6. <u>Miscellaneous</u>

- 6.1 <u>Limited Damages.</u> Except as specifically provided in this Agreement, or as otherwise required by law, no officer, director, trustee, employee, contractor, or agent of Celotex Settlement Trust will be held liable for any indirect, incidental, special or consequential damages by reason of Primary Counsel's use of the Celotex Online.
- 6.2 <u>No Assignment.</u> Neither Party may assign or otherwise transfer in any way any of its rights and obligations arising out of this Agreement without the prior written consent of the other party.
- 6.3 <u>Termination</u>. Either Party may terminate this Agreement upon written notice to the other Party. Upon effective date of termination of this Agreement, Celotex Settlement Trust will cease providing the Primary Counsel with access to its system, and the Primary Counsel will cease making on-line claim submissions or having access to Celotex Online claim review.
- 6.4 <u>Disputes.</u> Any dispute between the Parties relating to the appropriate categorization or settlement value of a claim submitted by the Primary Counsel to Celotex Settlement Trust will be resolved pursuant to Celotex Settlement Trust's dispute resolution procedures.
- 6.5 <u>Force Majeure.</u> Neither party will be liable for any failure or delay in its performance under this Agreement due to any cause beyond its reasonable control, including acts of God, embargo, riot, sabotage, labor shortage or dispute, governmental act, or failure of Internet service, provided that the delaying party:

 (a) gives the other party prompt notice of such cause, and (b) uses reasonable efforts to correct promptly such failure or delay in performance.
- 6.6 <u>Applicable Law.</u> This Agreement shall be interpreted, construed and enforced according to the laws of the State of Delaware.
- 6.7 <u>Binding Effect.</u> Each of the undersigned persons represents and warrants that they are authorized to sign this Agreement on behalf of the Party they represent, and that they have the full power and authority to bind such Party to each and every provision of this Agreement. A signature on a copy of this Agreement transmitted by facsimile machine will have the force of an original signature.
- 6.8 <u>Severability.</u> If any term or provision of this Agreement should be declared invalid or unenforceable by a court of competent jurisdiction or by operation of law, the remaining terms and provisions of this Agreement shall remain in full force and effect.
- 6.9 <u>Entire Agreement and Waiver.</u> This Agreement constitutes the entire agreement and understanding between and among the Parties concerning the matters set forth herein. This Agreement may not be amended or modified except by another written

(D) (C) (D)	Date:
(Primary Counsel)	
	Email:
(print)	
	Date:
(for CELOTEX SETTLEMENT TRUST)	

ATTACHMENT 1 ELECTRONIC POP-UP SCREEN

The following will be imported into a pop up screen that will appear each time a user logs on for the first time:

By clicking the "I AGREE" box below the user affirms that all on-line submissions to Celotex Settlement Trust conform to the terms of the Electronic Filer Agreement executed by the Primary Counsel for the law firm and the Celotex Asbestos Settlement Trust. User further affirms that all submissions made during this on-line session are true and accurate to the best of the individual user's, User Administrator's, and Primary Counsel's knowledge, information and belief. In order to have access to this system you must click the "I AGREE" button below indicating your assent to these terms and conditions. In the event you wish to review the full text of Electronic Filer Agreement you may do so by clicking the hyperlink.

I AGREE

1227841.8

ATTACHMENT 2 DEFICIENCY NOTIFICATION PREFERENCE

	led each Attorney User the option of receiving hard copy deficiency and/onail. Please select notification preference:
DEFICIENCIES:	SEND DEFICIENCIES BY LETTER
	NOTIFICATION BY EMAIL
	ВОТН

EXHIBIT N

IN THE SUPERIOR COURT OF THE STATE OF DELAWARE IN AND FOR NEW CASTLE COUNTY

IN RE: ASBESTOS LITIGATION CHESTER LINK, et al.

06M-10-061 MMJ

AHLSTROM PUMPS, LLC, et al.

BEFORE: HONORABLE COMMISSIONER DAVID A. WHITE

APPEARANCES:

٧.

ROBERT JACOBS, ESQ. JACOBS & CRUMPLAR, P.A. for the Plaintiff

KATHARINE L. MAYER, ESQ. McCARTER & ENGLISH and PAUL SCRUDATO, ESQ. SCHIFF HARDIN LLP for the Defendant

ERIN EDWARDS, ESQ. YOUNG, CONAWAY, STARGATT & TAYLOR, LLP and DANIEL J. DONNELLON, ESQ KEATING, MUETHING & KLEKAMP, P.L.L. for the Celotex trust

> MOTION TRANSCRIPT **DECEMBER 7, 2006**

PATRICIA L. GANCI, RPR, CRR SUPERIOR COURT OFFICIAL REPORTERS 500 N. King Street, Suite 2609, 2nd Floor Wilmington, Delaware 19801-3725

December 7, 2006 Courtroom 8C 2:30 p.m.

2

PRESENT:

As noted.

THE COURT: All right. Why don't we go back on the record? I have also on the calendar for today some matters in an Ohio case named Chester Link which Judge Johnston had some involvement in which then led to the motions that are now currently pending. The motions were originally scheduled for a few weeks ago which we then past to today, and in part it was my hope that there would be some degree of resolution. And I'm just talking to the masses at this point. No one in particular. I was hopeful to hear that there's been some resolution, but who's going to take lead on this? Let's talk about it.

MS. MAYER: Your Honor, good afternoon. Katharine Mayer on behalf of Owens-Illinois.

THE COURT: Ms. Mayer, how are you? whole thing started with our motion to compel. So I

MS. MAYER: Good. How are you? I think this

would say we would start this process this afternoon. I do have with me my cocounsel, Paul Scrudato, S-C-R-U-D-A-T-O. And we did file a motion for admission pro hac vice, but it has not yet been entered. I would ask that he be able to participate today.

THE COURT: That's fine. That's perfectly acceptable. Am I correct that there's no agreement on any issue with respect to any of the papers that are before me?

MS. MAYER: I would say that there's been no agreement. We did propose a resolution yesterday, and it was rejected. So we are at a standstill as of today.

THE COURT: All right. That's fine.

MS. MAYER: Thank you.

THE COURT: Good afternoon, Welcome.

MR. SCRUDATO: Thank you. Good afternoon.

THE COURT: It would help me and maybe the folks on the other side if you could identify the issue or issues that we need to talk about rather than a full-blown presentation on the motion. Let's cut to the --

MR. SCRUDATO: This case, as you know, is pending in Ohio. An Ohio Judge requested that this

Court issue a subpoena permitting us to get discovery out of the Celotex trust. The discovery we're specifically interested in, Your Honor, is the production of the actual claims forms that Mr. Link presented to the trust as well as a very short, what they call in Ohio, a deposition on written questions. It's about 10 questions basically authenticating the documents.

THE COURT: I saw that.

MR. SCRUDATO: I would have expected that this was a fairly sort of routine and pro forma request --

THE COURT: I saw an e-mail a couple of weeks ago which I thought that was the telegraph I was attempting to send, but ...

MR. SCRUDATO: Well, we -- we have tried to work out an understanding with opposing counsel, Your Honor. I have not actually met -- I've not had an opportunity to meet the lawyers who are here for the trust or for the plaintiffs today since I'm from New York. Plaintiffs lawyers I believe are from Pennsylvania, and the trust lawyer I think is from either Ohio or Indiana.

The issue is very simple, Your Honor --

THE COURT: We should have name tags, I guess, shouldn't we?

MR. SCRUDATO: Would probably be a good idea.
Clearly, Your Honor, the Ohio Judge thinks this
information is important. He has asked that the
subpoena be issued explicitly out of this Court. Judge
Johnston signed that subpoena I believe on October 18th.
That is the motion we're seeking. We're seeking to
compel the production of the documents pursuant to
that ---

THE COURT: Let's make sure we understand or they understand. You are looking for copies of the original claim forms that were submitted on behalf of Mr. Link?

MR. SCRUDATO: Yes.

THE COURT: To the Celotex trust?

MR. SCRUDATO: Yes.

THE COURT: In addition to the claim forms themselves, any other document that you're looking for?

MR. SCRUDATO: It's really the claims forms and anything that was sent to the trust in addition to the claims forms by the plaintiff to substantiate the plaintiff's claim. So, for example, Your Honor, there

could be a couple of pages of medical records, that sort of thing.

THE COURT: So that would be, and I'm looking at your question No. 13 on the written interrogatory which makes reference to all claims submitted to the trust by or on behalf of Mr. Link, and here's the important part, and all supporting documentation filed therewith.

MR. SCRUDATO: Yes, that would be supporting documentation filed by Mr. Link. So anything he sent to the trust to justify receiving money from the trust, that's what we're looking for, Your Honor. As well as that short — you know, they call it in Ohio a deposition on written questions. It's just a way to authenticate. They're complete, authentic, that sort of thing.

THE COURT: Do Mr. Link's attorneys in the Ohio case, his personal injury attorneys, do they maintain copies of the very documents you're looking for?

MR. SCRUDATO: Yes, Your Honor. It is my understanding -- in fact, I just saw an order this morning -- that the Judge in Ohio actually directed Mr. Link's attorneys to give us this information.

THE COURT: So you wouldn't have to go through the trust to get it?

MR. SCRUDATO: Well, that's only part of the way -- gets us part of the way there. The Judge did direct the plaintiff's lawyers to provide this information to us.

THE COURT: By what date?

MR. SCRUDATO: I don't think there's a date in the order. I just looked at it, but I'm sure that they'll be on top of that in Ohio. We'll get it -- you know, the trial was set I believe in the first part of '07. That raises -- Your Honor is raising I think an issue that I suspect that will be raised today which is, Well, if you're getting it from the plaintiff's lawyers, why do you need to get it from the trust?

THE COURT: Yes, yes.

MR. SCRUDATO: That makes perfect sense, and we're sympathetic to that, Your Honor. There's two reasons for that. First of all, we are, I believe -- I believe, Your Honor, we're entitled to just make sure that the information we get from the plaintiff's lawyers is complete. You know, we want to --

THE COURT: So if you ask them as officers of

8

the court to provide you with copies of all of the documents submitted to the trust, including all claims and all supporting documentation filed therewith, and they submit documents to you as officers of the court in response of that, you don't think that's adequate?

MR. SCRUDATO: We just want to make sure, Your Honor, and the solution that we proposed yesterday --

THE COURT: Just the answer is, no, you don't think that's adequate?

MR. SCRUDATO: The answer is no.

THE COURT: Now, tell me why.

MR. SCRUDATO: Well, Your Honor, I guess to answer that question, I'm not sure of a rule being applied in any personal injury case anywhere in the country in any jurisdiction where a defendant gets medical records, employment records, from the plaintiff's lawyer that the defendant isn't then entitled to go to the employer, to the hospital, or to the doctor to make sure that those records are complete. What we proposed, Your Honor—

THE COURT: So you ask in your request, Provide us with a complete set of all claims submitted and a complete set of all supporting documentation filed

therewith, that still wouldn't be sufficient?

MR. SCRUDATO: It would not be sufficient. But let me just make a suggestion, Your Honor. I understand why you're — and this is a logical question and we're sympathetic to it. What we proposed yesterday was this: All right. Look, if the plaintiff's lawyers are going to give us this package of information, you know, 15, 20, 30 pieces of paper, we'll just take the 30, 40 pieces of paper, mail it to the trust, and just ask them to confirm that this is complete. We did not think that was onerous.

THE COURT: Why don't --

MR. SCRUDATO: We did not think that was worse. I thought that was a sensible solution to the problem. All we want to do is make sure what we got is the complete set of documents. We're entitled to be sure that's the complete set of documents. The only way we can be assured of that is if the trust, which actually has the documents, checks off on it. We're willing to do anything we need to do to make it easy on the trust. I personally thought mailing the stuff to the trust, which what is we proposed yesterday, was a sensible solution.

10

THE COURT: Before you filed your papers in Ohio asking for a subpoena here in Delaware, did you file a request for production of documents in the underlying Chester Link case directly on the plaintiff?

MR, SCRUDATO: I don't know when the production of the documents was filed in the Chester Link case, Your Honor. What I do know is that when all of this started back in, I believe, late September, we had none of the documents. In fact, as of today, we still don't have the documents, even though they've been ordered to be produced I believe by Judge --

THE COURT: You don't know when the date is?

MR. SCRUDATO: I don't know when the date is,
but we still don't have them.

So that's where we are, Your Honor. We -- I anticipate that -- I think the duplication issue is easily solvable. I do think we're entitled to make sure they're complete. I don't think it's going to take a lot for the trust to check off that they're complete. I think what we're going to hear today --

THE COURT: Beside the issue of the unduly burdensome?

MR. SCRUDATO: Right, which either I can

address that. It's their issue. Maybe they should address it first, and I can respond to it. I'll handle that however Your Honor wants me to proceed.

THE COURT: All right. Thanks. Appreciate your comments. That's the essence of what we're all about here this afternoon right now?

MR. SCRUDATO: That's it, Your Honor.

THE COURT: Okay. Let's hear from the plaintiffs or the trust, whoever wants to talk. Am I going to hear from both the plaintiffs and the trust?

MS. EDWARDS: Yes.

THE COURT: I just ask that we not duplicate any argument.

MR. JACOBS: I won't. The order entered by the Ohio Court gives until the 15th of December to produce any of the documents, the complete file that was submitted by plaintiff's counsel to Celotex.

THE COURT: Did the order in Ohio in any way indicate that if there was a disagreement on whether those documents are complete, that they could come to Delaware and ask for them to be authenticated by the trust?

MR. JACOBS: No, Your Honor.

12

THE COURT: Because I don't have a copy of the order.

MR. JACOBS: I've got an unsigned copy, but it doesn't talk about that. It just indicates --

THE COURT: Was there a hearing that prompted that order?

MR. JACOBS: Yes, Your Honor, on the 1st of December, and there was a bench -- an order put out including --

THE COURT: Okay.

MR. JACOBS: -- the two week --- -

THE COURT: And was the Judge in Ohio aware of what we were about to do here in Delaware?

MR. JACOBS: I assumed so since he signed the order on the original subpoena.

THE COURT: All right. Good afternoon.

MS. EDWARDS: Good afternoon. Erin Edwards from Young, Conaway, Stargatt and Taylor on behalf of the Celotex Settlement Trust. First, I'd like to address that we also didn't find out about the order until yesterday, and we prompted a conference call between all of the parties to see once again if we could resolve it in light of this order being entered on

December 1st.

I'm sorry. About their proposal to us, that actually is going to add more burden onto the trust and, in fact, liability to the trust. To make the trust take on the liability of verifying line by line to see if these items are exactly the same, what if there's a typo, what if the information is the same, but in a different place, it really --

THE COURT: Stop for a second. Is it logical or not logical to assume that documents submitted to the trust, stamped copies would be submitted and sent back to the plaintiff's attorneys in Ohio? No? No clock-in?

No stamp? No nothing?

MS. EDWARDS: Oh, I'm sorry. I apologize, Your Honor. My cocounsel, Dan Donnellon, from -- is also here. We did not admit him by pro hac to save some expense.

THE COURT: That's fine. I just want to know the answer to my question.

 $\label{eq:MS.EDWARDS: He can answer the questions} % \begin{subarray}{ll} \textbf{MS. EDWARDS: He can answer the questions} \\ \textbf{exactly for you.} \end{subarray}$

MR. DONNELLON: They're all stored electronically by the trust. And the claimant's law

1

firms, the vast majority of them, are electronic -- have the ability to go on and confirm that the trust has received, assigned a claim number, and what documents the trust has, and they can view them. The trust scans them. They can view what we have.

THE COURT: Okay. Aren't they given a document identification number or something like that?

MR. DONNELLON: They're given a claim number, but they're not coded with specific document identifications or file stamped like a court would.

It's a little less formal than that, but there is the availability of the -- of the plaintiff firm to confirm that they did receive them and to see where they are in that process.

THE COURT: So similar to our, if you're aware of it, our LexisNexis e-filing system where anyone can go into our docketing and pull off a copy of a document that was electronically filed or scanned and up-loaded. Can't do with the Celotex trust?

MR. DONNELLON: No, you can't. Not anyone. The documents that are filed --

THE COURT: For example, the plaintiff in Ohio case?

MR. DONNELLON: The plaintiff can, of course, but it's not a public record. So not everyone has access, but the plaintiff --

THE COURT: Let me --

MR. DONNELLON: -- can look at their own files.

THE COURT: Let me make sure I understand.

Mr. Link's plaintiffs -- Mr. Link's attorneys in Ohio have the ability to get on the web site of the Celotex trust and print out an accurate copy of the claim form and supporting documents for those claim forms that he submitted to the Celotex trust?

MR. DONNELLON: Absolutely. And he can then say, when he produces them, these are genuine and authentic as an officer of the court and they match exactly what the Celotex trust has.

THE COURT: All right. Let me stop you. How is — if they're able to do that, why is it that you have some suspicion that there may be some change in the document that was submitted as opposed to the one that was given to from you the plaintiff's attorney? I don't get it now.

MR. SCRUDATO: Your Honor, let me try and answer that question in the best way ${\bf I}$ can.

16

THE COURT: That's all I'm asking. Do you understand what -- at least what I understand is that the plaintiff's attorney in Ohio has the ability to get on the website for the Celotex trust and print out into his or her office a copy of the trust document and the supporting documents, print right out into his or her office.

MR. SCRUDATO: It is not at all a surprise to me, Your Honor, that the plaintiff's lawyer or us, for that matter, that I could get onto that website If I were given the appropriate log-in information, which I would suggest is the better solution. It's not a surprise to me --

THE COURT: Let me just stop you. I am at a loss for this distrust that I'm feeling between the defendants and the plaintiffs in the Ohio case. And you are not adequately explaining it to me.

MR. SCRUDATO: Your Honor, let me explain it.

THE COURT: You're an officer of the court -just stop. You're an officer of the court. The
plaintiff's attorneys in Ohio are officers of the court,
and you still do not believe they can provide you with
documents that are not true and correct copies of what

was submitted to the Celotex trust. And I don't get it. Now, explain.

MR. SCRUDATO: Your Honor, there have been other instances in Ohio cases where the information disclosed and the claims forms may or may not have accurately reflected what information the -- you know, the plaintiffs' actual exposures. And there -- we have every reason to believe that, you know, mistakes are made, Your Honor. Documents aren't always produced.

THE COURT: That doesn't make sense to me.

What you're saying doesn't make sense in light of what I just heard that a document is downloaded from the

Celotex trust into the plaintiff's attorney's office.

What mistake could possibly be made?

MR. SCRUDATO: We'd like to make sure we're getting the complete information, Your Honor, you know. And we are -- I -- we are entitled to make sure we're getting the complete information. We're not entitled to impose any burden on anybody. We understand that. What we're asking for is just an assurance that we're getting the complete file. It is the same assure --

THE COURT: So, for example, a well crafted request for admission submitted to the plaintiff's

18

attorney in Ohlo, admit or deny that the forms you've just submitted to me are true and correct copies of the forms submitted to the Celotex claims trust, that doesn't do it either?

MR. SCRUDATO: That would be, I believe, Your Honor, the second most effective way to solve this. The only real way we're going to be sure is if the trust assures us. The trust is the one who has the information.

THE COURT: All right.

MR. SCRUDATO: I'd just like to point one thing out, Your Honor, and I think this is important.

THE COURT: Go ahead.

MR. SCRUDATO: Counsel for the trust is absolutely correct. This stuff is all electronic. It's all submitted electronically. It's all stored electronically.

THE COURT: That's why I --

MR. SCRUDATO: What we're really talking about here --

THE COURT: Would you let me talk for a second? That's why I don't understand this lack of trust between the defendants and the plaintiffs in the Ohio case.

MR. SCRUDATO: Your Honor, the fact that there is -- a lack of trust exists is something I can't say other than it exists. I'm sorry for that. I apologize for that. It exists. I wish it -- I wish it didn't, but --

THE COURT: We don't practice law in Delaware that way. I'll leave it at that.

MR. SCRUDATO: I understand.

THE COURT: And I'm disappointed to hear -- at least what I'm hearing is that at least some segment of the Ohio bar practices law that way.

MR. SCRUDATO: I cannot help that, Your Honor. I would just like to point out sort of one thing on sort of how this could actually be accomplished. One way this could be accomplished that would satisfy my client's concern is that we go in and print out the information. What we're really talking about here, what we're really, really talking about here is I went to the Celotex website yesterday, Your Honor. If you go to the Celotex website, you're going to see that in 2005 they reviewed, reviewed, something like 135,000 claims.

THE COURT: I got an affidavit here on this record that says the same thing.

2

MR. SCRUDATO: What that means, Judge, is that every single day some person or persons at the Celotex trust are reviewing on the order of five or 600 claims a day, five or 600, and actually making a determination about those claims. What we're really asking them to do is pull up one of those claims and push print and send them to us. That's what we're asking.

THE COURT: I understand.

MR. SCRUDATO: That's not burdensome, Your Honor. And, again, I feel as though I've not appropriately addressed your concern about this trust issue. I'm from New York. I don't practice in Ohio. I don't practice in Delaware. It obviously exists.

THE COURT: It clearly does.

MR. SCRUDATO: Yeah.

THE COURT: All right. I appreciate it.

MS. MAYER: Your Honor, Katharine Mayer again. I just want to clarify one thing because I did participate in some of the phone calls in this case.

And I think one of the issues I would just like to point out is my understanding, and I could stand corrected by counsel for the trust, but my understanding was that the plaintiff's counsel actually submits information over

the Internet, the information highway, that then creates the claim form. That is an input into the claim. So I didn't have the impression from plaintiff's counsel that he actually was going to be producing to us something he was going to download off the Celotex trust. My understanding was that he — he gave me the impression, I may have misunderstood, that his information actually could be different from what the trust actually has in its possession because he just had the information in his hands versus they have the actual completed forms.

THE COURT: But I thought I asked and got the answer to my question, which was could the actual claim form itself and the documents attached to it be downloaded and printed out. And I thought I heard the answer was yes to that.

MS. MAYER: Today is the first time we heard that that would be an option that they would produce to us so we weren't aware. We were being told that their information actually could be different from what the trust had.

THE COURT: Here's what is bother, Ms. Mayer.

Here is what is bothering me. I've been involved in
this matter for what? A half an hour, You have been

involved since what? I heard since October. Why couldn't that question have been asked and answered?

MS. MAYER: The request for production for documents was served in Ohio, and I believe we do have a copy of the order with us granting the motion to compel, if you'd like to review it.

THE COURT: No, I don't think I need to, really.

MS. MAYER: I don't see the date of December 15th in there, but again I would stand corrected by Ohio counsel. And also I think the hearing was November 28th. So we're talking about recent events that just occurred in the last couple of weeks, and we didn't at that point have any other option. And, again, the trust, my understanding, has the complete documents. And they did send an e-mail to us back in September saying they found Chester Link's claim. So if we're talking about burden issues, they found the claim. We're asking to — they found the claim and we're offering to pay any costs, copying, witness fees, anything like that.

We did produce, you know, the questions that would be posed to the witness. I believe a witness was

even — a name was provided to us. We worked out the date and time for everything. So we did try to accommodate any issues with respect to that. We believed they were resolved at one point in time, and then, you know, things kind of steamrolled out of hand. And we realize they were not, and that's where we are today. But I did want to clarify that because I was involved in some of the phone calls that Mr. Scrudato was not involved in.

THE COURT: Thanks. I appreciate that. Ms. Edwards.

MS. EDWARDS: I would like to address that as well. We did from the outset let them know, what we're asking from us is we're a third party. We have nothing to do with this litigation, and these are forms prepared and submitted by the plaintiffs to us.

THE COURT: Is this sort of a macro, that information is typed in by Mr. Link's attorneys in Ohio onto a macro?

MS. EDWARDS: In fact, the claims forms are available. They're heavily located in a planned process to come up with a claim form. So I believe it's accurate to say that the plaintiff themselves have to

fill out the claim form. So from our -- our position from the start is, Yeah, we're just in the middle of this.

THE COURT: We've litigated similar-type issues here in Delaware in the Delaware asbestos cases. I've seen lots and lots of the claim forms.

MS. EDWARDS: Correct. And I also wanted to address their concern as well about this mistrust issue and about their ability — it's their — it's their ability to be able to pursue this discovery even though they know this duplication is out there. It is actually —

THE COURT: Well, they found out today apparently.

MS. EDWARDS: Right. I was like, Well, it's within the Court's purview under Rule 26(b)(1) --

THE COURT: Right.

MS. EDWARDS: -- `to be able to limit if the sources are available somewhere else that's less burdensome. And I believe our affidavit sets forth very clearly, you understand the enlarged scale of these asbestos claims. And, in fact, the Celotex trust is one of the largest claims processors, including now with all

of the new trusts actually being established. So we can't look at this in a vacuum to the trust. It's not just the one claim.

THE COURT: Ms. Edwards, let me just tell you, this Court has in the last six weeks issued new case management orders in the asbestos cases. Part of our case management order mandates, requires, a plaintiff in every single case to turn over to the defendants copies of trust forms and attachments to them. We do that.

 $\label{eq:MS.EDWARDS: I agree. That was next on our list. So thank you. \\$

THE COURT: All right. I preempted you. I'm

MS. EDWARDS: That's fine. I just wanted as one more note, as far as confidentiality, while some trusts are different, our trust claims resolution procedures which we attached to our response clearly say that we submitted to these plaintiffs. You're submitting these forms to us that we're going to keep them confidential, and they're deemed part of a settlement process. So it's not up to our discretion if our beneficiary of the trust says these are confidential.

2

THE COURT: But if I ordered you to turn them over, you would be obligated to turn them over.

MS. EDWARDS: Obviously.

MR. JACOBS: Your Honor, not to reargue.

THE COURT: No, Mr. Jacobs.

MR. JACOBS: Just some information. Until recently the defendants in Ohio had not requested from the plaintiff via request for production the forms which was done probably just a few weeks ago prompting the November 28th hearing and the December 1st order. Just so all of this information with the Celotex trust, the way I know it's done is the plaintiffs have the forms. They scan them in. Then they become electronic at the trust. So basically what they have is a PDF-type version which isn't changed once it gets to the trust. It's what the plaintiff had done.

So if we then download that document, it's obviously exactly what the trust has. In fact, it's really exactly what the plaintiff counsel would swear they have because it's the same document. It's just that one wound up there on a PDF because it was scanned.

So I believe that the real question that this Court has already answered was, even though these are confidential and settlement-type documents, some of the information within them is very -- is or may be relevant.

THE COURT: That's been litigated here before me and it's been decided by me.

MR. JACOBS: I'm not arguing this time.

Therefore, obviously, they should be produced, but I believe that being --

THE COURT: The question is whether they're relevant under Rule 26. That's all.

MR. JACOBS: When they're downloaded, so they can see the information, that is the actual document.

THE COURT: Right. Do you agree --

MR. JACOBS: I agree.

THE COURT: And I know you weren't happy with it, but do you agree that the defendants are entitled to see the claim forms, the content of the claim forms?

MR. JACOBS: And the supporting documents.

THE COURT: For purposes of alternative exposures and other -- whatever they deem relevant issues?

MR. JACOBS: Whatever, Your Honor. They are permitted, and I think that once plaintiff counsel in

2

Ohio downloads it with his access number and verifies that it is — in fact, they offered to stipulate that what they're giving them is what they downloaded. I think that should end it, and we shouldn't put the trust and expose the trust with the thousands and thousands of claims that this is going to come — come a practice and that counsel is going to say that we don't trust. I think that that is — goes contrary to what this Court should impose on attorneys. And that —

THE COURT: Well, for example, let me just throw out a hypothetical. What if Mr. Link's attorneys in Ohio don't have or couldn't get access to the claim forms from the trust electronically?

MR. JACOBS: I think even then if they have the original forms they up-loaded and they verify it in writing that this is the actual claim form filed, that unless there's something that I can be shown where these people lied in front of the Court before and there's some type of an order against them where they've been disbarred, I don't understand this distrust that New York counsel said.

THE COURT: Remind me again, and I'm sorry, remind me again the trial date for the Chester Link

case.

MR. JACOBS: January what? January 29th, Your Honor.

THE COURT: All right. So we have about 45 or so days.

MR. JACOBS: Roughly.

THE COURT: And what I'm about to say is for record purposes. If a defense attorney in Ohio was to receive copies of the claim forms and to submit a request for admissions to the plaintiff, is there still time for that plaintiff to respond to the request for admissions to determine the authenticity and accuracy of those claim forms that were submitted?

MR. JACOBS: Your Honor, why don't I put it this way. I think there is, but I think that is an unnecessary step.

THE COURT: I want an answer to my question.

MR. JACOBS: Yes, Absolutely. There is time
to do that, and I'm sure that plaintiff's counsel would
not oppose that that be an expedited process so that it
can be done in seven days after they get that request.
So there's no doubt that we don't have to wait whatever
time Ohlo might require. I don't see any difference

between that, though, Your Honor, and plaintiff's counsel putting in an affidavit form signed by him that this is -- when he submits the form that this is the accurate and exact copy of what is on the -- what is on their trust document and what was submitted. And I think that sort of leap frogs because all of these requests for admission done is the same thing. So I really think it should be denied now that they're being

MR. DONNELLON: Your Honor, there is one more thing.

produced.

THE COURT: Please come to the podium.

MR. DONNELLON: Good afternoon, Your Honor.

Dan Donnellon. I'm not admitted pro hac vice in this matter because it had already incurred enough expense, but I would like to address one thing. When we found out, because we've been saying at the trust all along, These documents are available from the plaintiff interaction. Why do you need to burden us or bother us? And when we found that it's actually been ordered yesterday and that they were told that they will stipulate to the genuineness and authenticity, this is exactly what the trust has, I was exasperated. And we

made the discovery proposal. I said, Let's still try to resolve this instead of wasting the Court's time and our time.

THE COURT: I've enjoyed this so far, I will admit.

MR. DONNELLON: I had said, Why don't we simply say from now on, Owens-Illinois, you ask the plaintiff for them and you're entitled to get them from the plaintiff. And if they're unavailable, if they say they were destroyed in Hurricane Katrina or for some reason you have good faith reason to believe that they shouldn't be trusted, then you can come to us and subpoena our records. But why bother us with that? And, instead, we're still here at 5 after four. I'm familiar with Your Honor's standing order. And what I was asking Ms. Edwards is If we could request some form of a standing order like that. That the trust that is located here in Delaware gets hundreds of these subpoenas, and the standing order would be that you don't get an order out of the Delaware Court for a miscellaneous docket action unless you've agreed to ask for the plaintiff to produce them.

THE COURT: That's why I asked the question a

1

half an hour ago.

 $\label{eq:mr.donnellon: Thank you. That's exactly what I was hoping. \\$

THE COURT: Let me ask you, though, just for my own purposes. The trust receives subpoenas throughout the year?

MR. DONNELLON: Yes.

THE COURT: Does the trust move to quash every subpoena all the time?

MR. DONNELLON: No. In fact, this is the very first time -- I've been involved with hundreds of them. This is the very first time that we've come into court even on a motion to compel. We have always managed to resolve them through extrajudicial means, and we take the position that -- not that Your Honor hasn't determined this aiready, about the confidentiality. We take the position it's confidential because we promised them we would keep it confidential. It doesn't make it confidential. We promised them we wouldn't give it out until we're ordered by a court to do so or you waive it.

So occasionally there are objections and disputes of that nature, but we've managed to resolve hundreds of them so far. And we will get hundreds more.

There are more waiting on this same issue where either they don't trust the plaintiff counsel in their case or they have never even bothered to request the plaintiff counsel to produce those documents. And, instead, they want to take time from us and our employees.

THE COURT: Well, we had a little -- we had some battles here in this court. Before we got some rulings, there were some battle lines drawn.

MR. DONNELLON: I'm familiar with that.

THE COURT: All right. Anything else from this side? All right. Thanks.

MR. SCRUDATO: Just very briefly, Your Honor.

I only want to address one thing. You know, we did -as I told Your Honor from the start, we did try from -we've been trying for months to get these materials.

All right. I -- you know, Your Honor, raised this
mistrust Issue which I think is very important.

THE COURT: Well, no, no. I didn't raise it.

Your response to my question prompted it.

MR. SCRUDATO: We -- it's peculiar to us, Your Honor, that when we first approached the trust to get these documents back in late September, the trust was prepared to work out an arrangement with us just as

34

Mr. Donell and his -- Donell?

MR. DONNELLON: Donnellon.

THE COURT: Donnellon.

MR. SCRUDATO: Donnellon is suggesting. That's how we've done this many times in the past. For some reason, Your Honor, after those first contacts, the posture of the trust changed. And I think the reason why the posture of the trust changed is not because of a decision the trust made, but because the plaintiff's lawyers in Ohio said, We're now going to object to this. So we're confronting a situation --

THE COURT: I wouldn't doubt about that.

MR. SCRUDATC: And that's how I think this
thing unfolded. So when this was originally presented
to the Ohio Judge, all right, the Ohio Judge is the one
who heard lots of argument or, you know, heard extensive
briefing from the plaintiffs and from Owens-Illinois why
this was the -- this, the order signed by Judge Johnston
a month or so ago here in Delaware, was the appropriate
resolution to the problem before the Judge in Ohio.

THE COURT: Didn't the Judge in Ohio ask the folks to come to Delaware and get a commission if that's what they wanted to do?

MR. SCRUDATO: Well, that's exactly -THE COURT: It was an invitation. I read it or
understand it that the Judge in Ohio in some way to
appease both sides said, I don't need to tell one or the

other. I'm going to let the Delaware court decide it.

MR. SCRUDATO: Well, I think what they thought was that they didn't have the jurisdiction over the trust because the trust is in Delaware. And they did what they should have done procedurally which is to issue the commission to the Delaware Judge who would then sign the commission which gives us a lawful subpoena. I don't think it was a function of not wanting to decide it. They had no jurisdiction to decide it.

THE COURT: Well, that's fine.

MR. SCRUDATO: For whatever reason the solution -- you know, the order that the Judge in Ohio signed involving the case before that Judge, this was the appropriate solution. And, again, I was not -- you know, I did not write those papers. I was not at that hearing. But I would ask, Your Honor, to, you know, bear in mind that the case that -- the Judge that has this case used this, our relief, as the appropriate

resolution to this problem.

THE COURT: Why then did the Judge issue an order compelling the plaintiff to turn the documents over?

MR. SCRUDATO: Your Honor, that is something -you know, I looked at that. Every judge in the country,
you know, I think it's been addressed in 14
jurisdictions. It's a pro forma order. I don't know of
any judge in the United States that is not ordering the
plaintiffs to turn those documents over when they're
requested. I don't think that was a particularly
noteworthy turn in the case. I'm pleased they're going
to turn that information over.

The only issue before Your Honor is, how are we assured that those documents are complete? And I would just remind Your Honor that in -- you know, we always confront in this jurisdiction -- I mean, in this litigation asbestos-only rules. You know, when I do a nonasbestos case, if I'm representing a defendant who is sued in an auto case and if I'm given medical records from the plaintiff's lawyer, in every single case in the United States we go to the hospital and order up those medical records. That's the only way we're assured

they're complete. And that's exactly what we're asking to do here, Your Honor.

For some reason in this case, we've gotten all of these objections. It's new to us. We're not asking Your Honor to do anything that's particularly, you know, unique as a civil practice issue. It's what we do in every single case that we have that's not an asbestos case.

THE COURT: I will tell you that I have not asbestotized my thinking process in this matter in any way. I rule -- I view Rule 26 the same whether it's an asbestos case or a nonasbestos case. If it's relevant, it's got to be turned over. I don't draw distinction. Thanks for your comments.

MR. SCRUDATO: Thank you, Your Honor.

THE COURT: I'm going to deny the application I guess formally known as a motion for an order compelling the deposition of the records custodian in light of the responses to the questions that I had from the bench, which is two things. No. 1, the Judge in Ohio has already ordered that these trust documents and the attachments thereto be turned over. Whether it's December 15th, I don't know. I've been told it's

December 15.

In response to my question that these documents are scanned and up-loaded and maintained electronically by the Celotex trust, compels me in making my decision that there is nothing that's going to be different from what the plaintiff's attorneys in Ohio download from what the Celotex trust would be providing to the defendants if they were to download the documents themselves here in Delaware. There's no difference. They're maintained electronically. There's no way that I can think of that those documents would be any different. It's not a question of privilege. It's not a question of undue burden. It's a question that I have thought about and answered that those documents are available from another source. The source being the very plaintiff in the Ohio case. His attorney can provide those documents which are exact duplicates of the claim forms and the attachments that were submitted to the trust.

If there is ongoing disagreement or a lack of trust after those documents are provided, then I think I telegraphed at least one, maybe two, possible solutions to reduce that distrust: an affidavit from the

plaintiff's attorney which accompanies those documents, a request for admissions by the defense attorney in Ohio sent to the plaintiff's attorney in Ohio. Both of those as officers of the court would satisfy this Court that there is no deviation from the documents provided in response to the Judge's order in Ohio. No difference between those documents and documents that the trust would provide directly to the defendants in the Ohio case.

Again, if it's an issue of relevance, that issue has been argued and decided here in Delaware. Those documents are relevant. They're relevant to the defendant in Ohio to determine whether there are alternative exposures, whether there are things that the plaintiff may have said in his deposition in the Ohio case which differs from the content of material that was submitted to the Celotex trust here in Delaware, whether coworkers say things differently under path in Ohio than what's submitted in those trust documents. Any number of things could be different which this Court has determined is relevant to the Chester Link case in Ohio.

The issue is whether there's some distinction between what the plaintiffs will be providing the

40

defendant and what the trust will be providing the defendant. It's not an issue of undue burden. It's an issue that the plaintiff in the Ohio case, the Chester Link case, can provide exact duplicates of those documents without any burden whatsoever put on the trust to provide the exact same document. And to me, that's an easy one. At that point it's an easy one. Get the documents from the plaintiff's attorney. They've already been ordered to provide those documents, and there will not be one difference between those documents. So for those reasons, I'm going to deny the application.

Questions? Are counsel okay allowing the transcript of that decision to serve as my order or do you want to submit an order? I'll leave it up to you to make the decision.

MR. JACOBS: Plaintiffs are fine with the transcript, Your Honor.

 $\label{eq:mr.scrudato:} \textbf{MR. SCRUDATO: That's fine, Your Honor.}$ Thanks.

THE COURT: Any other issues that we can decide or determine with respect to the Chester Link case?

incurred a lot of costs fighting this issue, and we were the ones trying to --

THE COURT: Don't. I would recommend you not go down that road.

MS. EDWARDS: Sure. Thank you.

THE COURT: For intellectual purposes, this has been enlightening to me. I hope it's been somewhat enlightening to you. Attorney fees don't get awarded in those types of cases. I don't find any misuse of the process at all. I just appreciate the opportunity to have reviewed it and decided it for you.

Anything else? Okay. Thanks for your time. Stand in recess.

(Whereupon the proceedings concluded at $4:11 \ p.m.$)

42

STATE OF DELAWARE:

NEW CASTLE COUNTY:

I, Patricia L. Ganci, Official Court Reporter of the Superior Court, State of Delaware, do hereby certify that the foregoing is an accurate transcript of the proceedings had, as reported by me in the Superior Court of the State of Delaware, in and for New Castle County, in the case therein stated, as the same remains of record in the Office of the Prothonotary at Wilmington, Delaware, and that I am neither counsel nor kin to any party or participant in said action nor interested in the outcome thereof.

WITNESS	my	hand	this	 day	0
	2	กกล			

Patricia L. Ganci, RPR, CRR Cert.#169-PS